Contents lists available at ScienceDirect

## Expert Systems with Applications

journal homepage: www.elsevier.com/locate/eswa





CrossMark

Applicatio

## Abbas Mardani<sup>a,\*</sup>, Ahmad Jusoh<sup>a</sup>, Edmundas Kazimieras Zavadskas<sup>b</sup>

<sup>a</sup> Department of Management, Faculty of Management, Universiti Teknologi Malaysia (UTM), Skudai, Johor 81300, Malaysia <sup>b</sup> Department of Construction Technology and Management, Vilnius Gediminas Technical University, Sauletekio al. 11, LT-10223 Vilnius, Lithuania

#### ARTICLE INFO

Article history: Available online 10 January 2015

Keywords: Fuzzy multiple criteria decision making Multiple criteria decision making Literature review

# ABSTRACT

MCDM is considered as a complex decision-making tool involving both quantitative and qualitative factors. In recent years, several fuzzy FMCDM tools have been suggested to choosing the optimal probably options. The purpose of this paper is to review systematically the applications and methodologies of the fuzzy multi decision-making (FMCDM) techniques. This study reviewed a total of 403 papers published from 1994 to 2014 in more than 150 peer reviewed journals (extracted from online databases such as ScienceDirect, Springer, Emerald, Wiley, ProQuest, and Taylor & Francis). According to experts' opinions, these papers were grouped into four main fields: engineering, management and business, science, and technology. Furthermore, these papers were categorized based on authors, publication date, country of origin, methods, tools, and type of research (FMCDM utilizing research, FMCDM developing research, and FMCDM proposing research). The results of this study indicated that, in 2013, scholars have published papers more than other years. In addition, hybrid fuzzy MCDM in the integrated method and fuzzy AHP in the individual section were ranked as the first and second methods in use. Additionally, Taiwan was ranked as the first country that contributed to this survey, and engineering was ranked as the first field that has applied fuzzy DM tools and techniques.

© 2015 Elsevier Ltd. All rights reserved.

#### 1. Introduction

Multiple-criteria decision making (MCDM) has grown as a part of operations research, concerning with designing computational and mathematical tools for supporting the subjective evaluation of performance criteria by decision makers (Banaitiene, Banaitis, Kaklauskas, & Zavadskas, 2008; Behzadian, Khanmohammadi Otaghsara, Yazdani, & Ignatius, 2012; Zavadskas, Skibniewski, & Antucheviciene, 2014). Several studies have been carried out to develop MCDM (Dadelo, Turskis, Zavadskas, & Dadeliene, 2014; Shyur & Shih, 2006; Yazdani-Chamzini, Shariati, Haji Yakhchali, & Zavadskas, 2014). In recent years several previous studies have employed MCDM tools and applications for solve areas problems such as engineering (Zavadskas, Antucheviciene, Hajjagha, & Hashemi, 2014), science (Zavadskas et al., 2014), technology (Bagočius, Zavadskas, & Turskis, 2014; Dadelo et al., 2014; Streimikiene, Balezentis, Krisciukaitienė, & Balezentis, 2012). In real world, problems in regard to decision making are generally uncertain in a number of ways. Lack of information can lead to an unclear future state of the system. This uncertainty has been addressed using the probability theory and statistics. Though, in various situations of daily life; for evaluation, judgment, and decision, natural language is often employed in order to articulate thinking and subjective perceptions. In these natural languages, words might not have a clear and well-defined meaning. As a result, if the word is used as a label for a set, the boundaries of the set to which objects do or do not belong will become fuzzy. In addition, when individuals are judging an event, even using the same words, they may significantly differ since each of them has different subjective perception or personality.

To overcome this problem, fuzzy numbers are introduced in a way to help linguistic variables be expressed appropriately. Due to the fact that investors often evaluate investment strategies based on their own subjective preferences in terms of numerical values from different criteria, it is better to regard this situation as a Fuzzy Multiple Criteria Decision-Making (FMCDM) problem.

The present study has the following contributions: Fuzzy FMCDM is one of the most widely used decision methodologies in engineering, technology, science and management and business. FMCDM approaches improve the quality of decisions by creating



<sup>\*</sup> Corresponding author. Department of Management, Faculty of Management, Universiti Teknologi Malaysia (UTM), Skudai, Johor, Malaysia. Tel.: +60 1127347432.

E-mail addresses: mabbas3@live.utm.my (A. Mardani), ahmadj@management. utm.my (A. Jusoh), Edmundas.Zavadskas@vgtu.lt (E.K. Zavadskas).

the development more efficient, rational and explicit. Several studies (Behzadian, Kazemzadeh, Albadvi, & Aghdasi, 2010; Ho, 2008; Vaidva & Kumar, 2006) have demonstrated the vitality of the field and reported several methods proposed in the literature. A large number of approaches and techniques have been introduced in this area of study. However, previously-conducted surveys have not kept the pace. Thus, we believe that there is a need for a new systematic survey to consolidate recent research conducted on this area of study. In recent decade, the FMCDM methods have received a great deal of attention from practitioners and researchers. This paper attempts to document the exponentially grown interest in the FMCDM methods and provide a state-of the-art review of the literature regarding the FMCDM applications and methodologies. Based on a classification scheme, a reference repository has been established, including 403 papers published in more than 150 international journals from 1994. Papers are classified based on the year of publication, application areas, authors' nationality of authors, and FMCDM approaches combined with other methods. This paper is evolving a categorizing structure with focusing on applicable considerations, presenting an organized review in a way to provide a guide to previous studies on the FMCDM methods, and recognizing topics for future research. Additionally, in our study, two new perspectives are taken into consideration to review the articles, namely categorization of the articles into four main fields (business, science, engineering, and technology) and examination of the type of study (FMCDM utilizing research, FMCDM developing research, FMCDM proposing research).

In this paper, the literature related to the descriptors of FMCDM has been reviewed comprehensively using academic databases of Springer, ScienceDirect, Emerald, Wiley, ProQuest, and Taylor & Francis. Following a methodological decision analysis on the whole collected articles, a total of 403 international journal articles published from 1994 to 2014 were reviewed. The present paper attempts to answer the following questions: (1) which fuzzy decision-making (DM) techniques have been used frequently? (2) Which type of study has been conducted on these FMCDM techniques? (3) Which one of the four fields (science, business, technology, and engineering) has further used these FMCDM techniques? (4) What kinds of FMCDM tools have been employed in these years based on four main fields? (5) Which countries have published articles related to these FMCDM tools based on the number of publishing in these four main fields? (6) In which one of the years authors have further published articles related to FMCDM tools based in the four fields?

The remainder of this paper is organized as follows. Section 2 provides a brief overview on literature review and framework. Section 3 describes the research methodology and the procedure of this study. Section 4 provides findings of this review based on the research objectives and questions. Section 5 discusses the results based on the research questions. Finally, Section 6 presents conclusion, limitations, and recommendations for future studies.

#### 2. Literature review on the fuzzy decision-making approaches

Over the years, numerous FMCDM methods have been proposed in the literature, which are different in areas such as the type of questions asked, theoretical background, and type of obtained results. A number of methods have been designed for a particular problem, hence inapplicable to other problems. Recently, a number of FMCDM methods have been introduced to choose the best compromise options. The FMCDM approaches have been developed not only by the motivation received from various real-life problems that require the consideration of multiple criteria, but also by the desire of practitioners for enhancing decision-making techniques through recent developments occurred in computer technology, scientific computing, and mathematical optimization (Wiecek, Ehrgott, Fadel, & Rui Figueira, 2008). All methods are mainly aimed to make the decision making process better-informed and more formalized. In a number of previous studies, MCDM and fuzzy MCDM have been classified into various fields and approaches (Baležentis, Valkauskas, & Baležentis, 2010; Chen, Hwang, Beckmann, & Krelle, 1992; Liou, 2013; Liou & Tzeng, 2012; Ölçer & Odabaşi, 2005; Ribeiro, 1996; Zavadskas & Turskis, 2011). The MCDM approach fall into two categories (Bashiri, Badri, & Hejazi, 2011; Pawlak, 1982; Wang & Lee, 2007; Wang & Lee, 2009; Xu & Da, 2002): classical MCDM and fuzzy FMCDM (Wang, Lee, & Lin, 2003). Fig. 1 presents the Fuzzy Multicriteria decision-making process. In the FMCDM approach, alternatives are ranked and selected from among a set of feasible alternatives.

The MCDM problems have different nature: thus, a variety of techniques have been proposed as the solution. The first proposed methods were complete aggregation ones. For example, SAW and two stages in weighting (MacCrimmon, 1968), MAUA (Keenev & Raiffa, 1976), WASPAS (Zavadskas, Turskis, Antucheviciene, & Zakarevicius, 2012), TOPSIS (C. Hwang & Yoon, 1981), VIKOR (Opricovic & Tzeng, 2004), COPRAS (Zavadskas, Kaklauskas, & Sarka, 1994), MOORA (Brauers & Zavadskas, 2006), COPRAS-G, ARAS-F, ARAS-G and MULTIMOORA (Brauers & Zavadskas, 2010; Turskis & Zavadskas, 2010b; Zavadskas & Turskis, 2008), ARAS (Turskis & Zavadskas, 2010a; Yager, 1994a). As examples of partial aggregation methods, PROMETHEE (Mareschal & Brans, 1992), ELECTRE (Roy, 1996), and NAIADE (Munda, 1998) can be listed, which involve the pair-wise comparisons of alternatives. In addition, the ANP and AHP are relied on the pair-wise comparisons (Saaty, 1988; Saaty, 2003; Saaty, 2005; Saaty & Vargas, 2006).

FMCDM can be categorized as a fuzzy multi objective decisionmaking (FMODM) and fuzzy multi attribute decision-making (FMADM) approach (Kadane, 2011; Liou & Tzeng, 2012; Ölçer & Odabaşi, 2005; Simões-Marques, Ribeiro, & Gameiro-Marques, 2000; Tzeng & Huang, 2011). Liou and Tzeng (2012), addressed



Fig. 1. Multicriteria decision process.

the development of MADM from 1738 to 2012. They put MADM into three classes: evaluating or choosing models (e.g., DEMATEL, fuzzy DEMATEL, ISM, fuzzy ISM, fuzzy cognitive map (FCM), linear structure equation models (LISEM, or called "SEM"), formal concept analysis, and input-output analysis), weighting models (ANP/fuzzy ANP, AHP/fuzzy AHP, entropy measure, neural network weighting, and dynamic weighting), and normalizing models (additive types: TOPSIS, SAW, VIKOR, ELECTRE, PROMETHEE, and gray relation and non-additive types: fuzzy integral neural network plus fuzzy). In another classification related to MCDM tools and approaches, Zavadskas and Turskis (2011) and Hwang and Yoon (1981) have grouped MCDM tools in different way, in two these studies, MCDM divided into three types information of actors including; no information (dominance, Maxmin and Minmax classes), information about criteria (standard level (conjunctive and Disjunctive), ordinal (lexicographic, elimination by aspects and permutation), cardinal (liner assignment, SAW, HSAW, TOPSIS, ELECTRE, PROMETHEUS, ORESTE, VIKOR, COPRAS, ARAS and MOORA) and marginal rate of substitution (hierarchical tradeoffs)) and information of alternative (pairwise preference (LINMAP and SAW) and order of pairwise proximity (multidimensional scaling)).

The objective of FMADM is finite and implicit, whereas the objective of the FMODM approach is infinite and explicit. In FMADM, the decision maker's objectives are unified under the decision-makers' utility that is a super function, which is dependent upon the selection criteria. In FMODM, objectives of the decision-makers, e.g., optimal resource utilization and quality improvement and remain explicit are assigned with fuzzy weights that reflect their relative significance. The most important benefit of FMCDM models is their capability of considering many selection criteria. Bellman and Zadeh (1970) and Zimmermann (1978) applied fuzzy sets to the MCDM field. According to Yager (1978), the fuzzy set of a decision is the intersection of the whole fuzzy goals. Kickert (1978), prepared a summary of applications of the fuzzy set theory to MADM problems.

Literature contains a number of classifications of MCDM tools with fuzzy theory sets. For instance, Peneva and Popchev (2008) stated that if the weights are given as real numbers, the operators such as Weighted Mean (Chiclana, Herrera, & Herrera-Viedma, 1998), Weighted MaxMin and Weighted MinMax (Fodor & Roubens, 1995), and Weighted Geometric (Chiclana, Herrera, & Herrera-Viedma, 2000) can be applied to the aggregation of fuzzy relations. In the mathematical model, there are operators in which the weights do not present the operators: Min, Max, MaxMin (Altrock, 1997), Gamma (Altrock, 1997), and Generalized Mean (da Costa Sousa & Kaymak, 2001). The idea of using the given weights in this case was suggested in (Yager, 1994b). The two other categories proposed by Hwang, Chen, and Hwang (1992) consist of some ways for finding a ranking, including the degree of optimality, linguistic ranking methods, comparison function, Hamming distance, proportion to the ideal, fuzzy mean and spread, centroid index, left and right scores, and area measurement. The second category contains methods that employ different ways for assessing the relative significance of multiple attributes, including analytic hierarchy process, fuzzy simple additive weighting methods, fuzzy outranking methods, fuzzy conjunctive/disjunctive methods, and maximin methods. Inuiguchi, Ichihashi, and Tanaka (1990), carried out a survey on recent developments occurred in fuzzy programming. In this work, they employed applications such as flexible programming, possibilistic programming, possibilistic linear programming with fuzzy goals, possibilistic programming with fuzzy preference relations, possibilistic linear programming using fuzzy max, and robust programming.

Based on the relationship among aggregated arguments, the aggregation operators can be divided roughly into two classes:

those operators that consider the aggregated arguments dependently and those that consider the aggregated arguments independently. In case of the first class, Yager (1986) introduced the ordered weighted averaging (OWA) operator for reordering the arguments prior to being aggregated. This operator motivated Chiclana et al. (2000) and Xu and Da (2002) to propose the ordered weighted geometric (OWG) operator. Yager (2004), used the continuous interval-valued arguments to develop the continuous ordered weighted averaging (C-OWA) operator. Torra (2010) and Torra and Narukawa (2009) developed the hesitant fuzzy sets (HFSs) concept to present the hesitant fuzzy information, which covers the arguments with a set of possible values. This is considered as a new efficient tool for collecting and representing the arguments under uncertainty, particularly in the decision making process. Zhu, Xu, and Xia (2012), investigated the geometric BMs combined with hesitant fuzzy information and introduced the hesitant fuzzy geometric BMs (HFGBM), Yu. Wu, and Zhou (2012). developed the generalized hesitant fuzzy BM (GHFBM) with its application in the multicriteria group decision making.

The aggregation techniques have a great effect on the MCDM problems, and the aggregation operators have been broadly applied to MCDM. In a fuzzy environment, Chen and Tan (1994) developed a number of functions for measuring the extent to which each alternative is suitable regarding a set of criteria in MCDM. Hong and Choi (2000), used the maximum and minimum operations for the development of some approximate techniques of addressing the MCDM problems. In addition, the aggregation operators have been extended to intuitionistic fuzzy environment wherein IFSs (Atanassov, 1986) play the role of basic elements reflecting preference values or judgements of decision makers. Li (2005), designed a number of linear programming models and introduced corresponding decision making methods by means of IFSs. Liu and Wang (2007), proposed new series of score functions to be applied to the MCDM problems in accordance with the evaluation functions and the intuitionistic fuzzy point operators. Based on interval-valued IFSs, Chen, Wang, and Lu (2011) developed an approach of multi-criteria group decision making. However, comparatively, very few studies have been focused on the MCDM problems under the hesitant fuzzy environment. Moreover, in the process of decision making, hesitancy and uncertainty are generally considered as unavoidable problems. To express the evaluation information of decision makers more objectively, a number of tools have been developed in the literature, including fuzzy set (Zadeh, 1965), intuitionistic fuzzy set (Atanassov, 1986) and fuzzy multiset (Miyamoto, Liu, & Kunii, 2000; Yager, 1986), interval-valued fuzzy set (Zadeh, 1975), linguistic fuzzy set (Xu, 2004a; Xu, 2004b) and type-2 fuzzy set (Dubois & Prade, 1980).

## 3. Research methodology

This paper reviews the literature to recognize the articles that have been published in popular journals and provided the most important information to practitioners and researchers who investigate issues related to the FMCDM methods. To this end, an extensive search was carried out to find FMCDM in titles, abstracts, keywords, and research methodologies of the papers. This paper attempts to document the exponentially grown interest in the FMCDM methods and provide a state-of the-art review of the literature regarding the FMCDM applications and methodologies. According to a classification scheme, a reference repository, including a total of 403 published papers in more than 150 journals since 1994, has been established. The papers are classified in terms of the application areas, publication year, the authors' nationality, the journal's name, and other FMCDM methods. The present paper has three contributions: the development of a classification scheme with a focus on practical considerations, structurally reviewing the literature to guide the research on the FMCDM methods, and the identification of issues to be studied in future. Additionally, two new perspectives are taken into consideration to review the articles, namely categorization of the articles into four main fields (business, science, engineering, and technology) and examination of the type of study (FMCDM utilizing research, FMCDM developing research, FMCDM proposing research).

In particular, we targeted six library databases: Springer, ScienceDirect, ProQuest, Emerald, John Wiley, and Taylor & Francis, covering the most important journals in four main fields. Items such as doctoral dissertations, master's theses, textbooks, conference proceeding papers, and unpublished papers were ignored in this review. For this review, the primary data were collected from 1081 cited articles related to MCDM and fuzzy MCDM methods published since 1994. For choose 1081 scholarly journal papers we have used most of international journals specially related to DM methods. Some of journals cited in this review were, Expert Systems with Applications, Applied soft computing, Journal of Intelligent and Fuzzy Systems, Information sciences, International Journal of Production Research, Technological and Economic Development of Economy, European Journal of Operational Research, International Journal of Intelligent Systems, International Journal of Production Economics, Mathematics and Computers in Simulation, Fuzzy sets and systems, Omega, Knowledge-Based Systems, International Journal of Information Technology & Decision Making, Computers in Industry, etc. The majority of papers on FMCDM have been published since 1994; as a result, this year was chosen as the starting date for this study. It is noticeable that since online database access point is limited, some papers could not be downloaded; for that reason, they were overlooked in this survey. After reviewing each paper, the paper was summarized and highlighted. An article is taken into consideration in this review if it discusses thoroughly the application and development of FMCDM.

MCDM and fuzzy MCDM are the most well-known branches of decision making. In the decision making approach, the selection is made from amongst the decision alternatives that are described by their attributes. Over time, a large number of MCDM methods have been proposed, which are different in their theoretical background, the type of questions asked, and the type of obtained results. For a given problem, a number of methods have been particularly proposed, which cannot be applied to other problems. Several keywords and criteria should be taken into account for the selection of an MCDM method. In this review paper, for identify and finding the scholarly papers related to DM methods in mentioned databases we have searched several keywords, these keywords were, MCDM, fuzzy MCDM, DM (Decision Making), AHP & fuzzy AHP, TOPSIS & fuzzy TOPSIS, VIKOR & fuzzy VIKOR, ELECTRE & fuzzy ELECTRE, DEMATEL & fuzzy DEMATEL, ANP & fuzzy ANP, PROM-ETHEE, FWA, MCGDM, MCDA, OWA, SAW, FDM, Entropy & fuzzy Entropy, Hybrid FMCDM, Hybrid MCDM, and So on. We have searched these keywords because some articles have integrated MCDM methods with fuzzy numbers and mentioned in the research methodology parts. After primary search and collecting these scholarly articles, the articles related to fuzzy numbers and fuzzy set theory were selected.

#### 4. Results

## 4.1. Classifications and observations

This survey is based on a literature review and classification of international journal articles from 1994 to 2014. The majority of the journals are specialist journals in the fuzzy and MCDM journal. For the purpose of this part of paper, some of journals are listed based on publishers, and some journals (e.g., Emerald, John Wiley, Springer and Tylor and Francis expect ScienceDirect) are integrated based on their publishers.

Research on FMCDM continued and found many applications to different fields. Multi Criteria Decision Making (MCDM) provides strong decision making in domains where selection of the best alternative is highly complex. This paper reviews the main streams of considerations in multi criteria decision making theory and practice in detail, and it is mainly aimed to identify various applications and approaches and suggest approaches that can be most robustly and effectively used to identify the best alternative. This survey also addresses the problems in fuzzy multi criteria decision making techniques. MCDM method has been applied to many domains to choose the best alternatives. Where many criteria have come into existence, the best one can be obtained by analyzing different scopes of the criteria, weights of the criteria, and the selection of the optimum ones using any multi criteria decision making techniques.

This survey investigates the developments of various methods of FMCDM and its applications. In our daily life, many decisions are being made based on various criteria; thus the decision can be made by assigning weights to different criteria and all the weights are obtain from expert groups. It is important to determine the structure of the problem and explicitly evaluate multi criteria. For example, in building a nuclear power plant, certain decisions have been taken based on different criteria. There are not only very complex issues involving multi criteria, some criteria may have effect on some problems; however, to have an optimum solution, all alternatives must have common criteria, which clearly lead to more informed and better decisions. AHP method is used in the analysis of the health-safety and environmental risk assessment of refineries for the location of the power plant, the risk factors such as health-safety risk, technology risk, etc. (Rezaian & Jozi, 2012). TOPSIS has been applied to the selection of the best strategic technology for the fuel cell in the automotive industry (Sadeghzadeh & Salehi, 2011).

In all these works, different methods have been used for different applications where each method has its own characteristics in finding the best alternatives. The applications developed to solve multi choice problems and the selected FMCDM methods provide better performance in cases such as supplier chain management in business applications, safety assessment in marine engineering, watershed location, and urban distribution centers in public sectors.

#### 4.2. Field of category

Due to wide range of applications of fuzzy MCDM in the real world, there is a strong motivation to categorize these applications across several areas and particular sub-areas. The studies that have used FMCDM are categorized into three groups: FMCDM utilizing research, FMCDM developing research, and FMCDM proposing research. To identify the differences and similarities, the 403 papers were categorized into four fields: (1) science, management and business, engineering, and technology. In case of the papers that could fall into more than one category, based on the targeted audience defined by the paper's objectives, the best possible choice was selected. This ensured the absence of any duplication in the classification scheme. In the following sections, the papers are briefly presented and each topic is further summarized using tables corresponding to their sub-areas. In each table, the papers are summarized and highlighted according to their introductions, research methods, and the results of the study. Similarly, previous studies (e.g., Behzadian et al. (2012)) have categorized TOPSIS papers based on area of applications like manufacturing systems, supply chain issue, business and management, and so on.

#### 4.3. Engineering field

In this survey, engineering was considered as a field that has mostly used the fuzzy MCDM methods and approaches. In this paper, engineering fields involve several specific sub-fields, some recent applications of fuzzy MCDM approaches in including, civil engineering (Bagočius, et al., 2014; Dadelo et al., 2014; Yazdani-Chamzini et al., 2014), industrial engineering (Avikal, Jain, & Mishra, 2014; Avikal, Mishra, & Jain, 2014; Keskin, 2014; Mokhtari, Alinejad-Rokny, & Jalalifar, 2014), computer science (Herrmann & Herrmann, 2014; Kaya & Kahraman, 2014; Yazdani-Chamzini, 2014), electrical engineering (Kurt, 2014), and mechanical engineering (Azadnia, Saman, & Wong, 2014; Bairagi, Dey, Sarkar, & Sanyal, 2014; Hadi-Vencheh & Mohamadghasemi, 2014). Li, Jin, and Wang (2014), have used TOPSIS and QFD for selection and evaluation of KMS under fuzzy environment. Baležentis and Baležentis (2014), attempted to identify the role MOORA and MULIMOORA play in engineering areas for technological and economic development studies to be conducted in future. Wang and Wu (2014), have applied F-AHP, F-DEMATEL and FDM for evaluation of PLC. Kubler et al. (2014), employed F-AHP for selection of data in communicating material. Zavadskas, Antucheviciene, et al. (2014), developed WASPAS and results compared with COPRAS-IVIF-TOPSIS-IVIF and IFOWA for improve of WPM and WSM accuracy. Zavadskas, Turskis, et al. (2014), used ARAS-F and AHP for solve of problems in construction site selection in Eastern Baltic see. Bairagi et al. (2014), have employed F-AHP, F-VIKOR- F-TOPSIS and COPRAS-G for selection of robots. Mokhtarian, Sadi-nezhad, and Makui (2014a) and Mokhtarian, Sadi-nezhad, and Makui (2014b) proposed IVF-VIKOR and IVF-TOPSIS for solve problems in selection of facility location. Anojkumar, Ilangkumaran, and Sasirekha (2014), have employed FAHP-VIKOR, FAHP-PROMTHEE, FAHP-TOPSIS and FAHP-ELECTRE for selection of material in sugar industry. Rabbani, Zamani, Yazdani-Chamzini, and Zavadskas (2014), implemented of fuzzy COPRAS, ANP and SBSC for evaluation of performance in Iranian oil companies. Tadić, Zečević, and Krstić (2014), Utilized fuzzy ANP. fuzzy VIKOR and fuzzy DEMATEL for selection of city logistics. Hashemian, Behzadian, Samizadeh, and Ignatius (2014), have applied F-AHP and F-PROMETHEE for assessment of supplier process.Ghorabaee, Amiri, Sadaghiani, and Goodarzi (2014), used of F-COPRAS and interval type-2 for selection of supplier. Kucukvar, Gumus, Egilmez, and Tatari (2014), have applied F-ENTROPY and TOPSIS for ranking of life cycle sustainability performance. Altuntas, Selim, and Dereli (2014), utilized F-DEMATEL for solve problem in facility layout production systems. Yeh, Pai, and Liao (2014), have used F-AHP and F-DEMATEL for identifying the critical factors in NPD. Zare Mehrjerdi (2014), implemented of SAW, TOP-SIS and QSPM in selection of strategic system. Akdag, Kalaycı, Karagöz, Zülfikar, and Giz (2014), applied AHP, Yager's min-max and TOPSIS for evaluation of service quality in hospital. As an early study in engineering, Simões-Marques et al. (2000), evaluated technical and operational factors for repair of equipment under battle conditions by employed FMADM methodology.

In this survey, the researchers have reviewed a total of 217 article papers in different fields of engineering. According to the obtained results, industry engineering was ranked as the first field in the publication of papers among other fields. In addition, based on review findings, most of the sub-fields in industrial engineering were related to supply chain issues. This result was also confirmed by other scholars (e.g., Behzadian et al. (2012), indicating that, in the TOPSIS application, most of review papers were related to supply chain issues. The results of this field has shown that in the type of FMCDM utilizing research 165 (76%), 39 (17.97%) papers are in the type of FMCDM developing research, 11 papers (5.07%) were FMCDM proposing research. Moreover, in case of the publication

year, 42 papers were published in 2014, 32 papers in 2013, 25 papers in 2012, 39 papers in 2011, 26 papers in 2010, 18 papers in 2009, 13 papers in 2008, 10 papers in 2007, four papers in 2006, two papers in 2005, two papers in 2004, one paper in 2003, two papers in 2002 and one paper in 1994. In addition, in case of tools and approaches, 90 papers have employed fuzzy TOP-SIS and combined it with other methods, 102 papers have employed fuzzy AHP and combined it with other methods, 32 papers have employed fuzzy ANP and combined it with other methods, 16 papers have employed fuzzy VIKOR and combined it with other methods, 11 papers have employed ELECTRE and combined it with other methods, 13 papers have employed DEMATEL and combined it with other methods, 10 papers have employed PROMETHEE and combined it with other methods, and 32 papers have employed other combined tools and approaches such as ARAS. WASPAS.

#### 4.4. Management and business field

In the field of management and business, 122 studies have applied fuzzy MCDM tools and applications. In this category, there were some specific areas of management and business, including quality management (Amirzadeh & Shoorvarzy, 2013; Benítez, Martín, & Román, 2007; Tseng, 2009b), strategy management issues (Fouladgar, Yazdani-Chamzini, Lashgari, Zavadskas, & Turskis, 2012; Liou, Tzeng, Tsai, & Hsu, 2011), human resource (Dincer & Hacioglu, 2013), marketing (Lin, Lee, & Chen, 2009), risks management (Bayrakdaroğlu & Yalcın, 2013; Ganguly & Guin, 2013), information management (Aliei, Sazvar, & Ashrafi, 2012; Kahraman, Engin, Kabak, & Kaya, 2009), organizational performance (Cho, Lee, Ahn, & Hwang, 2012; Rostamzadeh & Sofian, 2011), knowledge management (Chen & Pang, 2010; Li, 2013), economic issues (Baležentis, Baležentis, & Misiunas, 2012), and other fields of management and business. Liu, Qin, Mao, and Zhang (2014), employed F-VIKOR for personal selection in HRM section of organization. Zamani, Rabbani, Yazdani-Chamzini, and Turskis (2014), have used ARAS-F and ANP for extension of brand in marketing strategies. Hajiagha, Mahdiraji, Zavadskas, and Hashemi (2014), developed F-VIKOR for solve problem in minimizing in ideal and anti-ideal solution. Zhang and Xu (2014), extended

Table 1Summary of applications of the DM techniques.

DM techniques	Frequency of application	Percentage (%)
Hybrid FMCDM	141	13.04
Hybrid MCDM	215	19.89
AHP	171	15.82
Fuzzy AHP	103	9.53
Fuzzy TOPSIS	79	7.31
TOPSIS	80	7.40
ANP	38	3.52
Fuzzy ANP	26	2.41
PROMETHEE	20	1.85
OWA	28	2.59
DEMATEL	30	2.78
VIKOR	22	2.04
MCGDM	16	1.48
ELECTRE	10	0.93
Fuzzy VIKOR	16	1.48
MCDA	7	0.65
Fuzzy ELECTRE	8	0.74
Fuzzy DEMATEL	9	0.83
Fuzzy PROMETHEE	5	0.46
FWA	1	0.09
Fuzzy ENTROPY	4	0.37
Other	52	4.81
Total	1081	100.00

	Ta	ble	2
--	----	-----	---

Distribution based on fuzzy AHP.

stribution bused on fuzzy fun.		
Authors	Type of study	Tools and approaches
Abdullah and Natib et al. (2014)	Approach	Droposed new scale of preference based on BUE ALD for desisting methods which
Abdullan and Najib et al. (2014)	Approach proposed	Proposed new scale of preference based on IVIF-AHP for decision making problems
Vahdat, Smith, and Amiri (2014)	Approach developed	Implemented of F-AHP for ranking risk factors for assessment of performance
Kubler et al. (2014)	Utilized approach	Employed F-AHP for selection of data in communicating material
van de Kaa, Rezaei, Kamp, and de Winter	Utilized approach	Used F-AHP for selection of photovoltaic technology system
(2014)		
Azadnia et al. (2014)	Utilized approach	Employed F-AHP and RBWF methods for selection of sustainable supplier
Dabbaghian, Hewage, Reza, Culver, and	Utilized approach	Used F-AHP for evaluation of performance in green roof systems
Sadig (2014)	I I	
Abdullah and Naiih (2014)	Approach presented	Presented a new IF-AHP for selection of energy technology
Dragović Turajlić Radojević and Petrović	Itilized approach	Applied F-AHP for solving problems in the web service
(2014)	otnized approach	Applied I All for solving problems in the web service
(2014) Cörkör Alberretr and Alertherretter (2014)	I Itilized engages	Employed E AUD for each using for exclusions desiring and merilesting startagies for
Gurbuz, Albayrak, aliu Alaybeyogiu (2014)	otnized approach	Employed F-AHP for analysis of marketing decisions and marketing strategies for
		development of new product
Kahraman, Süder, and Kaya (2014)	Utilized approach	Used F-AHP for assessment of investment in the health research
Roy, Ray, and Pradhan (2014)	Utilized approach	Applied F-AHP for selection of process of non-traditional machining
Najafi et al. (2014)	Utilized approach	Applied F-AHP for mineral prospectivity mapping in the eastern of Iran
Tansel İç, Yurdakul, and Dengiz (2013)	Utilized approach	Using a FAHP approach for robot selection
Vafai, Hadipour, and Hadipour (2013)	Utilized approach	Applied the FAHP in GIS environment for the sensitive coastal areas in Iran
Gülgen (2013)	Utilized approach	Employed FAHP weights in the road network hierarchy makes
Ganguly and Guin (2013)	Itilized approach	Using E-AHP for understanding of supply risks evaluation
Frtay Kahraman and Kaya (2013)	Itilized approach	Using FAHP and MACRETH for the evaluation of renewable energy
Paurakdaroğlu and Valcin (2012)	Utilized approach	Using FAHD to evaluating of operational risk factors of compareial banks
Jabiasha and Nauran (2012)	Utilized approach	Using PATH to Evaluating of operational fisk factors of commercial banks
isinzaka anu inguyen (2013)	Utilized approach	Application of FAMP for Identification of market training accounts
ru, chang, kao, chiu, and Lu (2013)	utilized approach	Using FAHP for identification of project training course
Jing, Chen, Zhang, and Peng (2013)	Approach developed	Developed hybrid FSAHP to aid decision making by incorporating fuzzy and stochastic
		uncertainty
Rezaei, Ortt, and Scholten (2013)	Approach developed	Utilized approach FAHP for evaluation of entrepreneurship orientation
Akadiri et al. (2013)	Utilized approach	Employed FAHP for selection of materials in the building projects
Chou, Pham, and Wang (2013)	Utilized approach	Proposed the FAHP regression-based simulation for bidding strategy to support decision-
		making
Tan, Aviso, Huelgas, and Promentilla	Approach developed	Proposed a FAHP variant to selection of process engineering problems
(2013)		
Calabrese Costa and Menichini (2013)	Utilized approach	Application of FAHP for assets of intellectual capital
Chou Liang and Chang (2012)	Approach developed	Presented a new FAUD method based on the concent of possibility evtent to solve the
chou, Liang, and Chang (2013)	Approach developed	EMCDM problem
He He Mar and Ye (2012)	TT:11	FINCDIN problem
He, Ho, Man, and Xu (2012)	Utilized approach	Proposed FAHP for solve the problems of multi-criteria transsnipment
Tsai and Lin (2012)	Utilized approach	Applied FAHP for the alternatives on the basis of the indices of the worldwide best
		indicators
Sevkli et al. (2012)	Utilized approach	Using of AHP for rank of SWOT factors
Zheng, Zhu, Tian, Chen, and Sun (2012)	Utilized approach	Applied FAHP for work assessment in humid and hot environments
Javanbarg et al. (2012)	Utilized approach	Implemented FAHP for evaluation and optimization of swarm particle
Gao and Hailu (2012)	Utilized approach	Used FAHP to evaluate of fishing recreational in the system of coral reef
Duru, Bulut, and Yoshida (2012)	Utilized approach	Used RS-FAHP to improve traditional AHP by employ regime switching model
Gao et al. (2012)	Approach developed	Developed hierarchical FAHP for distribution of energy-efficient clustering algorithm
Bilgen and Sen (2012)	Utilized approach	Using FAHP for selection of Six Sigma projects
Shakouri and Tayassoli (2012)	Approach presented	Composition of a FAHP and a FIS to differentiate between the criteria and attributed
Shakouri and Tavasson (2012)	hpproach presented	waightings to the criteria
Cao and Haily (2012)	Utilized approach	Using FAUD for evaluation approach to facilitate multi-criteria desision making
Gao and Hand (2012)	Utilized approach	Using FAILE for evaluation approach to facilitate multi-criteria decision making
	Utilized approach	Applied FAHP for performance measurement of service supply chain
Lin and TWU (2012)	ounzed approach	implemented of F-AHP for evaluation of fashion trend systems
Yueh-Hsiang Chen and Chao (2012)	Utilized approach	Applied of FAHP for vendor selection
Ju, Wang, and Liu (2012)	Utilized approach	Applied of FAHP and 2-tuple fuzzy linguistic to evaluation of response capacity
Samvedi, Jain, and Chan (2012)	Utilized approach	Integrated the FAHP and grey relational analysis approaches for the selection of a machine
		tools
Büyüközkan (2012)	Utilized approach	Applied a FAHP to rank the green suppliers
Das, Sarkar, and Ray (2012)	Utilized approach	Using COPRAS methodology and FAHP to measure relative performance of Indian technical
		institutions
Raiput, Milani, and Labun (2011)	Approach developed	Developed FAHP for statistical weighting and time dependency of decisions
Labib (2011)	Utilized approach	Comparison of FAHP and fuzzy logic for selection of supplier
Li Shi and Wang $(2011)$	Approach proposed	Pronosed FAHP for agile improvement system by using the action diagram and ECNN
Ly and Wang (2011)	Itilized approach	Application of FAHP to evaluation of project factors and criteria for ICP project
Zhang Cu Cu and Zhang (2011)	Utilized approach	Itilized approach FAHD and ICA for performance evaluation in building operation
Zhang, Gu, Gu, allu Zhalig (2011)	otilizeu appioaeli	conservation
Zaprovoinorbad and Mark-1-12 (2014)	Itilized annual	CUINCE VALIDIE
Zangoueineznad and Moshabaki (2011)	ounzed approach	Using r-AHP for analysis of university performance criteria
Iseng, Lin, and Chen (2011)	Utilized approach	Employed F-AHP for priority and identify of e-learning systems efficiency
Mıkaeil, Ataei, and Yousefi (2011)	Utilized approach	Application of FAHP for predicting of vibration rock sawing
Chiouy, Chou, and Yeh (2011)	Utilized approach	Using of F-AHP for ranking of supplier selection and evaluation
Zangoueinezhad, Azar, and Kazazi (2011)	Utilized approach	Using FAHP for supply chain competitiveness positioning
Ustundag and Serdar Kilinc (2011)	Utilized approach	Applied FAHP to select the best science park
Büyüközkan, Çifçi, and Güleryüz (2011)	Utilized approach	Using FAHP to evaluation of service quality framework
Padma and Balasubramanie (2011)	Utilized approach	Applied FAHP for the forecasting on neck and shoulder pains
Durán (2011)	Utilized approach	Applied FAHP for selection computer-aided maintenance management systems
An Chen and Baker (2011)	Utilized approach	Application FAHP in railway risk information system
Lee Mogi Lee and Kim (2011)	Utilized approach	Employed FAHD for rank weights of hydrogen energy technology in the sector of hydrogen
Lee, mogi, Lee, and Rill (2011)	omizeu approach	Employed main for rank weights of nydrogen energy technology in the sector of hydrogen

## Table 2 (continued)

Majundar (2010)Utilized approachUserUserLee, Mogi, Lee, Hui, and Kim (2010)Utilized approachUtilized approachUtilized approachShen et al. (2010)Utilized approachUtilized approachUtilized approachShen et al. (2010)Utilized approachUtilized approachCombined FAHP and max-min approach for pre-allocation and evaluation of operationsKahraman, Beskee, and Kaya (2010)Utilized approachApplication of FAHP and max-min approach for pre-allocation and evaluation of operationsKahraman, Reskee, and Kaya (2010)Utilized approachApplication of FAHP and GS for evaluation of operationsVadrevu, Eaturu, and Badarinath (2010)Utilized approachApplication of FAHP and GS for evaluation of operationsKahraman and Kaya (2010)Utilized approachApplied fuzzy (Jogic and AHP for classification inventoryChen, Wang, Chen, and Lee (2010)Utilized approachApplied fuzzy (Jogic and AHP for selection of RAB statisgic alliance partner"Chamodrakas, Batis, and Hoque (2010)Utilized approachApplied fuzzy (Jogic and AHP for descing for Wangottal)Jakowski, Biruk, and Bucon (2010)Utilized approachUsing VSH by integrating FCM algorith with AHP for vendor selection of mining methodJakowski, Biruk, and Bucon (2010)Utilized approachUsing GAHP for classification frame demandaturing systemsCabel (2009)Utilized approachUsing GAHP for classification frame demandaturing systemsJakowski, Biruk, and Bucon (2000)Utilized approachUsing GAHP for classification frame demandaturing systemsJakowski, Biruk, and Bucon (2000)	Authors	Type of study	Tools and approaches
Majundar (2010)Utilized approachUsel FAHP for raw material selection in industry of textile spinningLee, Moj, Lee, Hui, and Kim (2010)Utilized approachUtilized approachShen et al. (2010)Utilized approachUtilized approachSen, and Guar (2010)Utilized approachUtilized approachYadrevu, Eatru, and Badarinath (2010)Utilized approachApplication of FAHP for the selection among ERP outsourcing alternativesYadrevu, Eatru, and Badarinath (2010)Utilized approachApplication of FAHP for the selection among ERP outsourcing alternativesKahraman, Bekses, and Kaya (2010)Utilized approachApplication of FAHP for the selection among ERP outsourcing alternativesKahraman, Bekses, and Kaya (2010)Utilized approachApplication of FAHP for the selection of Supplet rpre-selectionChen, Wang, Chen, and Hogue (2010)Utilized approachApplication FAHP for selection of Supplet rise and factors for renewable energyAyag (2010)Utilized approachUtilized approachUsing USP hy integrating EVM algorithm with AHP for renewable energyAyag (2010)Utilized approachUsing USP hy integrating EVM algorithm with AHP for renewable energyAyag (2010)Utilized approachUsing USP hy integrating EVM algorithm with AHP for renewable energyJaskin, Mard Buoro (2010)Approach developedApplication of TAHP for facilitates defining criteria weights by aggregationGingrös, Frahkallogitu, and Keen (2009)Utilized approachUsing def TAHP for selection sourcesNaghadehi et al. (2009)Utilized approachCombined EXHP for selecting and Bactors for reversub			economy
Lee, Mogi, Lee, Hui, and Kim (2010)Utilized approachUtilized approachShen et al. (2010)Utilized approachUtilized approachShen et al. (2010)Utilized approachCombined FAHP and max—min approach for pre-allocation and evaluation of operatorsKahraman, Beskees, and Kaya (2010)Utilized approachApplication of FAHP and max—min approach for pre-allocation and evaluation of operatorsKahraman and Kaya (2010)Utilized approachApplication of FAHP and max—min method for supplier pre-selecction and evaluation of operatorsKahraman and Kaya (2010)Utilized approachApplication of FAHP and CBS for evaluation of operatorsKahraman and Kaya (2010)Utilized approachApplication of FAHP and CBS for evaluation of every policies indicatorsRezael and Dowlatshali (2010)Utilized approachApplication FAHP for selection of Rsb strategi alliance partner"Chamodrakas, Batis, and Martakos (2010)Utilized approachApplication of FAHP for selection of Subplier in electronicAnir Xadeh, Osanloo, and Ataci (2010)Utilized approachUsing VST by integrating FCM algorith with AHP for vendor selectionJakowski, Bruku, and Bucon (2010)Utilized approachUsing VST by integrating FCM algorith with AHP for vendor selection of FAHP for facilitates defining criteria weights by aggregationJagacowski, Bruku, and Bucon (2010)Utilized approachUsing Grazy FAHP approach Nicholas technique to selection of mining methodJagakowski, Bruku, and Bucon (2010)Utilized approachUsing Grazy FAHP approach Nicholas technique to selection of mining methodJagakowski, Bruku, and Bucon (2010)Utilized approach <td>Majumdar (2010)</td> <td>Utilized approach</td> <td>Used FAHP for raw material selection in industry of textile spinning</td>	Majumdar (2010)	Utilized approach	Used FAHP for raw material selection in industry of textile spinning
Shen et al. (2010)Utilized approachUtilized approach FAHP for solution of problems in Renewable Energy Development BillSen and Quari (2010)Utilized approachApplication of FAHP and max-min approach for pre-allocation and evaluation of operatorsSen, Sen, and Bashgil (2010)Utilized approachApplication of FAHP and max-min method for supplier pre-selectionVadreus, Eaturu, and Badamiath (2010)Utilized approachApplication of FAHP and max-min method for supplier pre-selectionKahnama, Berg, Call, CallUtilized approachApplication FAHP and APP for classification inventoryChen, Wang, Chen, and Lee (2010)Utilized approachApplication FAHP for selection of supplier pre-selectionChamodalasa, Batis, and Martakos (2010)Utilized approachApplication FAHP for selection of RABD strategic allance partner"Chamodalasa, Batis, and Martakos (2010)Utilized approachApplication FAHP for selection of MABD strategic allance partner"Anir Azadeh, Osando, and Actei (2010)Utilized approachUsing the FAHP to the selection of supprised to marketplacesAnir Azadeh, Osando, and Actei (2010)Approach developedApplication of FAHP for classification tracking use selectionJaskowski, Biruk, and Bucon (2010)Approach developedApplication of FAHP for designing the automated manufacturing systemsCalled (2009)Utilized approachCombined TRZ and FAHP for designing the automated manufacturing systemsJand Huang (2009)Utilized approachCombined TRZ and FAHP for designing the automated manufacturing systemsJand Huang (2009)Utilized approachAppliced FAHP for designing the automat	Lee, Mogi, Lee, Hui, and Kim (2010)	Utilized approach	Utilized approach FAHP and DEA for R&D efficiency performance in the national hydrogen
Shen et al. (2010)         Utilized approach         FAHP for solution of problems in Renewable Energy Development Bill           Sen and Carl         (2010)         Utilized approach         Cambrid EAHP and max-min approach for suppiler pre-election of operators           Kahraman, Beskese, and Kaya (2010)         Utilized approach         Application of FAHP for the selection among ERP outsourcing alternatives           Sen, Sen, and Kaya (2010)         Utilized approach         Application of FAHP and max-min method for suppiler pre-election           Kahrama and Kaya (2010)         Utilized approach         Suggested FAHP for the selection of forest risk fre           Kahrama and Kaya (2010)         Utilized approach         Applied fuzzy logic and AHP for selection of suppiler in electronic marketplaces           Ayag (2010)         Utilized approach         Application of FAHP for selection of suppiler in electronic marketplaces           Ayag (2010)         Utilized approach         Using VST by integrating ECM algorithm with AHP for vendors selection           Anir Azadeh, Osando, and Ataei (2010)         Utilized approach         Using fuzzy FAHP approach Nicholas technique to selection of maining method           Jakowski, Birk, Ka, Ad Bucon (2010)         Utilized approach         Using fuzzy FAHP approach Nicholas technique to selection of maining method           Jakowski, Birk, Ka, Ad Bucon (2010)         Utilized approach         Using fazzy FAHP approach Nicholas technique to selection of maining method     <			energy
Sen and Qinar (2010)Utilized approachCombined FAIP and max-min method for supplier pre-selection anong ERP outsouring alternativesKahrman, Beskese, and Kaya (2010)Utilized approachApplication of FAIP and CIS for evaluation of forest risk fireVadreux, Eatur, and Badarinath (2010)Utilized approachApplication of FAIP and CIS for evaluation of forest risk fireRezaei and Dowlatshahi (2010)Utilized approachSuggested FAIP for the selection of forest risk fireChen, Wang, Chen, and Lee (2010)Utilized approachApplication of FAIP for the selection of RBD strategic alliance partner*Chamodrakas, Batis, and Martakos (2010)Utilized approachApplication of FAIP for the selection of supplier in electronic marketplacesAyag (2010)Utilized approachApplication of FAIP for the selection application of FAIP for the selection adminism with AIP for vendor selectionAnnir Azadeh, Osanloo, and Ataei (2010)Utilized approachUsing fire FAIP for cestabilish the criteria and factors for renewable energyJaksowski, Biruk, and Bucon (2010)Utilized approachUsing for FAIP for compare ERP systems solutionsCebeci (2009)Utilized approachApplication of FAIP for designing the automated manufacturing systemsCebeci (2009)Utilized approachUsing of FAIP for designing the automate manufacturing systemsLi and Huang (2000)Utilized approachApplied FAIP for evaluation of service performance in foreign travel industryTang (2009)Utilized approachApplied FAIP for evaluation of Service performance in foreign travel industryLi and Usopo (2009)Utilized approachApplied FAIP for	Shen et al. (2010)	Utilized approach	Utilized approach FAHP for solution of problems in Renewable Energy Development Bill
Kahraman, Beskese, and Kaya (2010)Utilized approachApplication of FAHP for the selection among ERP outsourcing alternativesVadrevu, Faturu, and Badarinath (2010)Utilized approachIntegrated FAHP and GIS for evaluation of forest risk fireKahrama and Kaya (2010)Utilized approachIntegrated FAHP and GIS for cellulation of forest risk fireRezaei and Dowlatshahi (2010)Utilized approachApplied fuzzy logic and AHP for selection of supplier in electronic marketplacesChen, Wang, Chen, and Lee (2010)Utilized approachApplied the FAHP for selection of supplier in electronic marketplacesAya@ (2010)Utilized approachUsing VSF Hy integrating FCM algorithm with AHP for vendor selectionAli, Shii, Nine, Khan, and Hoque (2010)Utilized approachUsing fuzzy FAHP for he criteria and factors for renewable energyAnir Azach, Osanloo, and Ataei (2010)Utilized approachUsing fuzzy FAHP approach Nicholas technique to selection of minig methodJaskowski, Biruk, and Bacon (2010)Approach developedApplication of FAHP for designing ruteria weights by aggregationGengor, Schwald, Bacon (2010)Utilized approachUsing fazy FAHP for designing the automated manufacturing systemsCabeci (2009)Utilized approachUsing of FAHP for designing the automated manufacturing systemsSan et al. (2009)Utilized approachApplied FAHP for aclusiting the automated manufacturing systemsTang (2009)Utilized approachApplied FAHP for aclusiting the automated manufacturing systemsTang (2009)Utilized approachApplied FAHP for aclusiting and bacri of foreing trute industryTang	Şen and Çınar (2010)	Utilized approach	Combined FAHP and max-min approach for pre-allocation and evaluation of operators
Sen, Sen, and Başlaği (2010)       Utilized approach       Application of FAHP and GIS for evaluation of forest risk fire         Kahrman and Kaya (2010)       Utilized approach       Suggested FAHP for the ssteetion of forest risk fire         Rezari and Dowlatshahi (2010)       Utilized approach       Suggested FAHP for the ssteetion of RAD Strategic alliance partner,"         Chamodrakas, Batis, and Martakos (2010)       Utilized approach       Application FAHP for selection of Supplier in electronic marketplaces         All, Shil, Nine, Khan, and Hoque (2010)       Utilized approach       Application of FAHP for selection of supplier in electronic marketplaces         All, Shil, Nine, Khan, and Hoque (2010)       Utilized approach       Utilized approach       Using the FAHP for selection of Supplier in electronic marketplaces         Amir Azadeh, Osanloo, and Ataei (2010)       Utilized approach       Using fuzzy FAHP for personnel selection of mining method         Jakswaki, Biruk, and Bucon (2010)       Approach developed       Application of FAHP for designing the automated manufacturing systems         Ceheci (2009)       Utilized approach       Using of FAHP for designing the automated manufacturing systems         I and Huang (2009)       Utilized approach       Applied FAHP for selecting and best criteria audiground mining in Iran         Sun et al. (2009)       Utilized approach       Applied FAHP for designing the automated manufacturing systems         I and Huang (2009)	Kahraman, Beskese, and Kaya (2010)	Utilized approach	Application of FAHP for the selection among ERP outsourcing alternatives
Vadrew, Extury, and Badarinath (2010)Utilized approach Utilized approach Suggested FAHP for the selection of lorest risk tire Suggested FAHP for the selection of energy policies indicators Applied fuzzy logic and AHP for classification inventoryRezeai and Dowlatshahi (2010)Utilized approach Utilized approach Application FAHP for selection of Supplier in electronic marketplaces Applied the FAHP for the organization real-life productChamodrakas, Batis, and Marakos (2010)Utilized approach Utilized approach Applied the FAHP for the organization real-life productAli, Shii, Nine, Khan, and Hoque (2010)Utilized approach Utilized approachUsing the FAHP for stabilish the criteria and factors for renewable energy Using the FAHP for stabilish the criteria and factors for renewable energy Using the FAHP for scalitates defining criteria and factors for renewable energy Utilized approach Utilized approachJaslowski, Biruk, and Bucon (2010)Utilized approach Utilized approachUsing the FAHP for consparing FAM algorithm with AHP for designing the automated manufacturing systemsCebeci (2009)Utilized approach Utilized approachUsing of FAHP for seclution of for evaluation of dimensions Applied FAHP for determine weights for evaluation of dimensions to real industryTang (2009)Utilized approach Utilized approachApplied FAHP for evaluation of for evaluation of arcy in avain teoposed FAHP for evaluation of for evaluation of for evaluation of CP in Taivan teoposed FAHP for evaluation of averite enformance for evaluation of CP in Taivan teoposed FAHP for inventory desiret enformance for evaluation of averite enformance in a divel (2009)<	Şen, Şen, and Başlıgil (2010)	Utilized approach	Application of FAHP and max-min method for supplier pre-selection
Kahaman and Kaya (2010)Utilized approachSuggested FAHP for the selection of energy policies indicatorsRezaei and Dowlashahi (2010)Utilized approachApplied fizzy logic and AHP for classification inventoryChen, Wang, Chen, and Lee (2010)Utilized approachApplication FAHP for selection of Supplier in electronic marketplacesAyag (2010)Utilized approachApplication FAHP for selection of Supplier in electronic marketplacesAyag (2010)Utilized approachApplication FAHP for selection of supplier in electronic marketplacesAmir Azach, Osanloo, and Atai (2010)Utilized approachUsing WEH Py integrating FCM algorithm with AHP for vendors electrionGebeci (2009)Utilized approachUsing VEH Py integrating FCM algorithm with AHP for vendors electronic of mining methodJaskowski, Biruk, and Bucon (2010)Approach developedProposed FAHP for facilitates defining criteria weights by aggregationGebeci (2009)Utilized approachCombined TRIZ and FAHP for designing the automated manufacturing systemsLi and Huang (2009)Utilized approachUsing of FAHP for braselecting and besc ricrieria underground mining in IranSun et al. (2009)Utilized approachApplied FAHP for braselecting and besc ricrieria underground mining in IranLee (2009)Utilized approachApplied FAHP for braselecting and besc ricrieria underground mining in IranLee (2009)Utilized approachApplied FAHP for braselecting and besc ricrieria underground mining in IranLee (2009)Utilized approachApplied FAHP for braselecting and besc ricrieria anderground mining in IranLee (2009)Ut	Vadrevu, Eaturu, and Badarinath (2010)	Utilized approach	Integrated FAHP and GIS for evaluation of forest risk fire
Rezeit and Jowlatshahi (2010)Utilized approach Utilized approach Application of FAHP for selection of S&D strategic alliance partner"Chen, Wang, Chen, and Lee (2010)Utilized approach Utilized approach Application of FAHP for selection of S&D strategic alliance partner"Avag (2010)Utilized approach Utilized approach Heo, Kim, and Boo (2010)Utilized approach Utilized approach Applied FAHP for determine weighing the automated manufacturing systems Applied FAHP for determine weighing the automated manufacturing systems Applied FAHP for determine weighing the automated manufacturing systems Applied FAHP for successful implementation and adoption of C in Taiwan Ee (2009) Utilized approach Applied FAHP for successful implementation and adoption of C in Taiwan Ee (2009) Utilized approach Applied FAHP for invertory classification system Evaluated of buyer-supplier forms based on FAHP ecoremine weights for automating of dimensions Applied FAHP for successful implementation and adoption of C in Taiwan Ee (2009) Utilized approach Chan, and Chin (2009) Utilized approach Chang, Wu, and Chen (2008) Utilized approach Chang, Gu, and Hsu (2008) Utilized approach Chang, Gu, and Hsu (2008) Utilized approach Chang, Gu, and Hsu (2008) Utilized approach Chan, Gu	Kahraman and Kaya (2010)	Utilized approach	Suggested FAHP for the selection of energy policies indicators
Chen, Wang, Chen, and Lée (2010)Utilized approach Utilized approach Application of FAHP for selection of supplier in electronic marketplaces Applied the FAHP for selection of supplier in electronic marketplaces Applied the FAHP for selection of supplier in electronic marketplaces Applied the FAHP for selection of supplier in electronic marketplaces (Ling daporach developed Approach developed Application of FAHP for selection of supplier in electronic marketplaces (Ling daporach developed Application of FAHP for selection of supplier in electronic marketplaces (Ling daporach developed (Ling daporach developed (Ling daporach developed) (Ling daporach developed) (Li	Rezaei and Dowlatshahi (2010)	Utilized approach	Applied fuzzy logic and AHP for classification inventory
Chamodrakes, Bats, and Martakos (2010)Utilized approachApplication of FAHP for selection of supplier in electronic marketplacesAyag (2010)Utilized approachApplication of FAHP for selection of supplier in electronic marketplacesAyag (2010)Utilized approachUsing WSFI by integrating FCM algorithm with AHP for vendor selectionAmir Azadeh, Osanloo, and Ataei (2010)Utilized approachUsing the FAHP for selection of the criteria and factors for renewable energyJaskowski, Biruk, and Bucon (2010)Utilized approachUsing the FAHP for facilitates defining criteria weights by aggregationChebcei (2009)Utilized approachProposed FAHP for personnel selection systemCobiner (2009)Utilized approachOmbined TRIZ and FAHP for designing the automated manufacturing systemsNaghadehi et al. (2009)Utilized approachApplied FAHP for evaluation of service performance in foreign travel industryTang (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachUtilized approachChang, Hung, Li, and Hui (2008)Utilized approachEmployed FAHP for use in managing of strategies industrial developmentChang, Hung, Li, and Hui (2008)Utilized approachEmployed FAHP for usel (ados)Chang, Gu, and Pai (2008)Utilized approachEmployed FAHP for usel (ados)Chang, Gu, an	Chen, Wang, Chen, and Lee (2010)	Utilized approach	Application FAHP for selection of R&D strategic alliance partner"
Ayag (2010)Utilized approachApplied the FAHP for the organization real-life productAli, Shi, Nine, Khan, and Hoque (2010)Utilized approachUsing VSF by integrating FCM algorithm with AHP for vendor selectionAmir Azadek, Osanloo, and Ataei (2010)Approach developedApplication of FAHP for scabilish the criteria and factors for renewable energyJaskowski, Biruk, and Bucon (2010)Approach developedApplication of FAHP for compare ERP systems solutionsCebeci (2009)Utilized approachDising VSF Parsones selection systemCebeci (2009)Utilized approachCombined TRIZ and FAHP for designing the automated manufacturing systemsNaghadehi et al. (2009)Utilized approachApplied FAHP for compare ERP systems solutionsSun et al. (2009)Utilized approachApplied FAHP for designing the automated manufacturing systemsTang (2009)Utilized approachApplied FAHP for devaluation of service performance in foreign travel industryTang (2009)Utilized approachApplied FAHP for evaluation of service performance in foreign travel industryTang (2009)Utilized approachApplied FAHP for tacking the imprecision of SHAP techniqueLee (2009)Utilized approachApplied FAHP for tacking the imprecision of silicon wafer slicing and uncertaintyChang, Wu, and Chen (2008)Utilized approachLing AHP for toruse in managing of strategies industrial developmentChang, Hung, Li, and Hsu (2008)Utilized approachEmployed FAHP for designing of strategies industrial developmentDagdeviern and Yuksel (2008)Utilized approachEmployed FAHP for designing of strategi	Chamodrakas, Batis, and Martakos (2010)	Utilized approach	Application of FAHP for selection of supplier in electronic marketplaces
Alt, Shil, Nin, Khe, Khan, and Hoque (2010)Utilized approachUsing VSH by integrating FCM algorithm with AHP for vendor selectionHeo, Kim, and Boo (2010)Utilized approachUsing the FAHP to establish the criteria and factors for renewable energyJaskowski, Biruk, and Bucon (2010)Approach developedApplication of FAHP for facilitates defining criteria weights by aggregationGingor, Serhadhoglu, and Kesen (2009)Utilized approachUsing of FAHP for designing the automated manufacturing systemsCebeci (2009)Utilized approachUsing of FAHP for designing the automated manufacturing systemsNaghadehi et al. (2009)Utilized approachUsing of FAHP for designing the automated manufacturing systemsSun et al. (2009)Utilized approachApplied FAHP for designing the automated manufacturing systemsTang (2009)Utilized approachApplied FAHP for designing the automated manufacturing systemsIan et al. (2009)Utilized approachApplied FAHP for designing the automated manufacturing systemsIan et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranIseng, Lin, and Chu (2009)Utilized approachApplied FAHP for sultation of service performance in foreign travel industryIseng, Chu, and Chen (2008)Utilized approachUsing FAHP for sultation of service performance in foreign travel industryChang, Mu, Li, and Hsu (2008)Utilized approachUsing FAHP for toenting the imprecision of silicon on systemDağdevinen and Yüksel (2008)Utilized approachUsing FAHP for toenting the imprecision of silicon wafer slicing and uncertainty<	Ayağ (2010)	Utilized approach	Applied the FAHP for the organization real-life product
Heo, Kim, and Boo (2010)Utilized approachUsing the FAHP for selections of minim methodJaskowski, Biruk, and Bucon (2010)Approach developedApplication of FAHP for facilitates defining criteria weights by aggregationGungor, Serhadhoglu, and Kesen (2009)Utilized approachUsing the FAHP for facilitates defining criteria underground mining in IranNaghadehi et al. (2009)Utilized approachUsing of FAHP for selecting and hest criteria underground mining in IranSun et al. (2009)Utilized approachCombined TRIZ and FAHP for selecting and best criteria underground mining in IranSun et al. (2009)Utilized approachUsed FAHP for selecting and best criteria underground mining in IranSun et al. (2009)Utilized approachUsed FAHP for selecting of dimensionsTang (2009)Utilized approachApplied FAHP for successful implementation for acrospace companyLin et al. (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachApplied FAHP for tor successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachUsing AHP, fuzzy multi-criteria and DEX to determine the business performanceChang, Wu, and Chen (2008)Utilized approachUsing FAHP for tor use in managing of strategies industrial developmentDagdeviren and Yuksel (2008)Utilized approachUsing FAHP for tor use in managing of strategies industrial developmentLag, Hung, Li, and Hua (2008)Utilized approachUsing FAHP for derive pairwise comparison stilticatorsChang, Au, Guo (2008)Utilized approach <td>Ali, Shil, Nine, Khan, and Hoque (2010)</td> <td>Utilized approach</td> <td>Using VSFI by integrating FCM algorithm with AHP for vendor selection</td>	Ali, Shil, Nine, Khan, and Hoque (2010)	Utilized approach	Using VSFI by integrating FCM algorithm with AHP for vendor selection
Amin Azadeh, Osanioo, and Atae (2010)Ultilized approachUsing fuzzy relay approachUsing fuzzy relay approachJaskowski, Biruk, and Bucon (2010)Approach developedApplication of FAHP for compare ERP system solutionsCebeci (2009)Utilized approachUsing of FAHP for compare ERP system solutionsLi and Huang (2009)Utilized approachUsing of FAHP for selection systemNaghadehi et al. (2009)Utilized approachUsing of FAHP for selecting and best criteria underground mining in franSun et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in franTang (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in franLin et al. (2009)Utilized approachApplied FAHP for sevaluating of inforeign travel industryLee (2009)Utilized approachApplied FAHP for sevaluating of Intaven in foreign travel industryLee (2009)Utilized approachApplied FAHP for inventory classification systemLee (2009)Utilized approachUsing FAHP for inventory classification systemChang, Wu, and Chen (2008)Utilized approachUsing FAHP to tackling the imprecision of slicon wafer slicing and uncertaintyChang, Guo, and Pia (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachFHP approach applied to evaluating of Stategies industrial developmentLang, Guo, and Pia (2008)Utilized approachFHP approach applied to evaluating of Stategies industrial developmentLee, Chen, and Yuk (2007)Utiliz	Heo, Kim, and Boo (2010)	Utilized approach	Using the FAHP to establish the criteria and factors for renewable energy
Jaskowski, Bruk, and Bucon (2010)Approach developed ApproachApproach developed ApproachApproach developed ApproachApproach developed Proposed FAHP for resonnel selection systemCebeci (2009)Utilized approachUsing of FAHP for compare ERP systems solutionsLi and Huang (2009)Utilized approachCombined TRIZ and FAHP for designing the automated manufacturing systemsNaghadehi et al. (2009)Utilized approachApplied FAHP for budget allocation of aerospace companyLin et al. (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachEvaluated of buyer-supplier forms based on FAHP for classification systemTseng, Chiu, and Chen (2008)Utilized approachUsing FAHP for reveluent of silicon wafer slicing and uncertaintyCakir and Canoblat (2008)Utilized approachUsing FAHP for isventry classification systemChang, Hung, Li, and Hsu (2008)Utilized approachEmployed FAHP to determine FBR in work systemsDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to developed and fuscel systemsChang, Guo, and Hui (2008)Utilized approachFAHP aproach applied to evaluation of SIC performance indicatorsChing, Guo, and Hui (2008)Utilized approachEmployed FAHP to determine FBR in work systemsQuo, and Hui (2008)Utilized approachEmployed FAHP to developedChang, Guo, and Hui (2007)Utilized approachUsing FAHP to developed	Amir Azadeh, Usanloo, and Ataei (2010)	Utilized approach	Using fuzzy FAHP approach Nicholas technique to selection of mining method
Cubor CoopUtilized approachProposed FAHP for personnel selection systemCebeci (2009)Utilized approachUsing of FAHP for compare ERP systems solutionsNaghadehi et al. (2009)Utilized approachCombined TRIZ and FAHP for designing the automated manufacturing systemsNaghadehi et al. (2009)Utilized approachCombined TRIZ and FAHP for designing the automated manufacturing systemsTang (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranLin et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranIn et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranLin et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranLin et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranLee, Clena, Wu, and Chen (2008)Utilized approachUsing FAHP for successful implementation and adoption of CP in TaiwanCharg, Hung, Li, and Hsu (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Pai (2008)Utilized approachEmployed FAHP to for selection of gible or valuation of BSC performance indicatorsChang, Guo, and Pai (2008)Utilized approachEmployed FAHP to for selecting and uncertaintyWang, and Chen (2008)Utilized approachEmployed FA	Jaskowski, Biruk, and Bucon (2010)	Approach developed	Application of FAHP for facilitates defining criteria weights by aggregation
Cebeci (2009)Utilized approachUsing of HAPP for compare EXP systems solutionsLi and Huang (2009)Utilized approachCombined TRIZ and FAPP for designing the automated manufacturing systemsNaghadehi et al. (2009)Utilized approachApplied FAIP for determine weights for evaluating of dimensionsTang (2009)Utilized approachApplied FAIP for determine weights for evaluating of dimensionsLin et al. (2009)Utilized approachApplied FAIP for determine weights for evaluating of dimensionsLin et al. (2009)Utilized approachApplied FAIP for evaluation of service performance in foreign travel industryTseng, Lin, and Chu (2009)Utilized approachEvaluated of buyer-supplier forms based on FAIP techniqueLee (2009)Utilized approachUsing FAIP for zavecessful implementation and adoption of CP in TaiwanLee (2009)Utilized approachUsing FAIP for tackling the imprecision of silicon wafer slicing and uncertaintyChang, Wu, and Chen (2008)Utilized approachUsing FAIP for inventory classification systemChang, Hung, Li, and Hsu (2008)Utilized approachEmployed FAIP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAIP hand its applicationsVaag, Luo, and Hai (2008)Utilized approachUtilized Approach evelopedVaag, Luo, and Hai (2008)Utilized approachEmployed FAIP hand its applicationsVaag, Luo, and Hai (2008)Utilized approachEmployed FAIP hand its applicationsWaag, Luo, and Hai (2007)Utilized approachUsing FAIP hand its applica	Gungor, Serhadlioglu, and Kesen (2009)	Utilized approach	Proposed FAHP for personnel selection system
Li and Huang (2009)Utilized approachCombined NiZ and PAHP for designing the automated manufacturing systemsNaghadehi et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranSun et al. (2009)Utilized approachApplied FAHP for budget allocation for aerospace companyLin et al. (2009)Utilized approachApplied FAHP for budget allocation for aerospace companyLin et al. (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachUsing AHP, forzy multi-criteria and DEA to determine the business performanceChang, Wu, and Chen (2009)Utilized approachUsing AHP, forzy multi-criteria and DEA to determine the business performanceChang, Wu, and Chen (2008)Utilized approachUsing AHP for inventory classification systemChang, Luo, Li, and Hsu (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachUsing FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachUsing FAHP to determine fevels of WIP acceptable setWang and Chen (2008)Utilized approachUsing FAHP to detrive pairwise comparison matricesPan (2008)Approach developedUsing of FEAHP for selecting on the suitable bridge constructionChan and Kumar (2007)Utilized approachCombined FAHP for selecting of the suitable bridge constructionC	Cebeci (2009)	Utilized approach	Using of FAHP for compare ERP systems solutions
Nagnadehi et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranSun et al. (2009)Utilized approachUsed FAHP for selecting and best criteria underground mining in IranTang (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranLin et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranLin et al. (2009)Utilized approachApplied FAHP for selecting and best criteria underground mining in IranLee (2009)Utilized approachApplied FAHP for successful implementation and adoption of CP in TaiwanLee (2009)Utilized approachEvaluated of buyer-supplier forms based on FAHP techniqueTseng, Chiu, and Chen (2008)Utilized approachEmployed FAHP for inventory classification systemChang, Wu, and Chen (2008)Utilized approachEmployed FAHP for to for use in managing of strategies industrial developmentDagdeviren and Yuksel (2008)Utilized approachEmployed FAHP for determine the vels of WIP acceptable setLee, Chen, and Chang (2008)Utilized approachEmployed FAHP and its applicationsWang, Luo, and Pai (2008)Utilized approachUsing FAHP for selection of global supplierWang and Chen (2008)Utilized approachUsing FAHP for selection of global supplierPan (2008)Approach developedUsing of FAHP and its applicationsWang, Luo, and Huu (2007)Approach developedUsing of FAHP for selection of global supplierWang, Luo, and Chen (2007)Utilized approachUsing of FAHP	Li and Huang (2009)	Utilized approach	Combined TRIZ and FAHP for designing the automated manufacturing systems
Suit et al. (2009)Utilized approach Applied FAHP for determine Weights for evaluating of dimensionsTang (2009)Utilized approach Applied FAHP for evaluation of service performance in foreign travel industry Applied FAHP for evaluation of service performance in foreign travel industryLin et al. (2009)Utilized approach Utilized approachApplied FAHP for evaluation of service performance in foreign travel industry Applied FAHP for successful implementation and adoption of CP in Taiwan Lee (2009)Lee (2009)Utilized approach Utilized approachEvaluated of buyer-supplier forms based on FAHP the chniqueTseng, Chiu, and Chen (2008)Utilized approach Utilized approachUsing FAHP for inventory classification systemChang, Hung, Li, and Hsu (2008)Utilized approach Utilized approachEmployed FAHP to for use in managing of strategies industrial development Employed FAHP to determine FBR in work systemsLee, Chen, and Chen (2008)Utilized approach Utilized approachEmployed FAHP and its applicationsWang, Luo, and Hua (2008)Utilized approach Utilized approachUtilized F-AHP to determine FBR in work systemsVang, Luo, and Hua (2008)Utilized approach Utilized approachUsing FAHP approach for selection of the suitable bridge construction Using FAHP to derive pairwise comparison matricesPan (2008)Approach developed Utilized approachUsing of FEAHP for selection of KPS development Using FAHP to selection of KPS development toolsWang, Luo, and Wu (2007)Utilized approachUtilized approach Approach developedWang, Chu, and Wu (2007)Utilized approach Utilized approachEmployed	Nagnadeni et al. (2009)	Utilized approach	Applied FAHP for selecting and best criteria underground mining in Iran
Ining (2009)Utilized approachApplied FAHP and AIP for budget allocation for aerospace companyLin et al. (2009)Utilized approachApplied FAHP for evaluation of service performance in foreign travel industryTseng, Lin, and Chiu (2009)Utilized approachEvaluated of buyer-supplier forms based on FAHP techniqueLee (2009)Utilized approachEvaluated of buyer-supplier forms based on FAHP techniqueTseng, Chiu, and Chen (2008)Utilized approachEvaluated of buyer-supplier forms based on FAHP techniqueChang, Wu, and Chen (2008)Utilized approachEmployed FAHP for tackling the imprecision of silicon wafer slicing and uncertaintyCakir and Canbolat (2008)Utilized approachEmployed FAHP for tackling the imprecision of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachUsing FAHP to determine FBR in work systemsLee, Chen, and Kumar (2007)Utilized approachUsing FAHP to determine FBR in work systemsWang, Luo, and Hua (2008)ReviewReviewWang, Luo, and Kumar (2007)Approach developedEmployed FAHP and its applicationsWang, Chen, and Wu (2007)Utilized approachUsing FAHP for selection of global supplierMa, Chen, and Wu (2007)Utilized approachEmployed FAHP for selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachCombined	Sun et al. (2009)	Utilized approach	Used FAHP for determine weights for evaluating of dimensions
Line et al. (2009)Utilized approachApplied FAHP for evaluation of service performance in foreign index indexityLee (2009)Utilized approachSing AHP, fuzzy multi-criteria and DEA to determine the business performanceChang, Wu, and Chen (2008)Utilized approachUsing AHP, fuzzy multi-criteria and DEA to determine the business performanceCakir and Canbolat (2008)Utilized approachEmployed FAHP for inventory classification systemChang, Hung, Li, and Hsu (2008)Utilized approachEmployed FAHP for inventory classification systemDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachFAHP approach applied to evaluation of BSC performance indicatorsChiang, Guo, and Hua (2008)Utilized approachUtilized FAHP to determine levels of WIP acceptable setWang, Luo, and Hua (2008)ReviewReviewed on FAHP and its applicationsWang Luo, and Hua (2007)Utilized approachUsing FAHP for selection of slobal supplierMan (2007)Utilized approachEmployed FAHP for selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachProposed FAHP for evaluation of KPS development toolsKang and Wu (2007)Utilized app	lang (2009) Lin et el. (2000)	Utilized approach	Applied FAHP and AIP for budget allocation for aerospace company
Tseng, Lin, and Chin (2009)Utilized approachKappined FAFIP for Successing implementation and adoption of CP in TalwainLee (2009)Utilized approachEvaluated of buyer-supplier forms based on FAHP techniqueTseng, Chiu, and Chen (2008)Utilized approachUsing AHP, fuzzy multi-criteria and DEA to determine the business performanceChang, Wu, and Chen (2008)Utilized approachUsing FAHP for tackling the imprecision of silicon wafer slicing and uncertaintyCakir and Canbolat (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachFAHP approach applied to evaluation of BSC performance indicatorsWang, Luo, and Hua (2008)Utilized approachHAHP to determine levels of WIP acceptable setWang and Chen (2008)Utilized approachUsing FAHP to deterve pairwise comparison matricesPan (2008)Approach developedEmployed FAHP to selection of global supplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP to selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUsing FAHP for NPD environment evaluationKayag (2005)Utilized approachUsing FAHP for NPD	Lin et al. (2009) Teens Lin and Chiu (2000)	Utilized approach	Applied FAHP for evaluation of service performance in foreign travel industry
LeeChinzed approachEvaluated of bdyet-supplier forms based on PAPP techniqueTseng, Chu, and Chen (2009)Utilized approachUsing AHP, fuzzy multi-criteria and DEA to determine the business performanceChang, Wu, and Chen (2008)Utilized approachEmployed FAHP for tackling the imprecision of silicon wafer slicing and uncertaintyCakir and Canbolat (2008)Utilized approachEmployed FAHP for inventory classification systemDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachFAHP to determine levels of WIP acceptable setWang, Luo, and Hua (2008)ReviewReviewed on FAHP and its applicationsWang and Chen (2008)Utilized approachUsing FAHP to determine levels of WIP acceptable setWang, Chu, and Wu (2007)Approach developedEmployed FAHP for selection of the suitable bridge constructionChan and Kumar (2007)Approach developedUsing of FEAHP for selection of global supplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixes	Iseng, Lin, and Chiu (2009)	Utilized approach	Applied FAHP for successful implementation and adoption of CP in Taiwan
Tseng, Chu, and Chen (2009)Onized approachOsing APP, fu22y indir-criteria and DEA to determine the Dusines's performanceChang, Wu, and Chen (2008)Utilized approachEmployed FAHP for tackling the imprecision of silicon wafer slicing and uncertaintyDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP for tackling the imprecision of BSC performance indicatorsLee, Chen, and Chang (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachEAPP to determine levels of WIP acceptable setWang, Luo, and Hua (2008)Utilized approachUtilized F-AHP to determine levels of WIP acceptable setWang, Luo, and Hua (2008)Utilized approachUsing FAHP approach for selection of the suitable bridge constructionPan (2008)Utilized approachUsing FAHP to derive pairwise comparison matricesPan (2008)Utilized approachEmployed FAHP for selection of the suitable bridge constructionChan, and Kumar (2007)Approach developedEmployed FAHP to selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachCombined FAHP for realuation of technology criteria selectionBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUsing FAHP for cla	Lee (2009) Teena Chiu and Chan (2000)	Utilized approach	Evaluated of Duyer-supplier forms based on FAHP technique
Chaing, Wu, and Cheff (2008)Utilized approachEmployed FAPP for tacking the imprecision of sincon water sincing and uncertaintyCakir and Canbolat (2008)Utilized approachUsing FAHP for inventory classification systemDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachFAHP aproach applied to evaluation of BSC performance indicatorsLee, Chen, and Chang (2008)Utilized approachFAHP aproach applied to evaluation of BSC performance indicatorsWang, Luo, and Hua (2008)ReviewReviewed on FAHP and its applicationsWang and Chen (2008)Utilized approachUsing FAHP to determine levels of WIP acceptable setPan (2008)Utilized approachUsing FAHP to derive pairwise comparison matricesPan (2008)Approach developedEmployed FAHP for selection of global supplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP for selecting the best strategies of maintenanceWang and Lee (2007)Utilized approachCombined FAHP and image compositing technique for choosing the optimum productKreng and Wu (2007)Utilized approachCombined FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskos, Chan, Chan, and Humphreys (2006)Utilized approachUsing FAHP for classification	Change Wile and Chan (2009)	Utilized approach	Using AHP, luzzy multi-criteria and DEA to determine the business performance
Calar and Canobia (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachEmployed FAHP to determine levels of WIP acceptable setWang, Luo, and Pai (2008)Utilized approachUtilized F-AHP to determine levels of WIP acceptable setWang, Luo, and Hua (2008)ReviewReviewed on FAHP and its applicationsWang and Chen (2008)Utilized approachUsing FAHP to derive pairwise comparison matricesPan (2008)Approach developedEmployed FAHP to selection of global supplierChan and Kumar (2007)Utilized approachEmployed FAHP to selection of global supplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKarg and Lee (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUsing FAHP for classification of technology criteria selectionKapoor and Tak (2005)Utilized approachUsing FAHP for evaluation of robot problemsKapoor and Tak (2005)Utilized approachUsing FAHP for evaluation of robot problemsCheng, Chen, and Yu (2005)Utilized approachUsing FAHP for evaluating of strategy in broadband	Calding, Wu, and Cambolat (2008)	Utilized approach	Ling FALL for inventory classification system
Chang, Hung, H, and Hisu (2008)Utilized approachEmployed FAHP to for use in managing of strategies industrial developmentDağdeviren and Yüksel (2008)Utilized approachEmployed FAHP to determine FBR in work systemsLee, Chen, and Chang (2008)Utilized approachFAHP approach applied to evaluation of BSC performance indicatorsWang, Luo, and Pai (2008)Utilized approachUtilized approachWang and Chen (2008)Utilized approachUsing FAHP to determine levels of WIP acceptable setWang, Chu, and Wu (2007)Approach developedEmployed FAHP to for selection of the suitable bridge constructionMa, Chen, and Wu (2007)Utilized approachEmployed FAHP to selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for evaluation of tobt problemsCheng, Chen, and Yu (2005)Utilized approachUsing FAHP for evaluation of tobt problemsCheng, Chen, and Yu (2005)Utilized approachUsing FAHP for evaluation of strategy in broadband services	Chang Hung Li and Hey (2008)	Utilized approach	Employed FAUD to for use in managing of strategies industrial development
Diggevinent and runser (2008)Utilized approachEndposed FAIP to determine rob in work systemsLee, Chen, and Chang (2008)Utilized approachFAIP approach approach approach approachFAIP approach approach approach approachWang, Luo, and Hua (2008)Utilized approachUtilized F-AHP to determine levels of WIP acceptable setWang and Chen (2008)Utilized approachUsing FAHP to derive pairwise comparison matricesPan (2008)Approach developedEmployed FAHP approach for selection of the suitable bridge constructionChan and Kumar (2007)Approach developedUsing of FEAHP for selecting the best strategies of maintenanceWang, Chu, and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKreng and Wu (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachVising FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for lassification of technology criteria selectionKapoor and Tak (2005)Utilized approachUsing FAHP for evaluation of FAHP for evaluationKapoor and Tak (2005)Utilized approachUsing FAHP for evaluation of strategy in broadband servicesCheng, Chen, and Yu (2005)Utilized approachEmployed FAHP for evaluating of strategy in broadband services	Dağdoviron and Vüksel (2008)	Utilized approach	Employed FALIP to for use in managing of strategies mutistrial development
Lited productParticipationParticipationChiang, Guo, and Pai (2008)Utilized approachUtilized approachWang, Luo, and Hua (2008)ReviewReviewed on FAHP and its applicationsWang and Chen (2008)Utilized approachUsing FAHP to derive pairwise comparison matricesPan (2008)Approach developedEmployed FAHP approach for selection of the suitable bridge constructionChan and Kumar (2007)Approach developedUsing of FEAHP for selection of global supplierWang, Chu, and Wu (2007)Utilized approachCombined FAHP and image compositing technique for choosing the optimum productKreng and Wu (2007)Utilized approachCombined FAHP not evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachVilized approachChan, Chan, Chan, and Humphreys (2006)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for classification of rebot problemsKapoor and Tak (2005)Utilized approachUsing FAHP for evaluation of sobe problemsCheng, Chen, and Yu (2005)Utilized approachEmployed FAHP for evaluation of strategy in broadband services	Log Chop and Chang (2008)	Utilized approach	Employed PATIF to determine PDK in work systems
Chang, Guo, and Par (2003)Durized approachOutized approachOutized approachWang, Luo, and Hua (2008)Utilized approachReviewed on FAHP and its applicationsPan (2008)Utilized approachUsing FAHP to derive pairwise comparison matricesPan (2008)Approach developedUsing of FEAHP for selection of the suitable bridge constructionChan and Kumar (2007)Approach developedUsing of FEAHP for selection of global supplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP to selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachCombined FAHP and image compositing technique for choosing the optimum productKreng and Lee (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsBozbura, Beskese, and Kahraman (2007)Utilized approachProposed FAHP for measurement indicators of human capitalChan, Chan, Chan, and Humphreys (2006)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for NPD environment evaluationKapoor and Tak (2005)Utilized approachLyne classification of robot problemsCheng, Chen, and Yu (2005)Utilized approachEmployed FAHP for evaluating of strategy in broadband services	Chiang Cuo and Pai (2008)	Utilized approach	Itilized E AUD to determine levels of WID acceptable set
Wang, Euo, and Hud (2000)KerterKerterKerterKerterWang and Chen (2008)Utilized approachUsing FAHP to derive pairwise comparison matricesPan (2008)Approach developedUsing of FEAHP approach for selection of the suitable bridge constructionChan and Kumar (2007)Approach developedUsing of FEAHP for selection of global supplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP to selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachCombined FAHP and image compositing technique for choosing the optimum productKreng and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for NPD environment evaluationKapoor and Tak (2005)Utilized approachUsing FAHP for evaluating of strategy in broadband servicesCheng, Chen, and Yu (2005)Utilized approachEmployed FAHP for evaluating of strategy in broadband services	Wang Luo and Hua $(2008)$		Reviewed on FAHD and its applications
Wang and Cieff (2005)Offized approachEmployedPan (2008)Approach developedEmployed FAHP for selection of global supplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP for selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachCombined FAHP and image compositing technique for choosing the optimum productKreng and Vu (2007)Utilized approachCombined FAHP and image compositing technique for choosing the optimum productKreng and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for valuationKapoor and Tak (2005)Utilized approachUsing FAHP for evaluation of robot problemsCheng, Chen, and Yu (2005)Utilized approachEmployed FAHP for evaluating of strategy in broadband services	Wang and Chen (2008)	Itilized approach	Using FAHD to derive pairwise comparison matrices
This (2007)Approach developedUsing of FAHP for selection of global suplierWang, Chu, and Wu (2007)Utilized approachEmployed FAHP to selecting the best strategies of maintenanceMa, Chen, and Wu (2007)Utilized approachCombined FAHP and image compositing technique for choosing the optimum productKreng and Wu (2007)Utilized approachEmployed FAHP for selection of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for NPD environment evaluationKapoor and Tak (2005)Utilized approachUsilized approachCheng, Chen, and Yu (2005)Utilized approachEmployed FAHP for evaluating of strategy in broadband services	Pan $(2008)$	Approach developed	Employed FAHP approach for selection of the suitable bridge construction
Wang, Chu, and Wu (2007)Utilized approachEmployed FAHP for selection of giolar approachMa, Chen, and Wu (2007)Utilized approachEmployed FAHP for selection of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Humphreys (2006)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for valuationKapoor and Tak (2005)Utilized approachUsilized approachCheng, Chen, and Yu (2005)Utilized approachEmployed FAHP for evaluation of strategy in broadband services	Chan and Kumar (2007)	Approach developed	Using of FFAHP for selection of global supplier
Ma, Chen, and Wu (2007)Utilized approachCombined FAHP for evaluation of KPS development toolsKreng and Wu (2007)Utilized approachCombined FAHP for evaluation of KPS development toolsKang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Chan,	Wang Chu and Wu (2007)	Itilized approach	Employed FAHP to selecting the best strategies of maintenance
Kreng and Wu (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsKreng and Lee (2007)Utilized approachEmployed FAHP for evaluation of KPS development toolsBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachUtilized approachChan, Chan, Chan, and Humphreys (2006)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for vealuation of FAHP to selection of robot problemsKapoor and Tak (2005)Utilized approachApplication of FAHP for evaluation of strategy in broadband services	Ma Chen and Wu (2007)	Itilized approach	Combined FAHP and image compositing technique for choosing the optimum product
Kang and Lee (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesBozbura, Beskese, and Kahraman (2007)Utilized approachConstruct an FAHP method and entropy weight for rank of different priority mixesChan, Chan, Chan, and Humphreys (2006)Utilized approachUsing FAHP for measurement indicators of human capitalAyağ (2005)Utilized approachUsing FAHP for NPD environment evaluationKapoor and Tak (2005)Utilized approachUsing FAHP for vealuation of robot problemsCheng, Chen, and Yu (2005)Utilized approachEmployed F-AHP for evaluating of strategy in broadband services	Kreng and Wu (2007)	Utilized approach	Employed FAHP for evaluation of KPS development tools
Bozbura, Beskese, and Kahraman (2007)Utilized approachProposed FAHP for measurement indicators of human capitalBozbura, Beskese, and Kahraman (2007)Utilized approachProposed FAHP for measurement indicators of human capitalChan, Chan, and Humphreys (2006)Utilized approachUsing FAHP for classification of technology criteria selectionAyağ (2005)Utilized approachUsing FAHP for NPD environment evaluationKapoor and Tak (2005)Utilized approachApplication of FAHP to selection of robot problemsCheng, Chen, and Yu (2005)Utilized approachEmployed F-AHP for evaluating of strategy in broadband services	Kang and Lee (2007)	Utilized approach	Construct an FAHP method and entropy weight for rank of different priority mixes
Chan, Ch	Bozbura Beskese and Kabraman (2007)	Utilized approach	Proposed FAHP for measurement indicators of human capital
Ayağ (2005)Utilized approachUsing FAHP for NPD environment evaluationKapoor and Tak (2005)Utilized approachApplication of FAHP to selection of robot problemsCheng, Chen, and Yu (2005)Utilized approachEmployed F-AHP for evaluating of strategy in broadband services	Chan, Chan, Chan, and Humphreys (2006)	Utilized approach	Using FAHP for classification of technology criteria selection
Kapoor and Tak (2005)Utilized approachApplication of FAHP to selection of robot problemsCheng, Chen, and Yu (2005)Utilized approachEmployed F-AHP for evaluating of strategy in broadband services	Avağ (2005)	Utilized approach	Using FAHP for NPD environment evaluation
Cheng, Chen, and Yu (2005) Utilized approach Employed F-AHP for evaluating of strategy in broadband services	Kapoor and Tak (2005)	Utilized approach	Application of FAHP to selection of robot problems
	Cheng, Chen, and Yu (2005)	Utilized approach	Employed F-AHP for evaluating of strategy in broadband services
Buyukozkan and Feyziog-lu (2004) Utilized approach Employed FAHP for selection of strategy problem for new product development	Büyüközkan and Feyziog~lu (2004)	Utilized approach	Employed FAHP for selection of strategy problem for new product development
Hsjeh, Lu, and Tzeng (2004) Utilized approach Using FAHP for selection of design and planning alternatives in public office building	Hsieh, Lu, and Tzeng (2004)	Utilized approach	Using FAHP for selection of design and planning alternatives in public office building
Kahraman, Cebeci, and Ulukan (2003) Utilized approach Employed the FAHP for selection of suppliers	Kahraman, Cebeci, and Ulukan (2003)	Utilized approach	Employed the FAHP for selection of suppliers
Kuo, Chi, and Kao (2002) Utilized approach Using of FAHP to develop a decision support system for locating a new CVS	Kuo, Chi, and Kao (2002)	Utilized approach	Using of FAHP to develop a decision support system for locating a new CVS
Cheng (1997) Approach proposed Developed FAHP for evaluation of naval tactical missile systems	Cheng (1997)	Approach proposed	Developed FAHP for evaluation of naval tactical missile systems
Cheng and Mon (1994) Approach developed Proposed FAHP for evaluating weapon systems	Cheng and Mon (1994)	Approach developed	Proposed FAHP for evaluating weapon systems

## Table 3

Distribution based on fuzzy ELECTRE.

Authors	Type of study	Tools and approaches
Vahdani, Mousavi, Tavakkoli-Moghaddam, and Hashemi (2013)	Proposed approach	Extension of the ELECTRE, for FMCGDM problems
Devi and Yadav (2013)	Utilized approach	Applied F-ELECTRE for selecting of the best plant locations problems
Rouyendegh and Erkan (2013)	Utilized approach	Using F-ELECTRE for the selection of academic staffs
Hatami-Marbini, Tavana, Moradi, and Kangi (2013)	Utilized approach	Present a fuzzy group ELECTRE method for health assessment safety
Sepehriar, Eslamipoor, and Nobari (2013)	Approach developed	Employed F-ELECTRE for selection of the best supplier
Vahdani and Hadipour (2011)	Approach developed	Presented F-ELECTRE for solving MCDM problems based on interval-value
Wu and Chen (2011)	Utilized approach	Proposed intuitionistic F-ELECTRE, for solving MCDM problems
Sevkli (2010)	Utilized approach	Proposed F-ELECTRE to determine the criteria of supplier selection
Montazer, Saremi, and Ramezani (2009)	Utilized approach	Employed the F-ELECTRE III for rank the vendors based on the DSS evaluations of the vendors

TOPSIS based on Pythagorean under fuzzy environment. Moghimi and Anvari (2014), have employed F-AHP and TOPSIS for evaluating of performance in cements firms. Safaei Ghadikolaei, Khalili Esbouei, and Antucheviciene (2014), have employed F-AHP, F-COP-RAS, F-VIKOR, and ARAS-F for performance evaluation in Iranian companies. In this category, some previous studies, for example, (Ma, Chang, & Hung, 2013), integrated Delphi method and fuzzy AHP for the selection of technology process. Tsai, Chang, and Lin (2010), used fuzzy AHP and Delphi for evaluating the performance in hospital organization. Tavana, Zandi, and Katehakis (2013), have applied group FANP and TOPSIS to assess a community's overall egovernment readiness. From among 122 studies, 10 (8.20%) papers have been published in 2014, 19 (15.57%) papers have been published in 2013, 22 (18.03%) papers in 2012, 26 (21.31%) papers in 2011, 13 (10.66%) papers in 2010, 14 (11.48%) papers in 2009, 8 (6.56%) papers in 2008, three papers (2.46%) in 2007, two papers (1.64%) in 2006, two papers (1.64%) in 2004, one paper (0.82%) in 2002, two papers (1.64%) in 2000. In case of the type of study, most previous studies have been published in field of management and business using fuzzy MCDM, the numbers of papers that FMCDM utilizing research are 107 (87.70%) papers, 13 (10.66%) papers as FMCDM developing research, in the FMCDM proposing research only two paper (1.64%) published, and one paper (0.82%) was as review paper. The percentages of this section show that most of the studies in the management and business field have implemented the fuzzy MCDM as tools and methods for their decision making problems rather than developing and proposing these fuzzy MCDM tools and applications. In the category of tools and application, researchers in management and business fields have applied 45 papers fuzzy TOPSIS and mixed it with other methods, 64 papers have used fuzzy AHP and mixed it with other methods, 15 papers have used fuzzy ANP and mixed it with other methods, 13 papers have used fuzzy VIKOR and mixed it with other methods, six papers have used fuzzy DEMATEL and mixed it with other methods, and 15 papers have used other mixed tools and applications such as ARAS-F, F-ELECTRE, F-COPRAS and so on.

## 4.5. Science and technology field

Science and technology also was two of the four categories of this survey, which includes 63 papers. In this category there were

## Najib, 2014a; Abdullah & Najib, 2014b; Kelemenis & Askounis, 2010), energy and environmental (Akadiri, Olomolaiye, & Chinyio, 2013; Bagočius et al., 2014; Kabak, Köse, Kırılmaz, & Burmaoğlu, 2014; Shen, Lin, Li, & Yuan, 2010), transportation (Awasthi & Chauhan, 2011), natural resource and environmental management (Lee, Mogi, Kim, & Gim, 2008; Yazdani-Chamzini, 2014) and operations research (Antuchevičiene, 2005; Sun, Lin, & Tzeng, 2009; Zavadskas, Antucheviciene, et al., 2014), and other areas related to the science and technology category .Najafi, Karimpour, and Ghaderi (2014), applied F-AHP for mineral prospectively mapping in the eastern of Iran. Ribeiro, Falcão, Mora, and Fonseca (2014), proposed algorithm by applied FMCDM and mixture aggregation operators based on weighting functions for spacecraft landing with hazard avoidance. Hadi-Vencheh and Mohamadghasemi (2014). utilized F-TOPSIS. F-VIKOR and FWA selection of materials equipment problems, Gao, Jin, Song, Xu, and Wang (2012), developed hierarchical FAHP for distribution of energy-efficient clustering algorithm. Javanbarg, Scawthorn, Kiyono, and Shahbodaghkhan (2012), implemented FAHP for evaluation and optimization of swarm particle. Gao and Hailu (2012), used FAHP to evaluate of fishing recreational in the system of coral reef. Naghadehi, Mikaeil, and Ataei (2009), applied FAHP for selecting and best criteria underground mining in Iran. Results of these categories showed that From 63 published article papers, 49 studies (77.78%) were FMCDM utilizing research, 11 studies (17.46%) were FMCDM developing research and methods and three studies (4.76%) have proposing new approaches and methods based on fuzzy MCDM tools and applications. These studies have published in the following years: 16 papers (25.40%) in 2014, six papers (9.52%) in 2013, nine papers (14.29%) in 2012, 12 papers (19.05%) in 2011, eight papers (12.70%) in 2010, and three papers (4.76%) in 2009, two papers (3.17%) in 2008, two papers (3.17%) in 2007, one paper (1.59%) in 2006, three papers (4.76%) in 2005, and one paper (1.59%) in 1997. In case of the implementation of tools and applications in this category, 15 cases have utilized fuzzy TOPSIS and integrated it with other methods. 32 cases have utilized fuzzy AHP and integrated it with other methods: eight papers have utilized fuzzy DEMATEL and integrated it with other methods, five studies have utilized fuzzy VIKOR and integrated it with

some specific fields of science, include mathematic (Abdullah &

#### Table 4

#### Distribution based on fuzzy DEMATEL.

Authors	Type of study	Tools and approaches
Keskin (2014)	Utilized approach	Applied fuzzy C-means clustering and F-DEMATEL for selection and evaluation supplier performance
Jeng and Tzeng (2012)	Utilized approach	Applies DEMATEL technique to explore the UTAUT variables
Wu (2012)	Utilized approach	Used F-DEMATEL to success implementation of KM based on CSF's
Zhou, Huang, and Zhang (2011)	Utilized approach	F-DEMATEL to figure out CSFs in accordance with potentially numerous criteria
Lee, Li, Yen, and Huang (2011)	Approach developed	Developed F-DEMATEL for evaluation of technology acceptance models problems
Tseng (2010)	Utilized approach	Employed F-DEMATEL for evaluation of EKMC firms
Tseng (2009)	Utilized approach	Application of F-DEMATEL to ranking of real estate agent service quality expectation
Lin and Wu (2008)	Utilized approach	Proposed F-DEMATEL to selection of R&D project in Taiwanese companies
Wu and Lee (2007)	Utilized approach	Used F-DEMATEL for better promoting the competency development of global managers

other methods.

#### Table 5

Distribution based on fuzzy PROMETHEE.

Authors	Type of study	Tools and approaches
Gupta, Sachdeva, and Bhardwaj (2012)	Utilized approach	Employed the F-PROMETHEE method for logistic provider selection
Yilmaz and Dağdeviren (2011)	Approach developed	Combined zero-one goal programming and F-PROMETHEE for equipment selection
Chen, Wang, and Wu (2011)	Utilized approach	Applied F-PROMETHEE to evaluate suppliers
Saidi Mehrabad and Anvari (2010)	Utilized approach	Application of fuzzy C-Means and F-PROMETHEE for evaluation of FMS
Halouani, Chabchoub, and Martel (2009)	Approach developed	Integrated linguistic 2-tuples and PROMETHEE for selection of facilitate project task

## Table 6

## Distribution based on fuzzy TOPSIS.

Authors	Type of study	Tools and approaches
Zagorskas et al. (2014)	Utilized approach	Employed fuzzy gray TOPSIS for selection of best alternatives in the brick wall insulation
Liu, Ren, Wu, and Lin (2014)	Utilized approach	Implemented of ITL-TOPSIS for selection of optimal robots in manufacturing
Kurt (2014)	Utilized approach	Used of F-TOPSIS and Choquet fuzzy for selection of nuclear power plant location
Mokhtari et al. (2014)	Utilized approach	Employed of F-TOPSIS for selection of well control system in oil industry
Wang (2014) Molektarian Sadi norbad and Malusi (2014)	Approach developed	Developed FMCDM to solve problems in FMCDM based on TOPSIS
MOKILIAIIAII, SAUI-HEZHAU, AHU MAKUI (2014)	Approach proposed	Proposed IVF-TOPSIS for solve problems in selection of Identity location Used TOPSIS and OED for selection and evaluation of KMS under fuzzy environment
Mokhtarian (2014)	Approach developed	Extended the F-TOPSIS for solve problem in MCDM methods based on previous studies
Maldonado-Macías, Alvarado, García, and	Utilized approach	Used F-TOPSIS for assessment of ergonomic compatibility in advance manufacturing
Balderrama (2014)		technology
Arabzad, Ghorbani, Razmi, and Shirouyehzad (2014)	Utilized approach	Employed F-TOPSIS for selection of supplier and allocation problem
Zhang and Xu (2014)	Approach developed	Extended TOPSIS based on Pythagorean under fuzzy environment
Kilic (2013)	Approach developed	Integrated of FTOPSIS and linear programming model for selection of suppliers
(2013)	Utilized approach	Using the hierarchical FLOPSIS for selection of suppliers in Iran
Kim, Chung, Jun, and Kim (2013) Dymova, Sevastjanov, and Tikhonenko (2013)	Approach Proposed	Developed an F-IOPSIS for treated wastewater (IWW) Proposed a new TOPSIS method which is free of known methods limitations and heuristic assumptions
Wang and Chan (2013b)	Utilized approach	Using hierarchical F-TOPSIS to evaluation of different green initiatives
Wang and Chan (2013)	Utilized approach	Implemented of hierarchical F-TOPSIS for evaluation of green supply chain performance
Li (2013)	Utilized approach	Using F-TOPSIS for evaluation and selection of knowledge management system
Amirzaden and Shoorvarzy (2013) Singh and Benyoucef (2012)	Utilized approach	Application of FIOPSIS for evaluation of banks quality elements by SERVQUAL
Shigh and Benjoucer (2013) Shen, Olfat, Govindan, Khodaverdi, and Diabat	Utilized approach	Applied F-TOPSIS to generate an overall performance score for supplier
Vinodh, Mulanjur, and Thiagarajan (2013)	Utilized approach	Proposed F-TOPSIS to select the criteria for sustainability among several sustainability criteria
Maity and Chakraborty (2013)	Utilized approach	Applied F-TOPSIS to solve grinding wheel abrasive material selection problem
Dymova, Sevastjanov, and Tikhonenko (2013)	Approach developed	Extended F-TOPSIS for compromise solution to a FMCDM problem
Tansel İç (2012)	Utilized approach	Proposed FTOPSIS and LP approaches for the banks to determine the credit risks
Rouhani, Ghazanfari, and Jafari (2012)	Approach developed	Application of F-TOPSIS for business intelligence system
Huang and Peng (2012)	Approach developed	Employed the Fuzzy Rasch in TOPSIS to analyze the TDC in nine Asian countries
Uysal and Tosun (2012)	Utilized approach	Implemented of F-TOPSIS for selection of maintenance systems
Alsiali aliu Çulikaş (2012) Vəbdəni Mousəvi ənd Təyəkkoli-Moghəddəm	Approach developed	Applied F-IOPSIS for evaluation performance in Sugar Plants Proposed fuzzy modified TOPSIS for rapid prototyping process selection and the robot
(2011)	Approach developed	selection
Boran, Genc, and Akay (2011)	Approach developed	Extended and proposed the F-TOPSIS to select appropriate personnel among candidates
Chamodrakas, Leftheriotis, and Martakos (2011)	Approach developed	A fuzzy approach for ranking alternatives in MADM problems based on TOPSIS
Awasthi, Chauhan, Omrani, and Panahi (2011)	Utilized approach	Presented an F-TOPSIS for evaluation of service quality in urban transportation systems
Afshar, Mariño, Saadatpour, and Afshar (2011)	Utilized approach	Employed F-TOPSIS for solve the problems in real water resource management in Iran
Awasthi, Chauhan, and Omrani (2011)	Utilized approach	Employed F-TOPSIS for assessment and selection of transportation systems
Yalig, Bolisali, alia Walig (2011) Soper Kara (2011)	Utilized approach	Using F-10PSIS for ranking and selection of potential suppliers' problems
Singh and Benyoucef (2011)	Utilized approach	Used F-TOPSIS to solution of MCDM problems in selection of supply chain coordination
Kaya and Kahraman (2011)	Approach developed	Modified F-TOPSIS methodology for the selection of the best energy technology
La Scalia, Aiello, Rastellini, Micale, and Cicalese (2011)	Approach developed	Utilized TOPSIS for transplantation of the pancreatic islet
Eraslan and Iç (2011)	Utilized approach	Employed F-TOPSIS for socio-economic level of geographical investment regions
Kelemenis, Ergazakis, and Askounis (2011)	Utilized approach	Extended of F-TOPSIS for selection of support managers
Liao and Kao (2011)	Utilized approach	This study proposes integrated fuzzy TOPSIS and MCGP for supplier selection problem
топак, sevkii, sanai, and zaim (2011) Tan (2011)	Ounzeu approach Approach developed	Using r-10PSIS for business competition in the Turkish domestic airline industry Developed FGDM using Choquet integral based TOPSIS
Amiri and Golozari (2011)	Proposed approach	Application of F-TOPSIS for critical path by using quality cost time risk criteria
Jiang, Chen, Chen, and Yang (2011)	Approach developed	Proposed F-TOPSIS and Fuzzy BS for solve Group Belief MCDM problems
Krohling and Campanharo (2011)	Utilized approach	Applies F-TOPSIS for GDM in accidents with oil spill in the sea
Iç and Yurdakul (2010)	Utilized approach	Applied F-TOPSIS for model of credit scoring in manufacturing industries
Kelemenis and Askounis (2010)	Utilized approach	Applied F-TOPSIS for personal selection and ranking alternatives
Kaya and Kahraman (2010) Sadi-Nezhad and Khalili Damehani (2010)	Proposed approach	Modified FIUPSIS for the selection of the energy technology alternatives
Cavallaro (2010)	Utilized approach	Proposed F-TOPSIS to investigate the feasibility of utilizing a molten salt
Roghanian, Rahimi, and Ansari (2010)	Proposed approach	Evolution of first aggregation and last aggregation in fuzzy group TOPSIS
Yu and Hu (2010)	Utilized approach	Employed F-TOPSIS method to evaluate the performance of multiple manufacturing
(Sun and Lin (2009)	Utilized approach	Application of F-TOPSIS for evaluation of competitive advantages of shopping websites
Saremi, Mousavi, and Sanayei (2009)	Utilized approach	Using F-TOPSIS for selection of TQM external consultant environment
Ashtiani, Haghighirad, and Makui (2009)	Approach developed	Presented new approach of FTOPSIS for solving MCDM problems in which the weights of criteria are unequal
Chu and Lin (2009)	Approach developed	Suggested an F-TOPSIS for the interval arithmetic
Kahraman et al. (2009)	Utilized approach	Application of F-TOPSIS for selection of information systems outsourcing
Zeydan and Çolpan (2009) Athanasonoulos, Piba, and Athanasonoulou	Utilized approach	Application of DEA and F-TOPSIS for measurement performance
	otilizeu appiliacii	alternatives
Yurdakul and İc (2009)	Utilized approach	Using of F-TOPSIS to derive quality indexes for electronic packages
		· · ·

## Table 6 (continued)

Authors	Type of study	Tools and approaches
Chen and Tsao (2008) Wang (2008)	Approach developed Utilized approach	Presented interval-valued F-TOPSIS rankings determined by different distance measures Utilized F-TOPSIS for evaluation of airlines financial performance
Mahdavi, Mahdavi-Amiri, Heidarzade, and Nourifar (2008)	Approach developed	Designed an F-TOPSIS for obtain the ideal solutions in fuzzy environment
Kahraman, Çevik, Ates, and Gülbay (2007)	Proposed approach	Proposed a fuzzy hierarchical TOPSIS for the evaluation of multi-criteria in the industrial robotic systems
Kahraman, Ates, Çevik, and Gülbay (2007)	Utilized approach	Developed fuzzy hierarchical TOPSIS and applied to an e-service provider selection problem
Benítez et al. (2007)	Utilized approach	Application of F-TOPSIS for service quality measurement in hotel industry
Wang and Lee (2007)	Approach developed	Generalized TOPSIS to FMCGDM in a fuzzy environment
Kahraman, Ates, Çevik, Gülbay, and Erdogan (2007)	Utilized approach	Applied the "Hierarchical F-TOPSIS for selection of logistics information
Wang and Chang (2007)	Utilized approach	Employed F-TOPSIS for evaluation initial training aircraft
Dimova, Sevastianov, and Sevastianov (2006)	Approach developed	Proposed F-TOPSIS for supplier selection problems in the supply chain system
Chen, Lin, and Huang (2006)	Approach developed	Developed F-TOPSIS to deal with the supplier selection problems in the supply chain
Yong (2006)	Utilized approach	Using F-TOPSIS for selection of plant location
Wang and Elhag (2006)	Approach developed	Proposed a F-TOPSIS method for evaluation of risk assessment
Yong and Qi (2005)	Utilized approach	Proposed a new centroid-index ranking method of fuzzy numbers by using TOPSIS
Antuchevičiene (2005)	Approach developed	Developed F-TOPSIS for solving problems in the crisp value to model real-life situations
Karsak (2002)	Approach developed	Developed F-TOPSIS to considering strategic performance and economic criteria
Chen (2000)	Approach developed	Extended the F-TOPSIS for group decision making environment
Chen (2000)	Approach developed	Developed TOPSIS in fuzzy environment for group decision making

Table	7
-------	---

Distribution based on fuzzy ANP.

Authors	Type of study	Tools and approaches
Rabbani et al. (2014)	Utilized approach	Implemented of fuzzy COPRAS, ANP and SBSC for evaluation of performance in
		Iranian oil companies
Kabak et al. (2014)	Utilized approach	Employed F-ANP for evaluation of performance in building energy
Gürbüz and Albayrak (2014)	Utilized approach	Used ANP and Choquet Integral under fuzzy environment for assessment of HRM
	* *	performance
Palanisamy and Zubar (2013)	Utilized approach	Using F-QFD, ANP and mathematical for rank of vendor
Tavana, Momeni, Rezaeiniya, Mirhedayatian, and	Utilized approach	Using of COPRAS-G and FANP for selection of social media platform
Rezaeiniya (2013)		
Lin (2012)	Utilized approach	Integrated FANP with FMOLP for selection the best suppliers
Kang, Lee, and Yang (2012)	Utilized approach	Applied the FANP for selection of supplier about IC packaging
Vahdani, Hadipour, and Tavakkoli-Moghaddam (2012)	Approach developed	Proposed FANP to solution of MCDM problems
Yang and Chang (2012)	Utilized approach	Applied F-ANP for decision process of customers in bundles selection
Kuo and Liang (2011)	Utilized approach	Using the FANP construct fuzzy weights of all criteria for locations selection
Liou et al. (2011)	Utilized approach	Combined FANP and fuzzy preference programming to selection of strategic
		alliances partners
Büyüközkan and Çifçi (2011)	Utilized approach	Employed the FANP for selection of supplier with incomplete information
Vinodh, Anesh Ramiya, and Gautham (2011)	Utilized approach	Employed the FANP for selection of supplier in manufacturing organization
Chang, Horng, and Lin (2011)	Utilized approach	Using FANP for experts' knowledge-based systems algorithm
Chang, Horng, and Lin (2011)	Utilized approach	Using F-ANP for evaluation of management decision making
Özgen and Tanyas (2011)	Utilized approach	Proposed the FANP for selection of international roads and broker agencies
Ayağ and Özdemir (2011)	Utilized approach	Using of FANP for evolution the selection of machine tool
Boran and Goztepe (2010)	Utilized approach	Proposed an FANP for evaluation of vendor alternatives
Chen and Pang (2010)	Utilized approach	Using FANP for distribution of existing knowledge to create new knowledge
Vinodh, Gautham, Anesh Ramiya, and Rajanayagam	Utilized approach	Using the FANP for selection concept selection in agile manufacturing
(2010)		
Ayağ and Ozdemir (2009)	Utilized approach	Employed the FANP for selection of ERP software problems
Tseng, Chiang, and Lan (2009)	Utilized approach	Applied FANP for selection the optimal supplier in SCMS
Tuzkaya and Onüt (2008)	Utilized approach	Applied FANP for large-sized real-life problem in the transportation projects
Ayağ and Ozdemir (2007)	Utilized approach	Presented FANP for selection of ERP software problem
Mikhailov and Tsvetinov (2004)	Proposed approach	Proposed FANP for imprecision of the service evaluation process and tackling the
		uncertainty
Pang and Bai (2013)	Utilized approach	Using FANP for selection a supplier alternative and evaluation
Moalagh and Ravasan (2013)	Utilized approach	Using FANP for assessment ERP post-implementation success

## 4.6. Distribution based on fuzzy MCDM and MCDM tools and approaches

Table 1 shows frequency of both fuzzy MCDM and MCDM tools and approaches. Based on results presented in this table, a total of 1081 studies have employed these two kinds of DM tools and approaches, whereas 403 studies have used fuzzy MCDM and 745 studies have utilized MCDM. This table shows that hybrid FMCDM has been used more than other tools and approaches. The second one is the MCDM tools and approaches and traditional AHP is the third in this ranking. The frequency of other tools and approaches are presented in Table 1.

## 4.7. Distribution based on fuzzy MCDM tools and approaches

Tables 2–9 show implementation of each fuzzy MCDM tools and approaches. Based on results presented in these tables, a total of 403 studies have employed fuzzy DM tools and approaches, these tables show that fuzzy AHP with 100 papers has been used more than other tools and approaches. The second one is the other

Table	8
-------	---

Distribution based on fuzzy VIKOR.

Authors	Type of study	Tools and approaches
Mokhtarian et al. (2014)	Approach proposed	Proposed IVF-VIKOR for solve problems in selection of facility location
Liu et al. (2014)	Utilized approach	Employed F-VIKOR for personal selection in HRM section of organization
Hajiagha et al. (2014)	Approach developed	Developed F-VIKOR for solve problem in minimizing in ideal and anti-ideal solution
Kim and Chung (2013)	Utilized approach	Evaluated the vulnerability of the water supply in the South Korean by fuzzy VIKOR approach
Liao and Xu (2013)	Approach developed	Proposed the hesitant F-VIKOR for effective solving MCDM problems
Kumar, Singh, and Dureja (2012)	Utilized approach	Applied the F-VIKOR and CFPR for solution of logistic outsourcing problem
Yücenur and Demirel (2012)	Utilized approach	Applied extended VIKOR for the selection of best solution in Turkish insurance firms
Kuo and Liang (2012)	Utilized approach	Presented F-VIKOR to solve MCDM problems and performances
Jeya Girubha and Vinodh (2012)	Utilized approach	Extended the VIKOR for selection of materials problems
Kuo (2011)	Proposed approach	Proposed F-VIKOR, fuzzy sets and GRA improve levels of service quality
Shemshadi, Shirazi, Toreihi, and Tarokh	Approach developed	Extended the VIKOR with a mechanism to extract and deploy objective weights based on
(2011)		Shannon entropy concept
Park, Cho, and Kwun (2011)	Approach developed	Extended F-VIKOR for ranking and selection of optimal alternatives
Opricovic (2011)	Approach developed	Developed F-VIKOR for rank of fuzzy numbers
Sasikumar and Haq (2011)	Utilized approach	Proposed the F-VIKOR to determine the best 3PRLP selection process
Vahdani, Hadipour, Sadaghiani, and	Approach developed	Presented F-VIKOR for solving MCDM problems based on interval-valued
Amiri (2010)		
Chen and Wang (2009)	Utilized approach	Using F-VIKOR for selection partners in IS/IT outsourcing projects

and integrated papers of fuzzy MCDM tools and approaches and fuzzy TOPSIS is the third in this ranking.

#### 4.8. Distribution based on publication year

Fig. 2 presents important evidence based on the frequency of distribution by the year of publication. The results indicate that from 1994 to 2014, the information about the use of fuzzy MCDM and MCDM tools and approaches have grown increasingly. According to the findings of this section, the use of these tools and approaches in 1994 was four papers and this number increased to 10 papers in 1996. Surprisingly, from 2008 to 2009, the numbers of studies have dramatically increased. Although the use of fuzzy MCDM and MCDM tools and approaches have increased in each year, the number of those papers in 2012 (171) have decreased compared to 2011 (176). Another interesting result in this table is about 2013 which previous studies have applied tools and techniques more than other years. This year has the highest number of publications (188). Accordingly, it can be indicated that researchers in different fields and categories use the fuzzy MCDM and MCDM tools and approaches nowadays in their research, and it can be predicted that in coming years, these numbers will increase.

#### 4.9. Author distribution by nationality

Table 10 presents the nationality of the authors who have utilized the fuzzy MCDM and MCDM approaches in their studies. As can be seen, 59 countries have contributed to this survey. In addition, the table reveals that most of the papers are from Taiwan (21.46%), Turkey (12.21%), Iran (11.19%) and China (10.73%). These results with a few differences and similarities were confirmed by other review papers such as Behzadian et al. (2012) that ranked Taiwan as the first country, China as the second, and Iran as the third one.

## 5. Discussion

This study attempted to review papers published during 20 years (1994–2014) about fuzzy MCDM in popular international journals, which are accessible database systems such as ScienceDirect, Springer, Emerald, John Wiley, ProQuest, and Taylor & Francis. The first aim of this paper was systematically reviewing the studies conducted based on fuzzy MCDM tools and approaches. To this end, in the first step, a total of 1081 published papers about fuzzy

MCDM and MCDM were systematically and carefully chosen and summarized based on title, abstract, introduction, research method, and conclusion. In the next step, according to the predefined objective of this study, those papers related to fuzzy MCDM tools and approaches were selected. From 1081 articles, 403 studies (37.28%) have been focused on fuzzy DM tools and approaches and 678 studies (62.72%) have used DM tools and approaches. In this review, the obtained results were analyzed based on six research questions; these questions were (1) which fuzzy and non-fuzzy DM techniques have frequently been applied? (2) Which type of study has applied these fuzzy MCDM techniques? (3) Which one of the four fields (i.e., science, business, technology, and engineering) has used further the fuzzy MCDM techniques types? (4) What kind of fuzzy MCDM techniques have been employed in these years based on the four fields? (5) Which countries have published these fuzzy MCDM tools based on the number of publications in these four fields? And finally, (6) in which year authors have further published fuzzy MCDM tools based on frequency in the four fields? To answer the first question, we considered the results presented in Table 1 that showed the number and percentage of those DM tools and approaches in both fuzzy and non-fuzzy MCDM. This table revealed that hybrid fuzzy MCDM was ranked as the first tool among other tools and approaches; additionally, in the individual tools and approaches, AHP and fuzzy AHP were ranked as the second and third tools. The results obtained for this question were presented in Table 1. To answer the second question, we read the methodology section of each paper very carefully and classified the studies in three types. Based on our reading, some studies have used fuzzy and non-fuzzy DM as tool and technique for solve decision-making problems. Based on our experience and discussions held with some experts on fuzzy and non-fuzzy DM issues about this type of studies, we decided to call this type of study FMCDM utilizing research. Some scholars have attempted to develop fuzzy and non-fuzzy DM tools and approaches based on their objectives; therefore, the FMCDM developing research is considered as the second type of study. Furthermore; our review indicated that some researchers have proposed new approach based on fuzzy and non-fuzzy DM tools and techniques, which we named FMCDM proposing research type. The answers to questions three and four were presented sections 4.3–4.5. These sections indicated that 90 papers have employed fuzzy TOPSIS and combined it with other methods, 102 papers have employed fuzzy AHP and combined it with other methods, 32 papers have employed fuzzy ANP and combined it with other methods, 16 papers have employed fuzzy VIKOR and combined it

## Table 9

Distribution based on other fuzzy tools and integrated studies such as F- COPRAS, ARAS-F and F-WASPAS.

Authors	Type of study	Tools and approaches
Akdag et al. (2014)	Utilized approach	Applied AHP, Yager's min-max and TOPSIS for evaluation of service quality in hospital
Zare Mehrjerdi (2014)	Utilized approach	Implemented of SAW, TOPSIS and QSPM in selection of strategic system
Hadi-Vencheh and Mohamadghasemi (2014)	Utilized approach	Utilized F-TOPSIS, F-VIKOR and FWA selection of materials equipment problems
Yazdani-Chamzini et al. (2014)	Utilized approach	Used of AHP, TOPSIS and DEMATEL for investment strategy selection under fuzzy
Cil-Lafuente Merigó and Vizuete (2014)	Itilized approach	Environment Employed F-AHP and FDM to evaluate of luxury resort hotels criteria
Rikhtegar et al. (2014)	Utilized approach	Applied F-SAW and ANP for assessment of risks pertaining in the mining projects
Azadnia et al. (2014)	Utilized approach	Employed F-AHP and RBWF methods for selection of sustainable supplier
Rabbani et al. (2014)	Utilized approach	Implemented of fuzzy COPRAS, ANP and SBSC for evaluation of performance in Iranian oil
		companies
Tadić et al. (2014)	Utilized approach	Utilized fuzzy ANP, fuzzy VIKOR and fuzzy DEMATEL for selection of city logistics
Zavadskas, Antucheviciene et al. (2014)	Approach developed	Developed WASPAS and results compared with COPRAS-IVIF-TOPSIS-IVIF and IFOWA for
Zavadskas Turskis and Bagočius (2014)	Utilized approach	IMPFOVE OF WPM AND WSM ACCURACY Used ARAS, F and AHD for solve of problems in construction site selection in Fastern Baltic
Zavauskas, Turskis, and Dagocius (2014)	otilized approach	see
Avikal, Mishra et al. (2014)	Utilized approach	Applied F-AHP and PROMETHEE approaches for selection tasks in disassembly line
Akdag et al. (2014)	Utilized approach	Used fuzzy set, AHP, TOPSIS, OWA and Yager's min-max to evaluated of hospital service
		quality
Rabbani et al. (2014)	Utilized approach	Used ANP, COPRAS and BSC approaches for assessment of performance in oil companies
Kucukvar et al. (2014)	Utilized approach	Applied F-ENTROPY and TOPSIS for ranking of life cycle sustainability performance
Vinodh, Prasanna, and Hari Prakash (2014)	Utilized approach	Employed F-AHP and TOPSIS for selection and assessment of performance in plastic
Veh et al. $(2014)$	Utilized approach	Ised F-AHP and F-DEMATEL for identifying the critical factors in NPD
Tavana, Khalili-Damghani, and Rahmatian	Utilized approach	Employed F-ANP, DEMATEL and F-DEA for evaluation of performance in pharmaceutical
(2014)	otimbea approach	companies
Wang and Wu (2014)	Utilized approach	Applied F-AHP, F-DEMATEL and FDM for evaluation of PLC
Anojkumar et al. (2014)	Utilized approach	Employed FAHP-VIKOR, FAHP-PROMTHEE, FAHP-TOPSIS and FAHP-ELECTRE for selection
		of material in sugar industry
Zavadskas, Antucheviciene, Hajiagha, and	Approach developed	Developed WASPAS-IVIF for evaluation of performance in MCDM problems
Hashemi (2014)	I Idilian damage ab	Aralist FALLD and F DROMPTUPE for second of surplice answer
Chorabase et al. (2014)	Utilized approach	Applied F-AHP and F-PROMETHEE for assessment of supplier process Used of E-COPPAS and interval type. 2 for selection of supplier
Moghimi and Anyari (2014)	Utilized approach	Employed F-AHP and TOPSIS for evaluating of performance in cements firms
Uvgun, Kacamak, and Kahraman (2014)	Utilized approach	Utilized F-ANP and DEMATEL for selecting and evaluating of outsourcing provider
Avikal, Jain et al. (2014)	Utilized approach	Implemented of F-AHP, M-TOPSIS and KANO model for ranking the tasks in work stations
		bade on their assignment
Liou, Chuang, and Tzeng (2014)	Utilized approach	Applied ANP and DEMATEL for improve and evaluate supplier
Safaei Ghadikolaei et al. (2014)	Utilized approach	Employed F-AHP, F-COPRAS, F-VIKOR, ARAS-F for performance evaluation in Iranian
Kabin and Sumi (2014)	I Idilian damage ab	companies
Rabir and Summisonly (2014) Raykasonly and Dyrmysonly (2014)	Utilized approach	United F-AHP and PROMETHEE for selection of TQM consultant
Daykasogiu allu Dullilusogiu (2014)	otilized approach	level
Bairagi et al. (2014)	Utilized approach	Employed F-AHP, F-VIKOR- F-TOPSIS and COPRAS-G for selection of robots
Keršulienė and Turskis (2014)	Utilized approach	Utilized ARAS-F and AHP for selection of chief officer in the accounting department
Zamani et al. (2014)	Utilized approach	Used ARAS-F and ANP for extension of brand in marketing strategies
Kaya and Kahraman (2014)	Utilized approach	Implemented of F-TOPSIS and F-AHP for performance evaluation of intelligent building
Yazdani-Chamzini (2014)	Utilized approach	Applied F-AHP and F-TOPSIS for selection and evaluation of the feasible handling
	114 <sup>11</sup>	equipment
Intepe, Bozdag, and Koc (2013)	Utilized approach	Applied MOLP and TOPSIS in order to determine assembly process
Sakthivel et al. (2013) Charberi Mahammad Amhrad, and Shahir	Utilized approach	Proposed TOPSIS, GRA and FAHP to assess the best fuel blend
(2013)	otilizeu approach	епрюуси глаг ани г-тогою ю зенсион от supplier based он г-кано
Tavana, Zandi et al. (2013)	Utilized approach	Applied group FANP and TOPSIS to assess a community's overall e-government readiness
Samvedi, Jain, and Chan (2013)	Utilized approach	Integrated FAHP and F-TOPSIS for selection of risks in a supply chain
Jun, Chung, Kim, and Kim (2013)	Utilized approach	Using of TOPSIS, F-TOPSIS and WSM for determine the risks of flood in South Korea
Hsu, Liou, and Chuang (2013)	Approach developed	Combined ANP and DEMATEL methods for selection of outsourcing provider
Tavana, Khalili-Damghani, and Abtahi (2013)	Utilized approach	Application of ANP and F-TOPSIS for selection of location international distribution center
Baykasoğlu, Kaplanoğlu, Durmuşoğlu, and Şahin (2013)	Utilized approach	Integrating fuzzy hierarchical TOPSIS and F-DEMATEL for selection of truck
Dincer and Hacioglu (2013)	Utilized approach	Employed of F-VIKOR and F-AHP for satisfy of customer in Turkish banks
Vinodh, Varadharajan, and Subramanian (2013)	Utilized approach	Employed F-TOPSIS and VIKOR for best concept to enhance the agility in product design
LIN (2013) Kabraman, Suder, and Cabi (2012)	Utilized approach	Utilized of FDM, FAHP for evaluation of fashion design scheme criteria
Kahraman, Suder, and Cebi (2013)	Utilized approach	Employed F-AHP and F-10PSIS to evaluation of NPD performance
Ivia et al. (2013) Sari (2013)	Utilized approach	Integrated E-TOPSIS FAHP and Monte Carlo simulation for PEID selection
$\operatorname{Liu} \operatorname{Wu} \text{ and } \operatorname{Li} (2013)$	Utilized approach	Proposed OWA operator and VIKOR- for evaluation HCW disposal methods
Zouggari and Benyoucef (2012)	Utilized approach	Employed FAHP and F-TOPSIS for suppliers' selection
Kabak, Burmaoğlu, and Kazançoğlu (2012)	Utilized approach	Combined FANP, F-TOPSIS and F-ELECTRE sniper selection
Büyüközkan and Çifçi (2012)	Utilized approach	Combined F-DEMATEL, F-TOPSIS and F-ANP for supplier selection
Chou and Cheng (2012)	Utilized approach	Combined FANP and F-VIKOR for evaluation of website quality
Fouladgar, Yazdani-Chamzini, Lashgari et al. (2012)	Utilized approach	Employed FAHP and COPRAS for evaluation of maintenance strategy
Choudhary and Shankar (2012)	Utilized approach	Applied F-AHP-TOPSIS for selecting and evaluating of optimal locations in TTPs
Yalcin, Bayrakdaroglu, and Kahraman (2012)	Utilized approach	Using FAHP, TOPSIS and VIKOR for evaluation of financial performance

(continued on next page)

# Table 9 (continued)

Authors	Type of study	Tools and approaches
Fouladgar, Yazdani-Chamzini, Zavadskas, and	Utilized approach	Employed F-ANP and fuzzy COPRAS for working evaluation strategies
Haji Moini (2012) Baležentis et al. (2012)	Utilized approach	Lising E-VIKOR E-TOPSIS and E-ARAS for economic sectors assessment
Aliei et al. (2012)	Utilized approach	Using FAHP and F-TOPSIS for technology information assessment
Hsu, Wang, and Tzeng (2012)	Utilized approach	Combined DEMATEL and ANP with VIKOR to solve the recycled materials
Chou, Sun, and Yen (2012)	Utilized approach	Using FAHP and F-DEMATEL to evaluate Taiwanese universities performance
Chen and Chen (2012) Zaki Avez and Cörcer Önderrin (2012)	Utilized approach	Application F-DEMATEL, FANP and FAHP for TQM measurement criteria performance
Taba and Rostam (2012)	Utilized approach	Using ANP and modified TOPSIS for evaluation of machine tool alternatives
Kutlu and Ekmekçioğlu (2012)	Utilized approach	Application of FAHP and F-TOPSIS for failure effects and modes
Paksoy, Pehlivan, and Kahraman (2012)	Utilized approach	Using FAHP and F-TOPSIS for development of organization strategy in distribution channel management
Demirel, Yücenur, Demirel, and Muşdal (2012)	Utilized approach	Using FAHP and FANP for the evaluation of alternate land cover policies
Büyüközkan and Çifçi (2012)	Utilized approach	Application of F-DEMATEL, F-TOPSIS and FANP to evaluate the green supplier
Atalay and Eraslan (2012)	Utilized approach	Applied FAHP, F-TOPSIS, and FADT for electronic devices evaluation
Chen and Yang (2011)	Approach developed	Applied FAHP and F-TOPSIS for determine weights of criteria for selecting logistics tool Employed the E-TOPSIS and FAHP for supplier selection
Kuo (2011)	Approach developed	Developed F-DEMATEL and TOPSIS for selection of location in center of international distribution
Ka (2011)	Utilized approach	Applied F-ELECTRE and FAHP to selection of China dry port location
Tuzkaya, Gülsün, and Önsel (2011)	Utilized approach	Evaluation of MHESP by employ the FANP and F-PROMETHEE approaches
Deng and Chan (2011)	Approach developed	Combination of FST, TOPSIS and DST to deal with selection of supplier problem
Yu, Guo, Guo, and Huang (2011) Lashgari, Fouladgar, Yazdani-Chamzini, and Skibniewski (2011)	Utilized approach Utilized approach	Combined FAHP F-TOPSIS for ranking of B2C e-commerce websites
Shafia, Mazdeh, Vahedi, and Pournader (2011)	Utilized approach	Used F-TOPSIS and SAW for evaluation of CRM performance based on BSC
Aydogan (2011)	Utilized approach	Integrated Rough-AHP and F-TOPSIS for ranking performance measurement in Turkish aviation firms
Liao (2011)	Utilized approach	Applied FAHP and MSGP for select the best pricing strategy for NPD
Shen, Lin, and Tzeng (2011)	Utilized approach	Integrated FDM, DEMATEL and ANP for construct a technology selection model
Rostamzadeh and Sofian (2011)	Utilized approach	Using of FAHP and F-TOPSIS for prioritization effective of 7Ms performance
Hung $(2011)$	Utilized approach	Integrated DEMATEL FGP and ANP for activity-based divergent supply chain
Hadi-Vencheh and Mohamadghasemi (2011)	Approach developed	Implemented of FAHP, DEA and SAW for classify of ABC inventory
Dalalah, Hayajneh, and Batieha (2011)	Utilized approach	Modified F-DEMATEL and TOPSIS for evaluation and selection supplier based on criteria
Pires, Chang, and Martinho (2011)	Utilized approach	Employed AHP based on F-TOPSIS for evaluation of solid waste management in Portugal
Kahraman and Kaya (2011) Fu, Chu, Chao, Lee, and Liao (2011)	Utilized approach	Using F-AHP and F-ELECIUKE for evaluation of websites quality levels
Azadeh, Nazari-Shirkouhi, Hatami-Shirkouhi, and Ansarineiad (2011)	Utilized approach	Combined FAHP and TOPSIS for productive operators' assignment
Rathod and Kanzaria (2011)	Utilized approach	Application AHP, TOPSIS and F-TOPSIS to solve PCM selection problem
Ekmekçioğlu, Kaya, and Kahraman (2010)	Utilized approach	Applied FAHP and modified F-TOPSIS for the selection of appropriate disposal method and site for MSW
Chen and Chen (2010)	Utilized approach	Applied FAHP and VIKOR to complete the construction of the AIS
Liou and Chuang (2010) Tuzkava Gülsün Kahraman and Özgen (2010)	Utilized approach	Used FWCDW and FAHP for determine reputation and in corporate image citienta Implemented of F-ANP and F-PROMETHEF for evaluation of material handling equipment
Kaya (2010)	Utilized approach	Application of FAHP and F-TOPSIS- for website quality evaluation
Dursun and Karsak (2010)	Utilized approach	Using of F-TOPSIS and OWA operator for personnel selection
Wang, Fan, and Wang (2010)	Utilized approach	Integrated FAHP, TOPSIS and FPP methods for aero engine health assessment"
Unut, Efendigil, and Soner Kara (2010) Chatteriee Manikrao Athawale and Chakraborty	Utilized approach	Combined FAHP and F-TOPSIS for selection the shopping center site
(2010)	otinzeu appioacii	Using virtual and Electric to solve the robot selection problems
Chen and Hung (2010)	Utilized approach	Using F-AHP and F-TOSIS for selection of partners' strategic manufacturing
18a1 et al. (2010) Amiri (2010)	Utilized approach	Employed FAHP and Depni for evaluation of performance in hospital organization Using E-TOPSIS and AHP to analyze the structure of the project selection problem
Chen and Wang (2010)	Utilized approach	Using FDM and FAHP for evaluation firms adjust business strategy
Sun (2010)	Utilized approach	Employed FAHP and FTOPSIS, for the evaluation of performance
Chen and Chen (2010)	Utilized approach	Applied VIKOR and FAHP for complete the construction of the AIS
Torfi, Farahani, and Rezapour (2010)	Proposed approach	Applied FTOPSIS and FAHP to determine the relative weights and rank the alternatives
Onut, Kara, and ISIK (2009) Boran Genc Kurt and Akay (2009)	Utilized approach	Developed a supplier evaluation approach based on the TOPSIS and ANP Combined F-TOPSIS and IFWA for select supplier in group decision making
Gumus (2009)	Utilized approach	Applied FAHP and TOPSIS for evolution of hazardous waste in transportation firms
Demirel, Muşdal, Demirel, and Yücenur (2009)	Utilized approach	Using FAHP and FANP for agricultural strategy selection
Bashiri and Hosseininezhad (2009)	Proposed approach	Employed of FAHP and FIS for multi-facility location problems
Ertuğrul and Karakaşoğlu (2009)	Utilized approach	Using FAHP and TOPSIS for evaluation of performance in Turkish cement firms
Zaerpour, Rabbani, Gharehgozli and Tavakkoli-	Utilized approach	Employed FAHP and TOPSIS for classify of make to stock make to order and hybrid
Moghaddam (2009)	Stinzea appioacii	products
Mahdavi, Heidarzade, Sadeghpour-Gildeh, and Mahdavi-Amiri (2009)	Proposed approach	Defined FPIS and FNIS based of TOPSIS concept for decision making problems
llangkumaran and Kumanan (2009) Büyüközkan and Ruan (2008)	Utilized approach Utilized approach	Using of FAHP and F-TOPSIS for maintenance policy selection in textile industry Presented the F-VIKOR and FMCDM for measuring of software development projects
Hsia, Chen, and Chen (2008)	Utilized approach	performance Proposed FAHP and FMCDM to determine the relative weights between each independent common factor

## Table 9 (continued)

Authors	Type of study	Tools and approaches
Önüt, Kara, and Efendigil (2008)	Utilized approach	Application of FAHP and F-TOPSIS approaches for the selection of machine tools
Sheu (2008)	Utilized approach	Application of FAHP, FMCDM and TOPSIS in global logistics management
Lee et al. (2008) Önüt and Soner (2008)	Utilized approach Utilized approach Utilized approach	Implemented of FAHP and AHP for evaluation of hydrogen technology sector in Korea Applied F-TOPSIS and FAHP to solve the solid waste transshipment site selection
Ferçin (2008)	Utilized approach	Applied F-AHP, FTOPSIS for assessment of mornation sharing decision problems
Ertuğrul and Karakaşoğlu (2008)	Utilized approach	Applied FAHP and F-TOPSIS for selection of facility location
Gharehgozli, Rabbani, Zaerpour, and Razmi	Utilized approach	Employed TOPSIS and AHP for manufacturing system capacity
(2008) Kwok, Zhou, Zhang, and Ma (2007) Bilsel, Büyüközkan, and Ruan (2006)	Approach developed Utilized approach	Using of fuzzy OWA operator and AHP for selection of students IS group projects
Tesfamariam and Sadiq (2006)	Utilized approach	Used FAHP and AHP for vagueness type uncertainty in risk environments
Chen and Tzeng (2004)	Approach developed	Applied FAHP and FTOPSIS for selection expatriate host country
Tsaur, Chang, and Yen (2002)	Utilized approach	Using FAHP and F-TOPSIS for performance evaluation of three airline service quality

|--|

Author Distribution by Nationality.

No.	Country	Frequency	Percent (%)	No.	Country	Frequency	Percent (%)
1	Taiwan	232	21.46	31	Tunisia	4	0.37
2	Turkey	132	12.21	32	Saudi Arabia	3	0.28
3	Iran	121	11.19	33	Montenegro	3	0.28
4	China	116	10.73	34	Singapore	3	0.28
5	India	74	6.85	35	Jordan	2	0.19
6	USA	46	4.26	36	Oman	2	0.19
7	Republic of Korea	33	3.05	37	Denmark	2	0.19
8	UK	32	2.96	38	Slovenia	2	0.19
9	Italy	27	2.50	39	Kuwait	2	0.19
10	Lithuania	22	2.04	40	Chile	2	0.19
11	Canada	22	2.04	41	Bosnia and Herzegovina	1	0.09
12	Spain	20	1.85	42	Hungary	1	0.09
13	Australia	17	1.57	43	Austria	1	0.09
14	Malaysia	18	1.67	44	Israel	1	0.09
15	Greece	14	1.30	45	Romania	1	0.09
16	Hong Kong	12	1.11	46	Cuba	1	0.09
17	France	12	1.11	47	Indonesia	1	0.09
18	Poland	11	1.02	48	Yugoslavia	1	0.09
19	Finland	10	0.93	49	Sweden	1	0.09
20	Portugal	9	0.83	50	Norway	1	0.09
21	Japan	7	0.65	51	Pakistan	1	0.09
22	Germany	7	0.65	52	New Zealand	1	0.09
23	Netherlands	7	0.65	53	Bangladesh	1	0.09
24	South Africa	6	0.56	54	Cyprus	1	0.09
25	Belgium	6	0.56	55	Morocco	1	0.09
26	Serbia	6	0.56	56	Algeria	1	0.09
27	Brazil	5	0.46	57	Ireland	1	0.09
28	Mexico	5	0.46	58	Argentina	1	0.09
29	Egypt	4	0.37	59	Philippines	1	0.09
30	Thailand	4	0.37				



Fig. 2. Distribution of publications year.

with other methods, 11 papers have employed ELECTRE and combined it with other methods, 13 papers have employed DEMATEL and combined it with other methods, 10 papers have employed PROMETHEE and combined it with other methods, and 32 papers have employed other combined tools and approaches such as ARAS, WASPAS. In the category of tools and application, researchers in management and business fields have applied 45 papers fuzzy TOPSIS and mixed it with other methods, 64 papers have used fuzzy AHP and mixed it with other methods, 15 papers have used fuzzy ANP and mixed it with other methods, 13 papers have used fuzzy VIKOR and mixed it with other methods, six papers have used fuzzy DEMATEL and mixed it with other methods, and 15 papers have used other mixed tools and applications such as ARAS-F, F-ELECTRE, F-COPRAS and so on. In case of the implementation of tools and applications in this category, 15 cases have utilized fuzzy TOPSIS and integrated it with other methods, 32 cases have utilized fuzzy AHP and integrated it with other methods; eight papers have utilized fuzzy DEMATEL and integrated it with other methods, five studies have utilized fuzzy VIKOR and

integrated it with other methods. Question five was; which countries have published these fuzzy MCDM tools based on the number of publications in these four fields? The results related to this question were shown in Table 1). According to this table, the most contributing countries are Taiwan (21.46%), Turkey (12.21%), Iran (11.19%) and China (10.73%). The results for the last question were presented in Table 10. Question six was; which year authors have further published fuzzy MCDM tools based on frequency in the four fields? Accordingly, it can be indicated that, the use of these tools and approaches in 1994 was four papers and this number increased to 10 papers in 1996. Surprisingly, from 2008 to 2009, the numbers of studies have dramatically increased. Although the use of fuzzy MCDM and MCDM tools and approaches have increased in each year, the number of those papers in 2012 (171) have decreased compared to 2011 (176). Another interesting result in this table is about 2013 which previous studies have applied tools and techniques more than other years. This year has the highest number of publications (188).

## 6. Conclusion

In decision-making fuzzy applications and theories, different modeling techniques have been offered, a number of suitable approaches have been provided for modeling decision aiding, help is provided for the development of alternatives as they consider the complexity of the process. Choosing a problem solution approach and a model is dependent upon the actors that are involved in the process of decision making, desired goals, available information, time, and so on. A number of branches of the fuzzy decision theory have departed from the stand expected utility paradigm. The most important advantage of the fuzzy multiple criteria methods is their capability of addressing the problems that are marked by different conflicting interests. Using these techniques, actors are capable of solving the problems that are not possible to be solved by the use of common optimization models. This review paper is mainly focused on the overview of the utilization of fuzzy decision support tools, e.g., recent developments of fuzzy models of multicriteria decision analysis. These tools are being employed increasingly for the evaluation of alternatives and comparative analysis. Moreover, a number of significant concepts are discussed, which have not been addressed in previous studies. We provide a systematically review of FMCDM which classify papers in four difference main fields including engineering, science, technology and science. Several significant papers in FMCDM issues are introduced by this paper.

In this paper, the literature was reviewed for the classification and interpretation of the emerging issues that make use of the FMCDM methodology. In the present review, a total of 413 papers were collected from 150 journals, published since 1994, and they were categorized into four main fields. The papers were classified based on the journal's name, publication year, authors' nationality, application areas, and other combined FMCDM methods. This paper contributed to the development of a classification scheme focusing on practical considerations, reviewing structurally the literature to create a guide for further studies on the FMCDM methods, and the identification of issues for future studies. Additionally, in our study, two new perspectives are taken into consideration to review the articles, namely categorization of the articles into four main fields (business, science, engineering, and technology) and examination of the type of study (FMCDM utilizing research, FMCDM developing research and FMCDM proposing research).

Generally, the FMCDM methodology has been used successfully in various applications and industrial sectors with different subjects and terms; although, interdisciplinary and social decision problems should be further emphasized. Future study on the fuzzy MCDM anatomy can be further developed. In this study, a number of techniques have been studied as fuzzy individual techniques and they are integrated or combined with other techniques; however, many other conventional MCDM techniques have not been studied. Another recommendation for future research is the investigation on the distinct differences and similarities among fuzzy MCDM methods. The insights that were provided in the present review help channel research efforts and fulfill practitioners' and researchers' requirements for an easy reference to fuzzy MCDM publications and studies.

This study has some major limitations that can be considered as recommendations for future studies. First, this review is focused on the use of fuzzy DM techniques. Articles published in late 2014, if any, are not included in the present paper due to the limitation of reporting time. A future review can be expanded further in scope. In addition: our paper more focuses on four main fields of engineering, management and business, science and technology. In this regard, future studies can use this paper for classify based on different sub-fields and sub-areas. Another limitation is that the data were collected from journals, not including papers conference papers, textbooks, doctoral and master dissertations and thesis and unpublished papers in the FMCDM issues. As a result, in a future study, data can be collected from these scholarly journals and the obtained results can be compared with our results. The next limitation is that the all of papers were found on English language journals; then, the scholarly journals in the other languages were not involved in our review paper. It may mean that our paper is not complete; however, we believe that our paper comprehensively reviews and includes most of the papers presented by high-ranking journals. As a result, our review paper can provides a better understanding of FMCDM methods for future academic scholars. This study is hoped to be employed by academics and managers as a basis for further research, help practitioners make more appropriate decisions using these techniques, and guide scholars to enhance these methodologies. This paper selected and summarized carefully those papers that were available in some available publishers in web of science, although, a number of relevant outlets may have remained outside the scope of this study. Therefore, future studies can review those papers which did not mentioned in this review paper.

Recently, development of hybrid and modular methods is becoming increasingly important. They are based on previously developed well-known methods, such as FTOPSIS, FSAW, FDEA, FAHP, FANP, FVIKOR, FDEMATEL, FDEA, FPROMETHEE, FELECTRE and their modification, by applying fuzzy and gray number theory. Relatively recently developed MCDM methods, such as COPRAS, ARAS-F, MOORA, MULTIMOORA, SWARA and WASPAS are rapidly developed and applied to solve real life problems. In order to help researchers and practitioners interested in hybrid FMCDM techniques and applications of hybrid FMCDM methods, it is necessary to publish reviews on these issues in future. As another limitation the paper presents synopsis of numerous publications, which describe the use of FMCDM methods in journals and some of the relatively recently developed methods. However, this review does not cover recent methods that have not yet been reviewed in books.

#### References

- Abdullah, L., & Najib, L. (2014a). A new preference scale of intuitionistic fuzzy analytic hierarchy process in multi-criteria decision making problems. *Journal* of Intelligent and Fuzzy Systems, 26, 1039–1049.
- Abdullah, L., & Najib, L. (2014b). Sustainable energy planning decision using the intuitionistic fuzzy analytic hierarchy process: Choosing energy technology in Malaysia. *International Journal of Sustainable Energy*, 1–18.
- Afshar, A., Mariño, M. A., Saadatpour, M., & Afshar, A. (2011). Fuzzy TOPSIS multicriteria decision analysis applied to Karun reservoirs system. Water Resources Management, 25, 545–563.

- Akadiri, P. O., Olomolaiye, P. O., & Chinyio, E. A. (2013). Multi-criteria evaluation model for the selection of sustainable materials for building projects. Automation in Construction, 30, 113-125.
- Akdag, H., Kalaycı, T., Karagöz, S., Zülfikar, H., & Giz, D. (2014). The evaluation of hospital service quality by fuzzy MCDM. Applied Soft Computing, 23, 239-248.
- Ali, M. A., Shil, N. C., Nine, M. Z., Khan, M., & Hoque, M. H. (2010). Vendor selection using fuzzy integration. International Journal of Management Science and Engineering Management, 5, 376-382.
- Aliei, M., Sazvar, A., & Ashrafi, B. (2012). Assessment of information technology effects on management of supply chain based on fuzzy logic in Iran tail industries. The International Journal of Advanced Manufacturing Technology, 63, 215-223.
- Altrock, C. V. (1997). Fuzzy logic and NeuroFuzzy applications in business and finance. Prentice-Hall Inc.
- Altuntas, S., Selim, H., & Dereli, T. (2014). A fuzzy DEMATEL-based solution approach for facility layout problem: A case study. The International Journal of Advanced Manufacturing Technology, 1–23.
- Amiri, M. P. (2010). Project selection for oil-fields development by using the AHP and fuzzy TOPSIS methods. Expert Systems with Applications, 37, 6218–6224.
- Amiri, M., & Golozari, F. (2011). Application of fuzzy multi-attribute decision making in determining the critical path by using time, cost, risk, and quality criteria. The International Journal of Advanced Manufacturing Technology, 54, 393-401
- Amirzadeh, R., & Shoorvarzy, M. R. (2013). Prioritizing service quality factors in Iranian Islamic banking using a fuzzy approach. International Journal of Islamic and Middle Eastern Finance and Management, 6, 64-78.
- An, M., Chen, Y., & Baker, C. J. (2011). A fuzzy reasoning and fuzzy-analytical hierarchy process based approach to the process of railway risk information: A railway risk management system. Information Sciences, 181, 3946-3966.
- Anojkumar, L., Ilangkumaran, M., & Sasirekha, V. (2014). Comparative analysis of MCDM methods for pipe material selection in sugar industry. Expert Systems with Applications, 41, 2964-2980.
- Antuchevičiene, J. (2005). Evaluation of alternatives applying TOPSIS method in a fuzzy environment. Technological and Economic Development of Economy, 11, 242-247
- Arabzad, S. M., Ghorbani, M., Razmi, J., & Shirouyehzad, H. (2014). Employing fuzzy TOPSIS and SWOT for supplier selection and order allocation problem. The International Journal of Advanced Manufacturing Technology, 1-16.
- Arslan, M., & Çunkaş, M. (2012). Performance evaluation of sugar plants by fuzzy technique for order performance by similarity to ideal solution (TOPSIS). Cybernetics and Systems, 43, 529-548.
- Ashtiani, B., Haghighirad, F., & Makui, A. (2009). Extension of fuzzy TOPSIS method based on interval-valued fuzzy sets. Applied Soft Computing, 9, 457-461.
- Atalay, K. D., & Eraslan, E. (2012). Multi-criteria usability evaluation of electronic devices in a fuzzy environment. Human Factors and Ergonomics in Manufacturing & Service Industries.
- Atanassov, K. T. (1986). Intuitionistic fuzzy sets. Fuzzy Sets and Systems, 20, 87-96. Athanasopoulos, G., Riba, C. R., & Athanasopoulou, C. (2009). A decision support
- system for coating selection based on fuzzy logic and multi-criteria decision making. Expert Systems with Applications, 36, 10848-10853.
- Avikal, S., Jain, R., & Mishra, P. (2014). A Kano model, AHP and M-TOPSIS methodbased technique for disassembly line balancing under fuzzy environment. Applied Soft Computing.
- Avikal, S., Mishra, P., & Jain, R. (2014). A fuzzy AHP and PROMETHEE method-based heuristic for disassembly line balancing problems. International Journal of Production Research, 52, 1306-1317.
- Awasthi, A., & Chauhan, S. S. (2011). Using AHP and Dempster-Shafer theory for evaluating sustainable transport solutions. Environmental Modelling & Software, 26, 787-796.
- Awasthi, A., Chauhan, S. S., & Omrani, H. (2011). Application of fuzzy TOPSIS in evaluating sustainable transportation systems. Expert Systems with Applications, 38 12270-12280
- Awasthi, A., Chauhan, S. S., Omrani, H., & Panahi, A. (2011). A hybrid approach based on SERVQUAL and fuzzy TOPSIS for evaluating transportation service quality. Computers & Industrial Engineering, 61, 637–646.
- Ayağ, Z. (2005). A fuzzy AHP-based simulation approach to concept evaluation in a
- NPD environment. *IIE Transactions*, *37*, 827–842. Ayağ, Z. (2010). A combined fuzzy AHP-simulation approach to CAD software selection. *International Journal of General Systems*, *39*, 731–756.
- Ayağ, Z., & Gürcan Özdemir, R. (2012). Evaluating machine tool alternatives through modified TOPSIS and alpha-cut based fuzzy ANP. International Journal of Production Economics, 140, 630–636.
- Ayağ, Z., & Özdemir, R. (2007). An intelligent approach to ERP software selection through fuzzy ANP. International Journal of Production Research, 45, 2169-2194.
- Ayağ, Z., & Özdemir, R. G. (2009). A hybrid approach to concept selection through fuzzy analytic network process. Computers & Industrial Engineering, 56, 368-379
- Ayağ, Z., & Özdemir, R. G. (2011). An intelligent approach to machine tool selection through fuzzy analytic network process. Journal of Intelligent Manufacturing, 22, 163-177.
- Aydogan, E. K. (2011). Performance measurement model for Turkish aviation firms using the rough-AHP and TOPSIS methods under fuzzy environment. Expert Systems with Applications, 38, 3992-3998.
- Azadeh, A., Nazari-Shirkouhi, S., Hatami-Shirkouhi, L., & Ansarinejad, A. (2011). A unique fuzzy multi-criteria decision making: Computer simulation approach for productive operators' assignment in cellular manufacturing systems with

uncertainty and vagueness. The International Journal of Advanced Manufacturing Technology, 56, 329-343.

- Azadeh, A., Osanloo, M., & Ataei, M. (2010). A new approach to mining method selection based on modifying the Nicholas technique. Applied Soft Computing, 10.1040-1061.
- Azadnia, A. H., Saman, M. Z. M., & Wong, K. Y. (2014). Sustainable supplier selection and order lot-sizing: An integrated multi-objective decision-making process. International Journal of Production Research, 1-26.
- Bagočius, V., Zavadskas, E. K., & Turskis, Z. (2014). Multi-person selection of the best wind turbine based on the multi-criteria integrated additive-multiplicative utility function. Journal of Civil Engineering and Management, 20, 590-599.
- Bairagi, B., Dey, B., Sarkar, B., & Sanyal, S. (2014). Selection of robot for automated foundry operations using fuzzy multi-criteria decision making approaches. International Journal of Management Science and Engineering Management, 9, 221-232.
- Baležentis, T., & Baležentis, A. (2014). A survey on development and applications of the multi-criteria decision making method MULTIMOORA. Journal of Multi-Criteria Decision Analysis, 21, 209-222.
- Baležentis, A., Baležentis, T., & Misiunas, A. (2012). An integrated assessment of Lithuanian economic sectors based on financial ratios and fuzzy MCDM methods. Technological and Economic Development of Economy, 18, 34-53.
- Baležentis, A., Valkauskas, R., & Baležentis, T. (2010). Evaluating situation of Lithuania in the European Union: Structural indicators and MULTIMOORA method. Technological and Economic Development of Economy, 578-602.
- Banaitiene, N., Banaitis, A., Kaklauskas, A., & Zavadskas, E. K. (2008). Evaluating the life cycle of a building: A multivariant and multiple criteria approach. Omega, 36, 429–441.
- Bashiri, M., Badri, H., & Hejazi, T. H. (2011). Selecting optimum maintenance strategy by fuzzy interactive linear assignment method. Applied Mathematical Modelling, 35, 152-164.
- Bashiri, M., & Hosseininezhad, S. J. (2009). A fuzzy group decision support system for multifacility location problems. The International Journal of Advanced Manufacturing Technology, 42, 533–543.
- Baykasoglu, A., & Durmusoglu, Z. D. (2014). A hybrid MCDM for private primary school assessment using DEMATEL based on ANP and fuzzy cognitive map. International Journal of Computational Intelligence Systems, 7, 615-635.
- Baykasoğlu, A., Kaplanoğlu, V., Durmuşoğlu, Z. D., & Şahin, C. (2013). Integrating fuzzy DEMATEL and fuzzy hierarchical TOPSIS methods for truck selection. Expert Systems with Applications, 40, 899–907.
- Bayrakdaroğlu, A., & Yalçın, N. (2013). A fuzzy multi-criteria evaluation of the operational risk factors for the state-owned and privately-owned commercial banks in turkey. Human and Ecological Risk Assessment: An International Journal, 19, 443-461.
- Behzadian, M., Kazemzadeh, R. B., Albadvi, A., & Aghdasi, M. (2010). PROMETHEE: A comprehensive literature review on methodologies and applications. European Journal of Operational Research, 200, 198–215.
- Behzadian, M., Khanmohammadi Otaghsara, S., Yazdani, M., & Ignatius, J. (2012). A state-of the-art survey of TOPSIS applications. Expert Systems with Applications, 39, 13051-13069.
- Bellman, R. E., & Zadeh, L. A. (1970). Decision-making in a fuzzy environment. Management Science, 17, B-141-B-164.
- Benítez, J. M., Martín, J. C., & Román, C. (2007). Using fuzzy number for measuring quality of service in the hotel industry. Tourism Management, 28, 544-555.
- Bilgen, B., & Sen, M. (2012). Project selection through fuzzy analytic hierarchy process and a case study on Six Sigma implementation in an automotive industry. Production Planning & Control, 23, 2-25.
- Bilsel, R. U., Büyüközkan, G., & Ruan, D. (2006). A fuzzy preference-ranking model for a quality evaluation of hospital web sites. International Journal of Intelligent Systems 21 1181-1197
- Boran, F. E., Genç, S., & Akay, D. (2011). Personnel selection based on intuitionistic fuzzy sets. Human Factors and Ergonomics in Manufacturing & Service Industries, 21 493-503
- Boran, F. E., Genç, S., Kurt, M., & Akay, D. (2009). A multi-criteria intuitionistic fuzzy group decision making for supplier selection with TOPSIS method. *Expert* Systems with Applications, 36, 11363–11368.
- Boran, S., & Goztepe, K. (2010). Development of a fuzzy decision support system for commodity acquisition using fuzzy analytic network process. Expert Systems with Applications, 37, 1939–1945.
- Bozbura, F. T., Beskese, A., & Kahraman, C. (2007). Prioritization of human capital measurement indicators using fuzzy AHP. Expert Systems with Applications, 32, 1100-1112.
- Brauers, W. K. M., & Zavadskas, E. K. (2006). The MOORA method and its application to privatization in a transition economy. *Control and Cybernetics*, 35, 445-469.
- Brauers, W. K. M., & Zavadskas, E. K. (2010). Project management by MULTIMOORA as an instrument for transition economies. Technological and Economic Development of Economy, 5–24.
- Büyüközkan, G. (2012). An integrated fuzzy multi-criteria group decision-making approach for green supplier evaluation. International Journal of Production Research, 50, 2892-2909.
- Büyüközkan, G., Arsenyan, J., & Ruan, D. (2012). Logistics tool selection with twophase fuzzy multi criteria decision making: A case study for personal digital assistant selection. Expert Systems with Applications, 39, 142-153.
- Büyüközkan, G., & Çifçi, G. (2011). A novel fuzzy multi-criteria decision framework for sustainable supplier selection with incomplete information. Computers in Industry, 62, 164-174.

- Büyüközkan, G., & Çifçi, G. (2012a). A combined fuzzy AHP and fuzzy TOPSIS based strategic analysis of electronic service quality in healthcare industry. *Expert Systems with Applications*, 39, 2341–2354.
- Büyüközkan, G., & Çifçi, G. (2012b). A novel hybrid MCDM approach based on fuzzy DEMATEL, fuzzy ANP and fuzzy TOPSIS to evaluate green suppliers. *Expert Systems with Applications*, 39, 3000–3011.
- Büyüközkan, G., Çifçi, G., & Güleryüz, S. (2011). Strategic analysis of healthcare service quality using fuzzy AHP methodology. *Expert Systems with Applications*, 38, 9407–9424.
- Büyüközkan, G., & Feyziog~lu, O. (2004). A fuzzy-logic-based decision-making approach for new product development. *International Journal of Production Economics*, 90, 27–45.
- Büyüközkan, G., & Ruan, D. (2008). Evaluation of software development projects using a fuzzy multi-criteria decision approach. *Mathematics and Computers in Simulation*, 77, 464–475.
- Cakir, O., & Canbolat, M. S. (2008). A web-based decision support system for multicriteria inventory classification using fuzzy AHP methodology. *Expert Systems* with Applications, 35, 1367–1378.
- Calabrese, A., Costa, R., & Menichini, T. (2013). Using fuzzy AHP to manage intellectual capital assets: An application to the ICT service industry. *Expert Systems with Applications*, 40, 3747–3755.
- Cavallaro, F. (2010). Fuzzy TOPSIS approach for assessing thermal-energy storage in concentrated solar power (CSP) systems. *Applied Energy*, 87, 496–503.
- Cebeci, U. (2009). Fuzzy AHP-based decision support system for selecting ERP systems in textile industry by using balanced scorecard. *Expert Systems with Applications*, 36, 8900–8909.
- Chamodrakas, I., Batis, D., & Martakos, D. (2010). Supplier selection in electronic marketplaces using satisficing and fuzzy AHP. Expert Systems with Applications, 37, 490–498.
- Chamodrakas, I., Leftheriotis, I., & Martakos, D. (2011). In-depth analysis and simulation study of an innovative fuzzy approach for ranking alternatives in multiple attribute decision making problems based on TOPSIS. *Applied Soft Computing*, 11, 900–907.
- Chan, F., Chan, H., Chan, M., & Humphreys, P. (2006). An integrated fuzzy approach for the selection of manufacturing technologies. *The International Journal of Advanced Manufacturing Technology*, 27, 747–758.
- Chan, F. T., & Kumar, N. (2007). Global supplier development considering risk factors using fuzzy extended AHP-based approach. Omega, 35, 417–431.
- Chang, C.-W., Horng, D.-J., & Lin, H.-L. (2011a). Combing fuzzy theory and analytic network concept to construct a model in management decision-making. *Journal* of Information and Optimization Sciences, 32, 777–798.
- Chang, C.-W., Horng, D.-J., & Lin, H.-L. (2011b). A measurement model for experts knowledge-based systems algorithm using fuzzy analytic network process. *Expert Systems with Applications*, 38, 12009–12017.
- Chang, Y.-N., Hung, C.-Y., Li, M.-T., & Hsu, C.-P. (2008). Application of fuzzy MCDM to priority setting in agricultural biotechnology industries. *Journal of Statistics and Management Systems*, 11, 1059–1079.
- Chang, C.-W., Wu, C.-R., & Chen, H.-C. (2008). Using expert technology to select unstable slicing machine to control wafer slicing quality via fuzzy AHP. *Expert Systems with Applications*, 34, 2210–2220.
- Chatterjee, P., Manikrao Athawale, V., & Chakraborty, S. (2010). Selection of industrial robots using compromise ranking and outranking methods. *Robotics* and Computer-Integrated Manufacturing, 26, 483–489.
- Chen, C.-T. (2000a). Extensions of the TOPSIS for group decision-making under fuzzy environment. *Fuzzy Sets and Systems*, 114, 1–9.
- Chen, C.-T. (2000b). A fuzzy MCDM method based on interval analysis for solving plant location selection problem. *Journal of the Chinese Institute of Industrial Engineers*, 17, 111–120.
- Chen, Y.-H., & Chao, R.-J. (2012). Supplier selection using consistent fuzzy preference relations. *Expert Systems with Applications*, 39, 3233–3240.
- Chen, J.-K., & Chen, I. (2010a). Aviatic innovation system construction using a hybrid fuzzy MCDM model. Expert Systems with Applications, 37, 8387–8394.
- Chen, J.-K., & Chen, I. (2010b). Using a novel conjunctive MCDM approach based on DEMATEL, fuzzy ANP, and TOPSIS as an innovation support system for Taiwanese higher education. *Expert Systems with Applications*, 37, 1981–1990.
- Chen, J.-K., & Chen, I. (2012). A network hierarchical feedback system for Taiwanese universities based on the integration of total quality management and innovation. *Applied Soft Computing*, *12*, 2394–2408.
- Chen, L.-H., & Hung, C.-C. (2010). An integrated fuzzy approach for the selection of outsourcing manufacturing partners in pharmaceutical R&D. *International Journal of Production Research*, 48, 7483–7506.
- Chen, S.-J. J., Hwang, C.-L., Beckmann, M. J., & Krelle, W. (1992). Fuzzy multiple attribute decision making: Methods and applications. Springer-Verlag New York Inc.
- Chen, C.-T., Lin, C.-T., & Huang, S.-F. (2006). A fuzzy approach for supplier evaluation and selection in supply chain management. *International Journal of Production Economics*, 102, 289–301.
- Chen, H. H., & Pang, C. (2010). Organizational forms for knowledge management in photovoltaic solar energy industry. *Knowledge-Based Systems*, 23, 924–933.
- Chen, S.-M., & Tan, J.-M. (1994). Handling multicriteria fuzzy decision-making problems based on vague set theory. *Fuzzy Sets and Systems*, 67, 163–172.
- Chen, T.-Y., & Tsao, C.-Y. (2008). The interval-valued fuzzy TOPSIS method and experimental analysis. *Fuzzy Sets and Systems*, 159, 1410–1428.
- Chen, M.-F., & Tzeng, G.-H. (2004). Combining grey relation and TOPSIS concepts for selecting an expatriate host country. *Mathematical and Computer Modelling*, 40, 1473–1490.

- Chen, L. Y., & Wang, T.-C. (2009). Optimizing partners' choice in IS/IT outsourcing projects: The strategic decision of fuzzy VIKOR. International Journal of Production Economics, 120, 233–242.
- Chen, M.-K., & Wang, S.-C. (2010). The use of a hybrid fuzzy-Delphi-AHP approach to develop global business intelligence for information service firms. *Expert Systems with Applications*, 37, 7394–7407.
- Chen, S. H., Wang, P. W., Chen, C. M., & Lee, H. T. (2010). An analytic hierarchy process approach with linguistic variables for selection of an R&D strategic alliance partner. *Computers & Industrial Engineering*, *58*, 278–287.
- Chen, T.-Y., Wang, H.-P., & Lu, Y.-Y. (2011). A multicriteria group decision-making approach based on interval-valued intuitionistic fuzzy sets: A comparative perspective. *Expert Systems with Applications*, 38, 7647–7658.
- Chen, Y.-H., Wang, T.-C., & Wu, C.-Y. (2011). Strategic decisions using the fuzzy PROMETHEE for IS outsourcing. *Expert Systems with Applications*, 38, 13216–13222.
- Chen, Z., & Yang, W. (2011). An MAGDM based on constrained FAHP and FTOPSIS and its application to supplier selection. *Mathematical and Computer Modelling*, 54, 2802–2815.
- Cheng, C.-H. (1997). Evaluating naval tactical missile systems by fuzzy AHP based on the grade value of membership function. *European Journal of Operational Research*, *96*, 343–350.
- Cheng, J. Z., Chen, P.-T., & Yu, H.-C. D. (2005). Establishing a MAN access strategy for future broadband service. A fuzzy MCDM analysis of SONET/SDH and Gigabit Ethernet. *Technovation*, 25, 557–567.
- Cheng, C.-H., & Mon, D.-L. (1994). Evaluating weapon system by analytical hierarchy process based on fuzzy scales. *Fuzzy Sets and Systems*, 63, 1–10.
- Chiang, D. M., Guo, R.-S., & Pai, F.-Y. (2008). Improved customer satisfaction with a hybrid dispatching rule in semiconductor back-end factories. *International Journal of Production Research*, 46, 4903–4923.
- Chiclana, F., Herrera, F., & Herrera-Viedma, E. (2000). The ordered weighted geometric operator: Properties and application in MCDM Problems. In Proceedings of 8th conference information processing and management of uncertainty in knowledge based systems (IPMU), Citeseer.
- Chiclana, F., Herrera, F., & Herrera-Viedma, E. (1998). Integrating three representation models in fuzzy multipurpose decision making based on fuzzy preference relations. *Fuzzy Sets and Systems*, 97, 33–48.
- Chiouy, C.-Y., Chou, S.-H., & Yeh, C.-Y. (2011). Using fuzzy AHP in selecting and prioritizing sustainable supplier on CSR for Taiwan's electronics industry. *Journal of Information and Optimization Sciences*, 32, 1135–1153.
- Cho, D. W., Lee, Y. H., Ahn, S. H., & Hwang, M. K. (2012). A framework for measuring the performance of service supply chain management. *Computers & Industrial Engineering*, 62, 801–818.
- Chou, W.-C., & Cheng, Y.-P. (2012). A hybrid fuzzy MCDM approach for evaluating website quality of professional accounting firms. *Expert Systems with Applications*, 39, 2783–2793.
- Chou, C.-H., Liang, G.-S., & Chang, H.-C. (2013). A fuzzy AHP approach based on the concept of possibility extent. *Ouality & Ouantity*, 47, 1–14.
- Chou, J.-S., Pham, A.-D., & Wang, H. (2013). Bidding strategy to support decisionmaking by integrating fuzzy AHP and regression-based simulation. *Automation* in Construction, 35, 517–527.
- Chou, Y.-C., Sun, C.-C., & Yen, H.-Y. (2012). Evaluating the criteria for human resource for science and technology (HRST) based on an integrated fuzzy AHP and fuzzy DEMATEL approach. *Applied Soft Computing*, *12*, 64–71.
- Choudhary, D., & Shankar, R. (2012). An STEEP-fuzzy AHP-TOPSIS framework for evaluation and selection of thermal power plant location: A case study from India. *Energy*, 42, 510–521.
   Chu, T.-C., & Lin, Y.-C. (2009). An interval arithmetic based fuzzy TOPSIS model.
- Chu, T.-C., & Lin, Y.-C. (2009). An interval arithmetic based fuzzy TOPSIS model. Expert Systems with Applications, 36, 10870–10876.
- da Costa Sousa, J. M., & Kaymak, U. (2001). Model predictive control using fuzzy decision functions. IEEE Transactions on Systems, Man, and Cybernetics, Part B: Cybernetics, 31, 54–65.
- Dabbaghian, M., Hewage, K., Reza, B., Culver, K., & Sadiq, R. (2014). Sustainability performance assessment of green roof systems using fuzzy-analytical hierarchy process (FAHP). International Journal of Sustainable Building Technology and Urban Development, 1–17.
- Dadelo, S., Turskis, Z., Zavadskas, E. K., & Dadeliene, R. (2014). Multi-criteria assessment and ranking system of sport team formation based on objectivemeasured values of criteria set. *Expert Systems with Applications*, 41, 6106–6113.
- Dağdeviren, M., & Yüksel, İ. (2008). Developing a fuzzy analytic hierarchy process (AHP) model for behavior-based safety management. *Information Sciences*, 178, 1717–1733.
- Dalalah, D., Hayajneh, M., & Batieha, F. (2011). A fuzzy multi-criteria decision making model for supplier selection. *Expert Systems with Applications*, 38, 8384–8391.
- Das, M. C., Sarkar, B., & Ray, S. (2012). A framework to measure relative performance of Indian technical institutions using integrated fuzzy AHP and COPRAS methodology. Socio-Economic Planning Sciences, 46, 230–241.
- Demirel, T., Muşdal, H., Demirel, N. C., & Yücenur, G. N. (2009). Multi-criteria evaluation of land cover policies using fuzzy AHP and fuzzy ANP: The case of Turkey. *Human and Ecological Risk Assessment*, 15, 746–764.
- Demirel, N. Ç., Yücenur, G. N., Demirel, T., & Muşdal, H. (2012). Risk-based evaluation of Turkish agricultural strategies using fuzzy AHP and fuzzy ANP. *Human and Ecological Risk Assessment: An International Journal*, 18, 685–702.
- Deng, Y., & Chan, F. T. (2011). A new fuzzy dempster MCDM method and its application in supplier selection. *Expert Systems with Applications*, 38, 9854–9861.

- Devi, K., & Yadav, S. P. (2013). A multicriteria intuitionistic fuzzy group decision making for plant location selection with ELECTRE method. *The International Journal of Advanced Manufacturing Technology*, 66, 1219–1229.
- Dimova, L., Sevastianov, P., & Sevastianov, D. (2006). MCDM in a fuzzy setting: Investment projects assessment application. *International Journal of Production Economics*, 100, 10–29.
- Dincer, H., & Hacioglu, U. (2013). Performance evaluation with fuzzy VIKOR and AHP method based on customer satisfaction in Turkish banking sector. *Kybernetes*, 42, 1072–1085.
- Dragović, I., Turajlić, N., Radojević, D., & Petrović, B. (2014). Combining boolean consistent fuzzy logic and AHP illustrated on the web service selection problem. *International Journal of Computational Intelligence Systems*, 7, 84–93.
- Dubois, D., & Prade, H. (1980). Systems of linear fuzzy constraints. *Fuzzy Sets and Systems*, 3, 37–48.
- Durán, O. (2011). Computer-aided maintenance management systems selection based on a fuzzy AHP approach. Advances in Engineering Software, 42, 821–829. Dursun, M., & Karsak, E. E. (2010). A fuzzy MCDM approach for personnel selection.
- *Expert Systems with Applications*, 37, 4324–4330. Duru, O., Bulut, E., & Yoshida, S. (2012). Regime switching fuzzy AHP model for choice-varying priorities problem and expert consistency prioritization: A cubic
- fuzzy-priority matrix design. Expert Systems with Applications, 39, 4954–4964. Dymova, L., Sevastjanov, P., & Tikhonenko, A. (2013a). An approach to generalization of fuzzy TOPSIS method. Information Sciences, 238, 149–162.
- Dymova, L., Sevastjanov, P., & Tikhonenko, A. (2013b). Two-criteria method for comparing real-valued and interval-valued intuitionistic fuzzy values. *Knowledge-Based Systems*, 45, 166–173.
- Ekmekçioğlu, M., Kaya, T., & Kahraman, C. (2010). Fuzzy multicriteria disposal method and site selection for municipal solid waste. *Waste Management*, 30, 1729–1736.
- Eraslan, E., & Iç, Y. T. (2011). A multi-criteria approach for determination of investment regions: Turkish case. *Industrial Management & Data Systems*, 111, 890–909.
- Ertay, T., Kahraman, C., & Kaya, İ. (2013). Evaluation of renewable energy alternatives using MACBETH and fuzzy AHP multicriteria methods: The case of Turkey. *Technological and Economic Development of Economy*, 19, 38–62.
- Ertuğrul, İ., & Karakaşoğlu, N. (2008). Comparison of fuzzy AHP and fuzzy TOPSIS methods for facility location selection. The International Journal of Advanced Manufacturing Technology, 39, 783–795.
- Ertuğrul, İ., & Karakaşoğlu, N. (2009). Performance evaluation of Turkish cement firms with fuzzy analytic hierarchy process and TOPSIS methods. *Expert Systems* with Applications, 36, 702–715.
- Fodor, J. C., & Roubens, M. (1995). Characterization of weighted maximum and some related operations. *Information Sciences*, 84, 173–180.
- Fouladgar, M. M., Yazdani-Chamzini, A., Lashgari, A., Zavadskas, E. K., & Turskis, Z. (2012). Maintenance strategy selection using AHP and COPRAS under fuzzy environment. *International Journal of Strategic Property Management*, 16, 85–104.
- Fouladgar, M. M., Yazdani-Chamzini, A., Zavadskas, E. K., & Haji Moini, S. H. (2012). A new hybrid model for evaluating the working strategies: Case study of construction company. *Technological and Economic Development of Economy*, 18, 164–188.
- Fu, H. P., Chang, T. H., Kao, L. J., Chiu, C. C., & Lu, C. C. (2013). Combining multicriteria decision-making tools to identify critical success factors that affect the performance of training course projects. Systems Research and Behavioral Science.
- Fu, H.-P., Chu, K.-K., Chao, P., Lee, H.-H., & Liao, Y.-C. (2011). Using fuzzy AHP and VIKOR for benchmarking analysis in the hotel industry. *The Service Industries Journal*, 31, 2373–2389.
- Ganguly, K. K., & Guin, K. K. (2013). A fuzzy AHP approach for inbound supply risk assessment. Benchmarking: An International Journal, 20, 129–146.
- Gao, L. & Hailu, A. (2012). Ranking management strategies with complex outcomes: An AHP-fuzzy evaluation of recreational fishing using an integrated agentbased model of a coral reef ecosystem. *Environmental Modelling & Software*, 31, 3–18.
- Gao, T., Jin, R. C., Song, J. Y., Xu, T. B., & Wang, L. D. (2012). Energy-efficient cluster head selection scheme based on multiple criteria decision making for wireless sensor networks. Wireless Personal Communications, 63, 871–894.
- Gharehgozli, A., Rabbani, M., Zaerpour, N., & Razmi, J. (2008). A comprehensive decision-making structure for acceptance/rejection of incoming orders in maketo-order environments. *The International Journal of Advanced Manufacturing Technology*, 39, 1016–1032.
- Ghorabaee, M. K., Amiri, M., Sadaghiani, J. S., & Goodarzi, G. H. (2014). Multiple criteria group decision-making for supplier selection based on COPRAS method with interval type-2 fuzzy sets. *The International Journal of Advanced Manufacturing Technology*, 75, 1115–1130.
- Ghorbani, M., Mohammad Arabzad, S., & Shahin, A. (2013). A novel approach for supplier selection based on the Kano model and fuzzy MCDM. *International Journal of Production Research*, 51, 5469–5484.
- Gil-Lafuente, A. M., Merigó, J. M., & Vizuete, E. (2014). Analysis of luxury resort hotels by using the fuzzy analytic hierarchy process and the fuzzy Delphi method. *Economic Research-Ekonomska Istraživanja*, 27, 244–266.
- Gülgen, F. (2013). Road hierarchy with integration of attributes using fuzzy-AHP. *Geocarto International*, 1–21.
- Gumus, A. T. (2009). Evaluation of hazardous waste transportation firms by using a two step fuzzy-AHP and TOPSIS methodology. *Expert Systems with Applications*, 36, 4067–4074.

- Güngör, Z., Serhadlıoğlu, G., & Kesen, S. E. (2009). A fuzzy AHP approach to personnel selection problem. *Applied Soft Computing*, 9, 641–646.
- Gupta, R., Sachdeva, A., & Bhardwaj, A. (2012). Selection of logistic service provider using fuzzy PROMETHEE for a cement industry. *Journal of Manufacturing Technology Management*, 23, 899–921.
- Gürbüz, T., & Albayrak, Y. E. (2014). An engineering approach to human resources performance evaluation: Hybrid MCDM application with interactions. *Applied Soft Computing*, *21*, 365–375.
- Gürbüz, T., Albayrak, Y. E., & Alaybeyoğlu, E. (2014). Criteria weighting and 4P's planning in marketing using a fuzzy metric distance and AHP hybrid method. *International Journal of Computational Intelligence Systems*, 7, 94–104.
- Hadi-Vencheh, A., & Mohamadghasemi, A. (2011). A fuzzy AHP-DEA approach for multiple criteria ABC inventory classification. *Expert Systems with Applications*, 38, 3346–3352.
- Hadi-Vencheh, A., & Mohamadghasemi, A. (2014). A new hybrid fuzzy multi-criteria decision making model for solving the material handling equipment selection problem. *International Journal of Computer Integrated Manufacturing*, 1–17.
- Hajiagha, S. H. R., Mahdiraji, H. A., Zavadskas, E. K., & Hashemi, S. S. (2014). Fuzzy multi objective linear programming based on compromise Vikor method. International Journal of Information Technology & Decision Making.
- Halouani, N., Chabchoub, H., & Martel, J.-M. (2009). PROMETHEE-MD-2T method for project selection. European Journal of Operational Research, 195, 841–849.
- Hashemian, S. M., Behzadian, M., Samizadeh, R., & Ignatius, J. (2014). A fuzzy hybrid group decision support system approach for the supplier evaluation process. *The International Journal of Advanced Manufacturing Technology*, 1–13.
- Hatami-Marbini, A., Tavana, M., Moradi, M., & Kangi, F. (2013). A fuzzy group Electre method for safety and health assessment in hazardous waste recycling facilities. *Safety Science*, 51, 414–426.
- He, T., Ho, W., Man, C. L. K., & Xu, X. (2012). A fuzzy AHP based integer linear programming model for the multi-criteria transshipment problem. *The International Journal of Logistics Management*, 23, 159–179.
- Heo, E., Kim, J., & Boo, K.-J. (2010). Analysis of the assessment factors for renewable energy dissemination program evaluation using fuzzy AHP. *Renewable and Sustainable Energy Reviews*, 14, 2214–2220.
- Herrmann, R., & Herrmann, O. (2014). Hotel room rates under the influence of a large event: The Oktoberfest in Munich 2012. International Journal of Hospitality Management, 39, 21–28.
- Ho, W. (2008). Integrated analytic hierarchy process and its applications–a literature review. European Journal of Operational Research, 186, 211–228.
- Hong, D. H., & Choi, C.-H. (2000). Multicriteria fuzzy decision-making problems based on vague set theory. *Fuzzy Sets and Systems*, 114, 103–113.
- Hsia, T.-C., Chen, H.-T., & Chen, W.-H. (2008). Measuring the readability performance (RP) of aircraft maintenance technical orders by fuzzy MCDM method and RP index. Quality & Quantity, 42, 795–807.
- Hsieh, T.-Y., Lu, S.-T., & Tzeng, G.-H. (2004). Fuzzy MCDM approach for planning and design tenders selection in public office buildings. *International Journal of Project Management*, 22, 573–584.
- Hsu, C.-C., Liou, J. J., & Chuang, Y.-C. (2013). Integrating DANP and modified grey relation theory for the selection of an outsourcing provider. *Expert Systems with Applications*, 40, 2297–2304.
- Hsu, C.-H., Wang, F.-K., & Tzeng, G.-H. (2012). The best vendor selection for conducting the recycled material based on a hybrid MCDM model combining DANP with VIKOR. *Resources, Conservation and Recycling*, 66, 95–111.
- Huang, J.-H., & Peng, K.-H. (2012). Fuzzy Rasch model in TOPSIS: A new approach for generating fuzzy numbers to assess the competitiveness of the tourism industries in Asian countries. *Tourism Management*, 33, 456–465.
- Hung, S.-J. (2011). Activity-based divergent supply chain planning for competitive advantage in the risky global environment: A DEMATEL-ANP fuzzy goal programming approach. *Expert Systems with Applications*, 38, 9053–9062.
- Hwang, F. P., Chen, S. J., & Hwang, C. L. (1992). Fuzzy multiple attribute decision making: Methods and applications. Berlin Heidelberg: Springer.
- Hwang, C., & Yoon, K. (1981). Multiple attribute decision making: Methods and applications a state of the art survey. *Lecture notes in economics and mathematical systems* (vol. 186). New York, NY: Springer-Verlag.Berlin, Springer-Verlag.
- Iç, Y. T., & Yurdakul, M. (2010). Development of a quick credibility scoring decision support system using fuzzy TOPSIS. Expert Systems with Applications, 37, 567–574.
- Ilangkumaran, M., & Kumanan, S. (2009). Selection of maintenance policy for textile industry using hybrid multi-criteria decision making approach. *Journal of Manufacturing Technology Management*, 20, 1009–1022.
- Intepe, G., Bozdag, E., & Koc, T. (2013). The selection of technology forecasting method using a multi-criteria interval-valued intuitionistic fuzzy group decision making approach. Computers & Industrial Engineering, 65, 277–285.
- Inuiguchi, M., Ichihashi, H., & Tanaka, H. (1990). Fuzzy programming: A survey of recent developments. In Stochastic versus fuzzy approaches to multiobjective mathematical programming under uncertainty (pp. 45–68). Springer.
- Ishizaka, A., & Nguyen, N. H. (2013). Calibrated fuzzy AHP for current bank account selection. Expert Systems with Applications, 40, 3775–3783.
- Jaskowski, P., Biruk, S., & Bucon, R. (2010). Assessing contractor selection criteria weights with fuzzy AHP method application in group decision environment. *Automation in Construction*, *19*, 120–126.
- Javanbarg, M. B., Scawthorn, C., Kiyono, J., & Shahbodaghkhan, B. (2012). Fuzzy AHPbased multicriteria decision making systems using particle swarm optimization. *Expert Systems with Applications*, 39, 960–966.

- Jeng, D. J.-F., & Tzeng, G.-H. (2012). Social influence on the use of clinical decision support systems: Revisiting the unified theory of acceptance and use of technology by the fuzzy DEMATEL technique. *Computers & Industrial Engineering*, 62, 819–828.
- Jeya Girubha, R., & Vinodh, S. (2012). Application of fuzzy VIKOR and environmental impact analysis for material selection of an automotive component. *Materials & Design*, 37, 478–486.
- Jiang, J., Chen, Y.-W., Chen, Y.-W., & Yang, K.-W. (2011). TOPSIS with fuzzy belief structure for group belief multiple criteria decision making. *Expert Systems with Applications*, 38, 9400–9406.
- Jing, L, Chen, B., Zhang, B., & Peng, H. (2013). A hybrid fuzzy stochastic analytical hierarchy process (FSAHP) approach for evaluating ballast water treatment technologies. *Environmental Systems Research*, 2, 1–10.
- Ju, Y., Wang, A., & Liu, X. (2012). Evaluating emergency response capacity by fuzzy AHP and 2-tuple fuzzy linguistic approach. *Expert Systems with Applications*, 39, 6972–6981.
- Jun, K.-S., Chung, E.-S., Kim, Y.-G., & Kim, Y. (2013). A fuzzy multi-criteria approach to flood risk vulnerability in South Korea by considering climate change impacts. *Expert Systems with Applications*, 40, 1003–1013.
- Ka, B. (2011). Application of fuzzy AHP and ELECTRE to China dry port location selection. The Asian Journal of Shipping and Logistics, 27, 331–353.
- Kabak, M., Burmaoğlu, S., & Kazançoğlu, Y. (2012). A fuzzy hybrid MCDM approach for professional selection. *Expert Systems with Applications*, 39, 3516–3525.
- Kabak, M., Köse, E., Kırılmaz, O., & Burmaoğlu, S. (2014). A fuzzy multi-criteria decision making approach to assess building energy performance. *Energy and Buildings*, 72, 382–389.
- Kabir, G., & Sumi, R. S. (2014). Integrating fuzzy analytic hierarchy process with PROMETHEE method for total quality management consultant selection. *Production & Manufacturing Research*, *2*, 380–399.
- Kadane, J. B. (2011). Principles of uncertainty. CRC Press.
- Kahraman, C., Ates, N. Y., Çevik, S., & Gülbay, M. (2007). Fuzzy multi-attribute costbenefit analysis of e-services. *International Journal of Intelligent Systems*, 22, 547–565.
- Kahraman, C., Ates, N. Y., Çevik, S., Gülbay, M., & Erdogan, S. A. (2007). Hierarchical fuzzy TOPSIS model for selection among logistics information technologies. *Journal of Enterprise Information Management*, 20, 143–168.
- Kahraman, C., Beskese, A., & Kaya, I. (2010). Selection among ERP outsourcing alternatives using a fuzzy multi-criteria decision making methodology. *International Journal of Production Research*, 48, 547–566.
- Kahraman, C., Cebeci, U., & Ulukan, Z. (2003). Multi-criteria supplier selection using fuzzy AHP. Logistics Information Management, 16, 382–394.
- Kahraman, C., Çevik, S., Ates, N. Y., & Gülbay, M. (2007). Fuzzy multi-criteria evaluation of industrial robotic systems. *Computers & Industrial Engineering*, 52, 414–433.
- Kahraman, C., Engin, O., Kabak, Ö., & Kaya, İ. (2009). Information systems outsourcing decisions using a group decision-making approach. Engineering Applications of Artificial Intelligence, 22, 832–841.
- Kahraman, C., & Kaya, İ. (2010). A fuzzy multicriteria methodology for selection among energy alternatives. *Expert Systems with Applications*, 37, 6270–6281.
- Kahraman, C., & Kaya, T. (2011). A fuzzy approach to e-banking website quality assessment based on an integrated AHP-ELECTRE method. *Technological and Economic Development of Economy*, 313–334.
- Kahraman, C., Suder, A., & Cebi, S. (2013). Fuzzy multi-criteria and multi-experts evaluation of government investments in higher education: The case of Turkey. *Technological and Economic Development of Economy*, 19, 549–569.
- Kahraman, C., Süder, A., & Kaya, İ. (2014). Fuzzy multicriteria evaluation of health research investments. *Technological and Economic Development of Economy*, 1–17.
- Kang, H.-Y., & Lee, A. H. (2007). Priority mix planning for semiconductor fabrication by fuzzy AHP ranking. *Expert Systems with Applications*, 32, 560–570.
- Kang, H.-Y., Lee, A. H., & Yang, C.-Y. (2012). A fuzzy ANP model for supplier selection as applied to IC packaging. Journal of Intelligent Manufacturing, 23, 1477–1488.
- Kapoor, V., & Tak, S. S. (2005). Fuzzy application to the analytic hierarchy process for robot selection. Fuzzy Optimization and Decision Making, 4, 209–234.
- Karsak, E. (2002). Distance-based fuzzy MCDM approach for evaluating flexible manufacturing system alternatives. *International Journal of Production Research*, 40, 3167–3181.
- Kaya, T. (2010). Multi-attribute evaluation of website quality in E-business using an integrated fuzzy AHPTOPSIS methodology. *International Journal of Computational Intelligence Systems*, 3, 301–314.
- Kaya, T., & Kahraman, C. (2010). Multicriteria renewable energy planning using an integrated fuzzy VIKOR & AHP methodology: The case of Istanbul. *Energy*, 35, 2517–2527.
- Kaya, T., & Kahraman, C. (2011a). An integrated fuzzy AHP–ELECTRE methodology for environmental impact assessment. *Expert Systems with Applications*, 38, 8553–8562.
- Kaya, T., & Kahraman, C. (2011b). Multicriteria decision making in energy planning using a modified fuzzy TOPSIS methodology. *Expert Systems with Applications*, 38, 6577–6585.
- Kaya, İ., & Kahraman, C. (2014). A comparison of fuzzy multicriteria decision making methods for intelligent building assessment. *Journal of Civil Engineering and Management*, 20, 59–69.
- Keeney, R. L., & Raiffa, H. (1976). Decision analysis with multiple conflicting objectives. New York: Wiley & Sons.
- Kelemenis, A., & Askounis, D. (2010). A new TOPSIS-based multi-criteria approach to personnel selection. *Expert Systems with Applications*, 37, 4999–5008.

- Kelemenis, A., Ergazakis, K., & Askounis, D. (2011). Support managers' selection using an extension of fuzzy TOPSIS. *Expert Systems with Applications*, 38, 2774–2782.
- Keršulienė, V., & Turskis, Z. (2014). A hybrid linguistic fuzzy multiple criteria group selection of a chief accounting officer. *Journal of Business Economics and Management*, 15, 232–252.
- Keskin, G. A. (2014). Using integrated fuzzy DEMATEL and fuzzy C: Means algorithm for supplier evaluation and selection. *International Journal of Production Research*, 1–17.
- Kickert, W. J. (1978). Fuzzy theories on decision making: A critical review (Vol. 3). Springer.
- Kilic, H. S. (2013). An integrated approach for supplier selection in multi-item/ multi-supplier environment. Applied Mathematical Modelling, 37, 7752–7763.
- Kim, Y., & Chung, E.-S. (2013). Fuzzy VIKOR approach for assessing the vulnerability of the water supply to climate change and variability in South Korea. *Applied Mathematical Modelling*, 37, 9419–9430.
- Kim, Y., Chung, E.-S., Jun, S.-M., & Kim, S. U. (2013). Prioritizing the best sites for treated wastewater instream use in an urban watershed using fuzzy TOPSIS. *Resources, Conservation and Recycling*, 73, 23–32.
- Kreng, V. B., & Wu, C.-Y. (2007). Evaluation of knowledge portal development tools using a fuzzy AHP approach: The case of Taiwanese stone industry. *European Journal of Operational Research*, 176, 1795–1810.
- Krohling, R. A., & Campanharo, V. C. (2011). Fuzzy TOPSIS for group decision making: A case study for accidents with oil spill in the sea. *Expert Systems with Applications*, 38, 4190–4197.
- Kubler, S., Voisin, A., Derigent, W., Thomas, A., Rondeau, É., & Främling, K. (2014). Group fuzzy AHP approach to embed relevant data on "communicating material". *Computers in Industry*, 65, 675–692.
- Kucukvar, M., Gumus, S., Egilmez, G., & Tatari, O. (2014). Ranking the sustainability performance of pavements: An intuitionistic fuzzy decision making method. *Automation in Construction*, 40, 33–43.
- Kumar, R., Singh, H., & Dureja, J. (2012). An approach to analyze logistic outsourcing problem in medium-scale organization by CFPR and VIKOR. Journal of Manufacturing Technology Management, 23, 885–898.
- Kuo, M.-S. (2011a). A novel interval-valued fuzzy MCDM method for improving airlines' service quality in Chinese cross-strait airlines. *Transportation Research Part E: Logistics and Transportation Review*, 47, 1177–1193.
- Kuo, M.-S. (2011b). Optimal location selection for an international distribution center by using a new hybrid method. *Expert Systems with Applications*, 38, 7208–7221.
- Kuo, R. J., Chi, S.-C., & Kao, S.-S. (2002). A decision support system for selecting convenience store location through integration of fuzzy AHP and artificial neural network. *Computers in Industry*, 47, 199–214.
- Kuo, M.-S., & Liang, G.-S. (2011). A novel hybrid decision-making model for selecting locations in a fuzzy environment. *Mathematical and Computer Modelling*, 54, 88–104.
- Kuo, M.-S., & Liang, G.-S. (2012). A soft computing method of performance evaluation with MCDM based on interval-valued fuzzy numbers. *Applied Soft Computing*, 12, 476–485.
- Kurt, Ü. (2014). The fuzzy TOPSIS and generalized Choquet fuzzy integral algorithm for nuclear power plant site selection-a case study from Turkey. Journal of Nuclear Science and Technology, 1–15.
- Kutlu, A. C., & Ekmekçioğlu, M. (2012). Fuzzy failure modes and effects analysis by using fuzzy TOPSIS-based fuzzy AHP. *Expert Systems with Applications*, 39, 61–67.
- Kwok, R. C.-W., Zhou, D., Zhang, Q., & Ma, J. (2007). A fuzzy multi-criteria decision making model for IS student group project assessment. *Group Decision and Negotiation*, 16, 25–42.
- La Scalia, G., Aiello, G., Rastellini, C., Micale, R., & Cicalese, L. (2011). Multi-criteria decision making support system for pancreatic islet transplantation. *Expert Systems with Applications*, 38, 3091–3097.
- Labib, A. W. (2011). A supplier selection model: A comparison of fuzzy logic and the analytic hierarchy process. *International Journal of Production Research*, 49, 6287–6299.
- Lashgari, A., Fouladgar, M. M., Yazdani-Chamzini, A., & Skibniewski, M. J. (2011). Using an integrated model for shaft sinking method selection. *Journal of Civil Engineering and Management*, 17, 569–580.
- Lee, A. H. (2009). A fuzzy AHP evaluation model for buyer-supplier relationships with the consideration of benefits, opportunities, costs and risks. *International Journal of Production Research*, 47, 4255–4280.
   Lee, A. H., Chen, W.-C., & Chang, C.-J. (2008). A fuzzy AHP and BSC approach for
- Lee, A. H., Chen, W.-C., & Chang, C.-J. (2008). A fuzzy AHP and BSC approach for evaluating performance of IT department in the manufacturing industry in Taiwan. *Expert Systems with Applications*, 34, 96–107.
- Lee, Y.-C., Li, M.-L., Yen, T.-M., & Huang, T.-H. (2011). Analysis of fuzzy decision making trial and evaluation laboratory on technology acceptance model. *Expert Systems with Applications*, 38, 14407–14416.
- Lee, S. K., Mogi, G., Kim, J. W., & Gim, B. J. (2008). A fuzzy analytic hierarchy process approach for assessing national competitiveness in the hydrogen technology sector. *International Journal of Hydrogen Energy*, 33, 6840–6848.
- Lee, S. K., Mogi, G., Lee, S. K., Hui, K., & Kim, J. W. (2010). Econometric analysis of the R&D performance in the national hydrogen energy technology development for measuring relative efficiency: The fuzzy AHP/DEA integrated model approach. *International Journal of Hydrogen Energy*, 35, 2236–2246.
- Lee, S., Mogi, G., Lee, S., & Kim, J. (2011). Prioritizing the weights of hydrogen energy technologies in the sector of the hydrogen economy by using a fuzzy AHP approach. *International Journal of Hydrogen Energy*, 36, 1897–1902.

- Li, D.-F. (2005). Multiattribute decision making models and methods using intuitionistic fuzzy sets. Journal of Computer and System Sciences, 70, 73-85.
- Li, M. (2013). A multi-criteria group decision making model for knowledge management system selection based on TOPSIS with multiple distances in fuzzy environment. Kybernetes, 42, 1218-1234.
- Li, T.-S., & Huang, H.-H. (2009). Applying TRIZ and fuzzy AHP to develop innovative design for automated manufacturing systems. Expert Systems with Applications, 36. 8302-8312.
- Li, M., Jin, L., & Wang, J. (2014). A new MCDM method combining QFD with TOPSIS for knowledge management system selection from the user's perspective in intuitionistic fuzzy environment. Applied Soft Computing, 21, 28-37.
- Li, S., Shi, L., & Wang, L. (2011). The agile improvement of MMORPGs based on the enhanced chaotic neural network. Knowledge-Based Systems, 24, 642-651.
- Liao, C.-N. (2011). Fuzzy analytical hierarchy process and multi-segment goal programming applied to new product segmented under price strategy. Computers & Industrial Engineering, 61, 831–841.
- Liao, C.-N., & Kao, H.-P. (2011). An integrated fuzzy TOPSIS and MCGP approach to supplier selection in supply chain management. Expert Systems with Applications, 38, 10803-10811.
- Liao, H., & Xu, Z. (2013). A VIKOR-based method for hesitant fuzzy multi-criteria decision making. Fuzzy Optimization and Decision Making, 12, 373-392.
- Lin, R.-H. (2012). An integrated model for supplier selection under a fuzzy situation. International Journal of Production Economics, 138, 55-61.
- Lin, C. (2013). Application of fuzzy Delphi method (FDM) and fuzzy analytic hierarchy process (FAHP) to criteria weights for fashion design scheme evaluation. International Journal of Clothing Science and Technology, 25, 171-183.
- Lin, C.-T., Lee, C., & Chen, W.-Y. (2009). Using fuzzy analytic hierarchy process to evaluate service performance of a travel intermediary. The Service Industries Journal, 29, 281-296.
- Lin, C., & Twu, C. H. (2012). Fuzzy MCDM for evaluating fashion trend alternatives. International Journal of Clothing Science and Technology, 24, 141–153.
- Lin, C.-J., & Wu, W.-W. (2008). A causal analytical method for group decisionmaking under fuzzy environment. Expert Systems with Applications, 34, 205-213.
- Liou, J. J. (2013). New concepts and trends of MCDM for tomorrow-in honor of Professor Gwo-Hshiung Tzeng on the occasion of his 70th birthday. Technological and Economic Development of Economy, 19, 367-375.
- Liou, J. J., & Chuang, M.-L. (2010). Evaluating corporate image and reputation using fuzzy MCDM approach in airline market. Quality & Quantity, 44, 1079–1091.
- Liou, J. J., Chuang, Y.-C., & Tzeng, G.-H. (2014). A fuzzy integral-based model for supplier evaluation and improvement. Information Sciences, 266, 199-217.
- Liou, J. J., & Tzeng, G.-H. (2012). Comments on "multiple criteria decision making (MCDM) methods in economics: An overview". Technological and Economic Development of Economy, 18, 672-695.
- Liou, J. J., Tzeng, G.-H., Tsai, C.-Y., & Hsu, C.-C. (2011). A hybrid ANP model in fuzzy environments for strategic alliance partner selection in the airline industry. Applied Soft Computing, 11, 3515–3524.
- Liu, H. C., Qin, J. T., Mao, L. X., & Zhang, Z. Y. (2014). Personnel selection using interval 2-tuple linguistic VIKOR method. Human Factors and Ergonomics in Manufacturing & Service Industries.
- Liu, H.-C., Ren, M.-L., Wu, J., & Lin, Q.-L. (2014). An interval 2-tuple linguistic MCDM method for robot evaluation and selection. International Journal of Production Research, 52, 2867-2880.
- Liu, H.-W., & Wang, G.-J. (2007). Multi-criteria decision-making methods based on intuitionistic fuzzy sets. European Journal of Operational Research, 179, 220-233.
- Liu, H.-C., Wu, J., & Li, P. (2013). Assessment of health-care waste disposal methods using a VIKOR-based fuzzy multi-criteria decision making method. Waste Management, 33, 2744–2751.
- Lu, W.-M., & Wang, T.-C. (2011). A fuzzy multi-criteria model for the industrial cooperation program transaction strategies: A case in Taiwan. Expert Systems with Applications, 38, 1490–1500.
- Ma, D., Chang, C.-C., & Hung, S.-W. (2013). The selection of technology for latestarters: A case study of the energy-smart photovoltaic industry. Economic Modelling, 35, 10-20.
- Ma, M.-Y., Chen, C.-Y., & Wu, F.-G. (2007). A design decision-making support model for customized product color combination. Computers in Industry, 58, 504–518.
- MacCrimmon, K. R. (1968). Decision making among multiple-attribute alternatives: a survey and consolidated approach. In DTIC document.
- Mahdavi, I., Heidarzade, A., Sadeghpour-Gildeh, B., & Mahdavi-Amiri, N. (2009). A general fuzzy TOPSIS model in multiple criteria decision making. *The* International Journal of Advanced Manufacturing Technology, 45, 406-420.
- Mahdavi, I., Mahdavi-Amiri, N., Heidarzade, A., & Nourifar, R. (2008). Designing a model of fuzzy TOPSIS in multiple criteria decision making. Applied Mathematics and Computation, 206, 607-617.
- Maity, S. R., & Chakraborty, S. (2013). Grinding wheel abrasive material selection using fuzzy TOPSIS method. Materials and Manufacturing Processes, 28, 408-417.
- Majumdar, A. (2010). Selection of raw materials in textile spinning industry using fuzzy multi-criteria decision making approach. Fibers and Polymers, 11, 121-127
- Maldonado-Macías, A., Alvarado, A., García, J. L., & Balderrama, C. O. (2014). Intuitionistic fuzzy TOPSIS for ergonomic compatibility evaluation of advanced manufacturing technology. The International Journal of Advanced Manufacturing Technology, 70, 2283-2292.
- Mareschal, B., & Brans, J. P. (1992). PROMETHEE V: MCDM problems with segmentation constraints. In ULB-Universite Libre de Bruxelles.

- Mikaeil, R., Ataei, M., & Yousefi, R. (2011). Application of a fuzzy analytical hierarchy process to the prediction of vibration during rock sawing. Mining Science and Technology (China), 21, 611-619.
- Mikhailov, L., & Tsvetinov, P. (2004). Evaluation of services using a fuzzy analytic hierarchy process. Applied Soft Computing, 5, 23–33. Miyamoto, S., Liu, Z.-Q., & Kunii, T. (2000). Soft computing and human-centered
- machines. Springer-Verlag New York Inc.
- Moalagh, M., & Ravasan, A. Z. (2013). Developing a practical framework for assessing ERP post-implementation success using fuzzy analytic network process. International Journal of Production Research, 51, 1236-1257.
- Moghimi, R., & Anvari, A. (2014). An integrated fuzzy MCDM approach and analysis to evaluate the financial performance of Iranian cement companies. International Journal of Advanced Manufacturing Technology, 71, 685-698.
- Mokhtari, S. M., Alinejad-Rokny, H., & Jalalifar, H. (2014). Selection of the best well control system by using fuzzy multiple-attribute decision-making methods. Journal of Applied Statistics, 41, 1105–1121.
- Mokhtarian, M. N. (2014). A note on Extension of fuzzy TOPSIS method based on interval-valued fuzzy sets. Applied Soft Computing.
- Mokhtarian, M., Sadi-nezhad, S., & Makui, A. (2014a). A new flexible and reliable IVF-TOPSIS method based on uncertainty risk reduction in decision making process. Applied Soft Computing.
- Mokhtarian, M. N., Sadi-nezhad, S., & Makui, A. (2014b). A new flexible and reliable interval valued fuzzy VIKOR method based on uncertainty risk reduction in decision making process: An application for determining a suitable location for digging some pits for municipal wet waste landfill. Computers & Industrial Engineering, 78, 213-233.
- Montazer, G. A., Saremi, H. Q., & Ramezani, M. (2009). Design a new mixed expert decision aiding system using fuzzy ELECTRE III method for vendor selection. Expert Systems with Applications, 36, 10837-10847.
- Munda, G. (1998). Multicriteria evaluation in a fuzzy environment. Theory and Applications in Ecological Economics.
- Naghadehi, M. Z., Mikaeil, R., & Ataei, M. (2009). The application of fuzzy analytic hierarchy process (FAHP) approach to selection of optimum underground mining method for Jajarm Bauxite Mine, Iran. Expert Systems with Applications, 36 8218-8226
- Najafi, A., Karimpour, M. H., & Ghaderi, M. (2014). Application of fuzzy AHP method to IOCG prospectivity mapping: A case study in Taherabad prospecting area, eastern Iran. International Journal of Applied Earth Observation and Geoinformation, 33, 142-154.
- Ölçer, A., & Odabaşi, A. (2005a). A new fuzzy multiple attributive group decision making methodology and its application to propulsion/manoeuvring system selection problem. European Journal of Operational Research, 166, 93-114.
- Ölçer, A. İ., & Odabaşi, A. Y. (2005b). A new fuzzy multiple attributive group decision making methodology and its application to propulsion/manoeuvring system selection problem. European Journal of Operational Research, 166, 93-114.
- Önüt, S., Efendigil, T., & Soner Kara, S. (2010). A combined fuzzy MCDM approach for selecting shopping center site: An example from Istanbul, Turkey. Expert Systems with Applications, 37, 1973–1980.
- Önüt, S., Kara, S. S., & Efendigil, T. (2008). A hybrid fuzzy MCDM approach to machine tool selection. Journal of Intelligent Manufacturing, 19, 443-453.
- Önüt, S., Kara, S. S., & Isik, E. (2009). Long term supplier selection using a combined fuzzy MCDM approach: A case study for a telecommunication company. Expert Systems with Applications, 36, 3887–3895.
- Önüt, S., & Soner, S. (2008). Transshipment site selection using the AHP and TOPSIS approaches under fuzzy environment. Waste Management, 28, 1552–1559.
- Opricovic, S. (2011). Fuzzy VIKOR with an application to water resources planning. Expert Systems with Applications, 38, 12983-12990.
- Opricovic, S., & Tzeng, G.-H. (2004). Compromise solution by MCDM methods: A comparative analysis of VIKOR and TOPSIS. European Journal of Operational Research, 156, 445-455.
- Özgen, A., & Tanyas, M. (2011). Joint selection of customs broker agencies and international road transportation firms by a fuzzy analytic network process approach. Expert Systems with Applications, 38, 8251-8258.
- Padma, T., & Balasubramanie, P. (2011). A fuzzy analytic hierarchy processing decision support system to analyze occupational menace forecasting the spawning of shoulder and neck pain. Expert Systems with Applications, 38, 15303-15309.
- Paksoy, T., Pehlivan, N. Y., & Kahraman, C. (2012). Organizational strategy development in distribution channel management using fuzzy AHP and hierarchical fuzzy TOPSIS. *Expert Systems with Applications*, 39, 2822–2841.
- Palanisamy, P., & Zubar, H. A. (2013). Hybrid MCDM approach for vendor ranking. Journal of Manufacturing Technology Management, 24, 905–928.
- Pan, N.-F. (2008). Fuzzy AHP approach for selecting the suitable bridge construction method. Automation in Construction, 17, 958-965.
- Pang, B., & Bai, S. (2013). An integrated fuzzy synthetic evaluation approach for supplier selection based on analytic network process. Journal of Intelligent Manufacturing, 24, 163-174.
- Park, J. H., Cho, H. J., & Kwun, Y. C. (2011). Extension of the VIKOR method for group decision making with interval-valued intuitionistic fuzzy information. Fuzzy Optimization and Decision Making, 10, 233–253.
- Pawlak, Z. (1982). Rough sets. International Journal of Computer & Information Sciences, 11, 341-356.
- Peneva, V., & Popchev, I. (2008). Multicriteria decision making based on fuzzy relations. Cybernetics and Information Technologies, 8.

- Perçin, S. (2008). Use of fuzzy AHP for evaluating the benefits of informationsharing decisions in a supply chain. *Journal of Enterprise Information Management*, 21, 263–284.
- Pires, A., Chang, N.-B., & Martinho, G. (2011). An AHP-based fuzzy interval TOPSIS assessment for sustainable expansion of the solid waste management system in Setúbal Peninsula, Portugal. *Resources, Conservation and Recycling*, 56, 7–21.
- Rabbani, A., Zamani, M., Yazdani-Chamzini, A., & Zavadskas, E. K. (2014). Proposing a new integrated model based on sustainability balanced scorecard (SBSC) and MCDM approaches by using linguistic variables for the performance evaluation of oil producing companies. *Expert Systems with Applications*, 41, 7316–7327.
- Rajput, H. C., Milani, A. S., & Labun, A. (2011). Including time dependency and ANOVA in decision-making using the revised fuzzy AHP: A case study on wafer fabrication process selection. *Applied Soft Computing*, *11*, 5099–5109.
- Rathod, M. K., & Kanzaria, H. V. (2011). A methodological concept for phase change material selection based on multiple criteria decision analysis with and without fuzzy environment. *Materials & Design*, 32, 3578–3585.
- Rezaei, J., & Dowlatshahi, S. (2010). A rule-based multi-criteria approach to inventory classification. *International Journal of Production Research*, 48, 7107–7126.
- Rezaei, J., Ortt, R., & Scholten, V. (2013). An improved fuzzy preference programming to evaluate entrepreneurship orientation. *Applied Soft Computing*, 13, 2749–2758.
- Rezaian, S., & Jozi, S. A. (2012). Health-safety and environmental risk assessment of refineries using of multi criteria decision making method. APCBEE Procedia, 3, 235–238.
- Ribeiro, R. A. (1996). Fuzzy multiple attribute decision making: A review and new preference elicitation techniques. *Fuzzy Sets and Systems*, 78, 155–181.
- Ribeiro, R. A., Falcão, A., Mora, A., & Fonseca, J. M. (2014). FIF: A fuzzy information fusion algorithm based on multi-criteria decision making. *Knowledge-Based Systems*, 58, 23–32.
- Rikhegar, N., Mansouri, N., Ahadi Oroumieh, A., Yazdani-Chamzini, A., Zavadskas, E. K., & Kildienė, S. (2014). Environmental impact assessment based on group decision-making methods in mining projects. *Economic Research-Ekonomska Istraživanja*, 27, 378–392.
- Roghanian, E., Rahimi, J., & Ansari, A. (2010). Comparison of first aggregation and last aggregation in fuzzy group TOPSIS. *Applied Mathematical Modelling*, 34, 3754–3766.
- Roshandel, J., Miri-Nargesi, S. S., & Hatami-Shirkouhi, L. (2013). Evaluating and selecting the supplier in detergent production industry using hierarchical fuzzy TOPSIS. Applied Mathematical Modelling, 37, 10170–10181.
- Rostamzadeh, R., & Sofian, S. (2011). Prioritizing effective 7Ms to improve production systems performance using fuzzy AHP and fuzzy TOPSIS (case study). Expert Systems with Applications, 38, 5166–5177.
- Rouhani, S., Ghazanfari, M., & Jafari, M. (2012). Evaluation model of business intelligence for enterprise systems using fuzzy TOPSIS. Expert Systems with Applications, 39, 3764–3771.
- Rouyendegh, B. D., & Erkan, T. E. (2013). An application of the fuzzy electre method for academic staff selection. *Human Factors and Ergonomics in Manufacturing & Service Industries*, 23, 107–115.
- Roy, B. (1996). Multicriteria methodology for decision aiding (Vol. 12). Springer.
- Roy, M. K., Ray, A., & Pradhan, B. B. (2014). Non-traditional machining process selection using integrated fuzzy AHP and QFD techniques: A customer perspective. Production & Manufacturing Research, 2, 530–549.
- Saaty, T. L. (1988). What is the analytic hierarchy process? Springer.
- Saaty, R. W. (2003). Decision making in complex environment: The analytic hierarchy process (AHP) for decision making and the analytic network process (ANP) for decision making with dependence and feedback. Pittsburgh: Super Decisions.
- Saaty, T. L. (2005). Theory and applications of the analytic network process: Decision making with benefits, opportunities, costs and risks. RWS publications.
- Saaty, T. L., & Vargas, L. G. (2006). Decision making with the analytic network process. Springer.
- Sadeghzadeh, K., & Salehi, M. B. (2011). Mathematical analysis of fuel cell strategic technologies development solutions in the automotive industry by the TOPSIS multi-criteria decision making method. *International Journal of Hydrogen Energy*, 36, 13272–13280.
- Sadi-Nezhad, S., & Khalili Damghani, K. (2010). Application of a fuzzy TOPSIS method base on modified preference ratio and fuzzy distance measurement in assessment of traffic police centers performance. *Applied Soft Computing*, 10, 1028–1039.
- Safaei Ghadikolaei, A., Khalili Esbouei, S., & Antucheviciene, J. (2014). Applying fuzzy MCDM for financial performance evaluation of Iranian companies. *Technological and Economic Development of Economy*, 20, 274–291.
- Saidi Mehrabad, M., & Anvari, M. (2010). Provident decision making by considering dynamic and fuzzy environment for FMS evaluation. *International Journal of Production Research*, 48, 4555–4584.
- Sakthivel, G., Ilangkumaran, M., Nagarajan, G., Priyadharshini, G. V., Kumar, S. D., Kumar, S. S., et al. (2013). Multi-criteria decision modelling approach for biodiesel blend selection based on GRA–TOPSIS analysis. *International Journal of Ambient Energy*, 1–16.
- Samvedi, A., Jain, V., & Chan, F. T. (2012). An integrated approach for machine tool selection using fuzzy analytical hierarchy process and grey relational analysis. *International Journal of Production Research*, *50*, 3211–3221.
- Samvedi, A., Jain, V., & Chan, F. T. (2013). Quantifying risks in a supply chain through integration of fuzzy AHP and fuzzy TOPSIS. International Journal of Production Research, 51, 2433–2442.

- Saremi, M., Mousavi, S. F., & Sanayei, A. (2009). TQM consultant selection in SMEs with TOPSIS under fuzzy environment. *Expert Systems with Applications*, 36, 2742–2749.
- Sari, K. (2013). Selection of RFID solution provider: A fuzzy multi-criteria decision model with Monte Carlo simulation. *Kybernetes*, 42, 448–465.
- Sasikumar, P., & Haq, A. N. (2011). Integration of closed loop distribution supply chain network and 3PRLP selection for the case of battery recycling. *International Journal of Production Research*, 49, 3363–3385.
- Şen, C. G., & Çinar, G. (2010). Evaluation and pre-allocation of operators with multiple skills: A combined fuzzy AHP and max-min approach. *Expert Systems* with Applications, 37, 2043–2053.
- Şen, C. G., Şen, S., & Başlıgil, H. (2010). Pre-selection of suppliers through an integrated fuzzy analytic hierarchy process and max-min methodology. *International Journal of Production Research*, 48, 1603–1625.
- Sepehriar, A., Eslamipoor, R., & Nobari, A. (2013). A new mixed fuzzy-LP method for selecting the best supplier using fuzzy group decision making. *Neural Computing* and Applications, 23, 345–352.
- Sevkli, M. (2010). An application of the fuzzy ELECTRE method for supplier selection. *International Journal of Production Research*, 48, 3393–3405.
- Sevkli, M., Oztekin, A., Uysal, O., Torlak, G., Turkyilmaz, A., & Delen, D. (2012). Development of a fuzzy ANP based SWOT analysis for the airline industry in Turkey. *Expert Systems with Applications*, 39, 14–24.
- Shafia, M. A., Mazdeh, M. M., Vahedi, M., & Pournader, M. (2011). Applying fuzzy balanced scorecard for evaluating the CRM performance. *Industrial Management* & Data Systems, 111, 1105–1135.
- Shakouri, G. H., & Tavassoli, N. Y. (2012). Implementation of a hybrid fuzzy system as a decision support process: A FAHP-FMCDM-FIS composition. *Expert Systems* with Applications, 39, 3682–3691.
- Shemshadi, A., Shirazi, H., Toreihi, M., & Tarokh, M. J. (2011). A fuzzy VIKOR method for supplier selection based on entropy measure for objective weighting. *Expert Systems with Applications*, 38, 12160–12167.
- Shen, Y.-C., Lin, G. T., Li, K.-P., & Yuan, B. J. (2010). An assessment of exploiting renewable energy sources with concerns of policy and technology. *Energy Policy*, 38, 4604–4616.
- Shen, Y.-C., Lin, G. T., & Tzeng, G.-H. (2011). Combined DEMATEL techniques with novel MCDM for the organic light emitting diode technology selection. *Expert Systems with Applications*, 38, 1468–1481.
- Shen, L., Olfat, L., Govindan, K., Khodaverdi, R., & Diabat, A. (2013). A fuzzy multi criteria approach for evaluating green supplier's performance in green supply chain with linguistic preferences. *Resources, Conservation and Recycling*, 74, 170–179.
- Sheu, J.-B. (2008). A hybrid neuro-fuzzy analytical approach to mode choice of global logistics management. *European Journal of Operational Research*, 189, 971–986.
- Shyur, H.-J., & Shih, H.-S. (2006). A hybrid MCDM model for strategic vendor selection. Mathematical and Computer Modelling, 44, 749–761.
- Simões-Marques, M., Ribeiro, R. A., & Gameiro-Marques, A. (2000). A fuzzy decision support system for equipment repair under battle conditions. *Fuzzy Sets and Systems*, 115, 141–157.
- Singh, R. K., & Benyoucef, L. (2011). A fuzzy TOPSIS based approach for e-sourcing. Engineering Applications of Artificial Intelligence, 24, 437–448.
- Singh, R., & Benyoucef, L. (2013). A consensus based group decision making methodology for strategic selection problems of supply chain coordination. *Engineering Applications of Artificial Intelligence*, 26, 122–134.
- Soner Kara, S. (2011). Supplier selection with an integrated methodology in unknown environment. *Expert Systems with Applications*, *38*, 2133–2139.
- Streimikiene, D., Balezentis, T., Krisciukaitiene, I., & Balezentis, A. (2012). Prioritizing sustainable electricity production technologies: MCDM approach. Renewable and Sustainable Energy Reviews, 16, 3302–3311.
- Sun, C.-C. (2010). A performance evaluation model by integrating fuzzy AHP and fuzzy TOPSIS methods. *Expert Systems with Applications*, 37, 7745–7754.
- Sun, C.-C., & Lin, G. T. (2009). Using fuzzy TOPSIS method for evaluating the competitive advantages of shopping websites. *Expert Systems with Applications*, 36, 11764–11771.
- Sun, C.-C., Lin, G. T., & Tzeng, G.-H. (2009). The evaluation of cluster policy by fuzzy MCDM: Empirical evidence from HsinChu Science Park. *Expert Systems with Applications*, 36, 11895–11906.
- Tadić, S., Zečević, S., & Krstić, M. (2014). A novel hybrid MCDM model based on fuzzy DEMATEL, fuzzy ANP and fuzzy VIKOR for city logistics concept selection. *Expert Systems with Applications*, *41*, 8112–8128.
- Taha, Z., & Rostam, S. (2012). A hybrid fuzzy AHP-PROMETHEE decision support system for machine tool selection in flexible manufacturing cell. *Journal of Intelligent Manufacturing*, 23, 2137–2149.
- Tan, C. (2011). A multi-criteria interval-valued intuitionistic fuzzy group decision making with Choquet integral-based TOPSIS. Expert Systems with Applications, 38, 3023–3033.
- Tan, R. R., Aviso, K. B., Huelgas, A. P., & Promentilla, M. A. B. (2013). Fuzzy AHP approach to selection problems in process engineering involving quantitative and qualitative aspects. *Process Safety and Environmental Protection*.
- Tang, Y.-C. (2009). An approach to budget allocation for an aerospace company fuzzy analytic hierarchy process and artificial neural network. *Neurocomputing*, 72, 3477–3489.
- Tansel İç, Y. (2012). Development of a credit limit allocation model for banks using an integrated fuzzy TOPSIS and linear programming. *Expert Systems with Applications*, 39, 5309–5316.

- Tansel İç, Y., Yurdakul, M., & Dengiz, B. (2013). Development of a decision support system for robot selection. *Robotics and Computer-Integrated Manufacturing*, 29, 142–157.
- Tavana, M., Khalili-Damghani, K., & Abtahi, A.-R. (2013). A hybrid fuzzy group decision support framework for advanced-technology prioritization at NASA. *Expert Systems with Applications*, 40, 480–491.
- Tavana, M., Khalili-Damghani, K., & Rahmatian, R. (2014). A hybrid fuzzy MCDM method for measuring the performance of publicly held pharmaceutical companies. *Annals of Operations Research*, 1–33.
- Tavana, M., Momeni, E., Rezaeiniya, N., Mirhedayatian, S. M., & Rezaeiniya, H. (2013). A novel hybrid social media platform selection model using fuzzy ANP and COPRAS-G. *Expert Systems with Applications*, 40, 5694–5702.
- Tavana, M., Zandi, F., & Katehakis, M. N. (2013). A hybrid fuzzy group ANP-TOPSIS framework for assessment of e-government readiness from a CiRM perspective. *Information & Management*, 50, 383–397.
- Tesfamariam, S., & Sadiq, R. (2006). Risk-based environmental decision-making using fuzzy analytic hierarchy process (F-AHP). Stochastic Environmental Research and Risk Assessment, 21, 35–50.
- Torfi, F., Farahani, R. Z., & Rezapour, S. (2010). Fuzzy AHP to determine the relative weights of evaluation criteria and Fuzzy TOPSIS to rank the alternatives. *Applied Soft Computing*, *10*, 520–528.
- Torlak, G., Sevkli, M., Sanal, M., & Zaim, S. (2011). Analyzing business competition by using fuzzy TOPSIS method: An example of Turkish domestic airline industry. *Expert Systems with Applications*, 38, 3396–3406.
- Torra, V. (2010). Hesitant fuzzy sets. International Journal of Intelligent Systems, 25, 529–539.
- Torra, V., & Narukawa, Y. (2009). On hesitant fuzzy sets and decision. In IEEE International Conference on Fuzzy Systems, 2009. FUZZ-IEEE 2009 (pp. 1378–1382). IEEE.
- Tsai, H.-Y., Chang, C.-W., & Lin, H.-L. (2010). Fuzzy hierarchy sensitive with Delphi method to evaluate hospital organization performance. *Expert Systems with Applications*, 37, 5533–5541.
- Tsai, M.-C., & Lin, C.-T. (2012). Selecting an optimal region by fuzzy group decision making: Empirical evidence from medical investors. Group Decision and Negotiation, 21, 399–416.
- Tsaur, S.-H., Chang, T.-Y., & Yen, C.-H. (2002). The evaluation of airline service quality by fuzzy MCDM. *Tourism Management*, 23, 107–115.
- Tseng, M.-L. (2009a). A causal and effect decision making model of service quality expectation using grey-fuzzy DEMATEL approach. Expert Systems with Applications, 36, 7738–7748.
- Tseng, M.-L. (2009b). Using the extension of DEMATEL to integrate hotel service quality perceptions into a cause–effect model in uncertainty. *Expert Systems with Applications*, 36, 9015–9023.
- Tseng, M.-L. (2010). An assessment of cause and effect decision-making model for firm environmental knowledge management capacities in uncertainty. *Environmental Monitoring and Assessment*, 161, 549–564.
- Tseng, M.-L., Chiang, J. H., & Lan, L. W. (2009). Selection of optimal supplier in supply chain management strategy with analytic network process and choquet integral. *Computers & Industrial Engineering*, 57, 330–340.
- Tseng, F.-M., Chiu, Y.-J., & Chen, J.-S. (2009). Measuring business performance in the high-tech manufacturing industry: A case study of Taiwan's large-sized TFT-LCD panel companies. *Omega*, 37, 686–697.
   Tseng, M.-L., Lin, R.-J., & Chen, H.-P. (2011). Evaluating the effectiveness of e-
- Tseng, M.-L., Lin, R.-J., & Chen, H.-P. (2011). Evaluating the effectiveness of elearning system in uncertainty. *Industrial Management & Data Systems*, 111, 869–889.
- Tseng, M.-L., Lin, Y.-H., & Chiu, A. S. F. (2009). Fuzzy AHP-based study of cleaner production implementation in Taiwan PWB manufacturer. *Journal of Cleaner Production*, 17, 1249–1256.
- Turskis, Z., & Zavadskas, E. K. (2010a). A new additive ratio assessment (ARAS) method in multicriteria decision-making. *Technological and Economic Development of Economy*, 159–172.
- Turskis, Z., & Zavadskas, E. K. (2010b). A new fuzzy additive ratio assessment method (ARAS-F). Case study: The analysis of fuzzy multiple criteria in order to select the logistic centers location. *Transport*, 25, 423–432.
- Tuzkaya, G., Gülsün, B., Kahraman, C., & Özgen, D. (2010). An integrated fuzzy multicriteria decision making methodology for material handling equipment selection problem and an application. *Expert Systems with Applications*, 37, 2853–2863.
- Tuzkaya, G., Gülsün, B., & Önsel, Ş. (2011). A methodology for the strategic design of reverse logistics networks and its application in the Turkish white goods industry. *International Journal of Production Research*, 49, 4543–4571.
- Tuzkaya, U. R., & Önüt, S. (2008). A fuzzy analytic network process based approach to transportation-mode selection between Turkey and Germany: A case study. *Information Sciences*, 178, 3133–3146.
- Tzeng, G.-H., & Huang, J.-J. (2011). Multiple attribute decision making: Methods and applications. CRC Press.
- Ustundag, A., & Serdar Kilinc, M. (2011). Fuzzy multi-criteria selection of science parks for start-up companies. *International Journal of Computational Intelligence Systems*, 4, 217–227.
- Uygun, Ö., Kaçamak, H., & Kahraman, Ü. A. (2014). An integrated DEMATEL and fuzzy ANP techniques for evaluation and selection of outsourcing provider for a telecommunication company. *Computers & Industrial Engineering*.
- Uysal, F., & Tosun, Ö. (2012). Fuzzy TOPSIS-based computerized maintenance management system selection. *Journal of Manufacturing Technology Management*, 23, 212–228.

- Vadrevu, K. P., Eaturu, A., & Badarinath, K. (2010). Fire risk evaluation using multicriteria analysis—a case study. *Environmental Monitoring and Assessment*, 166, 223–239.
- Vafai, F., Hadipour, V., & Hadipour, A. (2013). Determination of shoreline sensitivity to oil spills by use of GIS and fuzzy model. Case study-the coastal areas of Caspian Sea in north of Iran. Ocean & Coastal Management, 71, 123–130.
- Vahdani, B., & Hadipour, H. (2011). Extension of the ELECTRE method based on interval-valued fuzzy sets. Soft Computing, 15, 569–579.
- Vahdani, B., Hadipour, H., Sadaghiani, J. S., & Amiri, M. (2010). Extension of VIKOR method based on interval-valued fuzzy sets. *The International Journal of Advanced Manufacturing Technology*, 47, 1231–1239.
- Vahdani, B., Hadipour, H., & Tavakkoli-Moghaddam, R. (2012). Soft computing based on interval valued fuzzy ANP-A novel methodology. *Journal of Intelligent Manufacturing*, 23, 1529–1544.
- Vahdani, B., Mousavi, S. M., & Tavakkoli-Moghaddam, R. (2011). Group decision making based on novel fuzzy modified TOPSIS method. Applied Mathematical Modelling, 35, 4257–4269.
- Vahdani, B., Mousavi, S. M., Tavakkoli-Moghaddam, R., & Hashemi, H. (2013). A new design of the elimination and choice translating reality method for multicriteria group decision-making in an intuitionistic fuzzy environment. *Applied Mathematical Modelling*, 37, 1781–1799.
- Vahdat, K., Smith, N. J., & Amiri, G. G. (2014). Fuzzy multicriteria for developing a risk management system in seismically prone areas. *Socio-Economic Planning Sciences*, 48, 235–248.
- Vaidya, O. S., & Kumar, S. (2006). Analytic hierarchy process: An overview of applications. European Journal of Operational Research, 169, 1–29.
- van de Kaa, G., Rezaei, J., Kamp, L., & de Winter, A. (2014). Photovoltaic technology selection: A fuzzy MCDM approach. *Renewable and Sustainable Energy Reviews*, 32, 662–670.
- Vinodh, S., Anesh Ramiya, R., & Gautham, S. (2011). Application of fuzzy analytic network process for supplier selection in a manufacturing organisation. *Expert Systems with Applications*, 38, 272–280.
- Vinodh, S., Gautham, S., Anesh Ramiya, R., & Rajanayagam, D. (2010). Application of fuzzy analytic network process for agile concept selection in a manufacturing organisation. *International Journal of Production Research*, 48, 7243–7264.
- Vinodh, S., Mulanjur, G., & Thiagarajan, A. (2013). Sustainable concept selection using modified fuzzy TOPSIS: A case study. *International Journal of Sustainable Engineering*, 6, 109–116.
- Vinodh, S., Prasanna, M., & Hari Prakash, N. (2014). Integrated fuzzy AHP-TOPSIS for selecting the best plastic recycling method: A case study. *Applied Mathematical Modelling*, 38, 4662–4672.
- Vinodh, S., Varadharajan, A. R., & Subramanian, A. (2013). Application of fuzzy VIKOR for concept selection in an agile environment. *The International Journal of Advanced Manufacturing Technology*, 65, 825–832.
- Wang, Y.-J. (2008). Applying FMCDM to evaluate financial performance of domestic airlines in Taiwan. Expert Systems with Applications, 34, 1837–1845.
- Wang, Y.-J. (2014). A fuzzy multi-criteria decision-making model by associating technique for order preference by similarity to ideal solution with relative preference relation. *Information Sciences*, 268, 169–184.
- Wang, X., & Chan, H. K. (2013a). A hierarchical fuzzy TOPSIS approach to assess improvement areas when implementing green supply chain initiatives. International Journal of Production Research, 51, 3117–3130.
- Wang, X., & Chan, H. K. (2013b). An integrated fuzzy approach for evaluating remanufacturing alternatives of a product design. *Journal of Remanufacturing*, 3, 1–19.
- Wang, T.-C., & Chang, T.-H. (2007). Application of TOPSIS in evaluating initial training aircraft under a fuzzy environment. *Expert Systems with Applications*, 33, 870–880.
- Wang, T.-C., & Chen, Y.-H. (2008). Applying fuzzy linguistic preference relations to the improvement of consistency of fuzzy AHP. *Information Sciences*, 178, 3755–3765.
- Wang, L., Chu, J., & Wu, J. (2007). Selection of optimum maintenance strategies based on a fuzzy analytic hierarchy process. *International Journal of Production Economics*, 107, 151–163.
- Wang, Y.-M., & Elhag, T. (2006). Fuzzy TOPSIS method based on alpha level sets with an application to bridge risk assessment. *Expert Systems with Applications*, 31, 309–319.
- Wang, J., Fan, K., & Wang, W. (2010). Integration of fuzzy AHP and FPP with TOPSIS methodology for aeroengine health assessment. *Expert Systems with Applications*, 37, 8516–8526.
- Wang, Y.-J., & Lee, H.-S. (2007). Generalizing TOPSIS for fuzzy multiple-criteria group decision-making. *Computers & Mathematics with Applications*, 53, 1762–1772.
- Wang, T.-C., & Lee, H.-D. (2009). Developing a fuzzy TOPSIS approach based on subjective weights and objective weights. *Expert Systems with Applications*, 36, 8980–8985.
- Wang, Y., Lee, H., & Lin, K. (2003). Fuzzy TOPSIS for multi-criteria decision-making. International Mathematical Journal, 3, 367–379.
- Wang, Y.-M., Luo, Y., & Hua, Z. (2008). On the extent analysis method for fuzzy AHP and its applications. *European Journal of Operational Research*, 186, 735–747.
- Wang, C.-H., & Wu, H.-S. (2014). A novel framework to evaluate programmable logic controllers: A fuzzy MCDM perspective. *Journal of Intelligent Manufacturing*, 1–10.
- Wiecek, M. M., Ehrgott, M., Fadel, G., & Rui Figueira, J. (2008). Multiple criteria decision making for engineering. Omega, 36, 337–339.

- Wu, W.-W. (2012). Segmenting critical factors for successful knowledge management implementation using the fuzzy DEMATEL method. *Applied Soft Computing*, 12, 527–535.
- Wu, M.-C., & Chen, T.-Y. (2011). The ELECTRE multicriteria analysis approach based on Atanassov's intuitionistic fuzzy sets. *Expert Systems with Applications*, 38, 12318–12327.
- Wu, W.-W., & Lee, Y.-T. (2007). Developing global managers' competencies using the fuzzy DEMATEL method. *Expert Systems with Applications*, 32, 499–507.
- Wu, H.-Y., Tzeng, G.-H., & Chen, Y.-H. (2009). A fuzzy MCDM approach for evaluating banking performance based on balanced scorecard. *Expert Systems* with Applications, 36, 10135–10147.
- Xu, Z. (2004a). EOWA and EOWG operators for aggregating linguistic labels based on linguistic preference relations. *International Journal of Uncertainty, Fuzziness* and Knowledge-based Systems, 12, 791–810.
- Xu, Z. (2004b). A method based on linguistic aggregation operators for group decision making with linguistic preference relations. *Information Sciences*, 166, 19–30.
- Xu, Z., & Da, Q. (2002). The ordered weighted geometric averaging operators. International Journal of Intelligent Systems, 17, 709–716.
- Yager, R. R. (1978). Fuzzy decision making including unequal objectives. Fuzzy Sets and Systems, 1, 87–95.
- Yager, R. R. (1986). A characterization of the extension principle. Fuzzy Sets and Systems, 18, 205–217.
- Yager, R. R. (1994a). Aggregation operators and fuzzy systems modeling. Fuzzy Sets and Systems, 67, 129–145.
- Yager, R. R. (1994b). On weighted median aggregation. International Journal of Uncertainty, Fuzziness and Knowledge-based Systems, 2, 101–113.
- Yager, R. R. (2004). OWA aggregation over a continuous interval argument with applications to decision making. *IEEE Transactions on Systems, Man, and Cybernetics, Part B: Cybernetics*, 34, 1952–1963.
- Yalcin, N., Bayrakdaroglu, A., & Kahraman, C. (2012). Application of fuzzy multicriteria decision making methods for financial performance evaluation of Turkish manufacturing industries. *Expert Systems with Applications*, 39, 350–364.
- Yang, Z., Bonsall, S., & Wang, J. (2011). Approximate TOPSIS for vessel selection under uncertain environment. *Expert Systems with Applications*, 38, 14523–14534.
- Yang, H.-W., & Chang, K.-F. (2012). Combining means-end chain and fuzzy ANP to explore customers' decision process in selecting bundles. *International Journal of Information Management*, 32, 381–395.
- Yazdani-Chamzini, A. (2014). An integrated fuzzy multi criteria group decision making model for handling equipment selection. *Journal of Civil Engineering and Management*, 20, 660–673.
- Yazdani-Chamzini, A., Shariati, S., Haji Yakhchali, S., & Zavadskas, E. K. (2014). Proposing a new methodology for prioritising the investment strategies in the private sector of Iran. Economic Research-Ekonomska Istraživanja, 27, 320–345.
- Yeh, T.-M., Pai, F.-Y., & Liao, C.-W. (2014). Using a hybrid MCDM methodology to identify critical factors in new product development. *Neural Computing and Applications*, 24, 957–971.
- Yilmaz, B., & Dağdeviren, M. (2011). A combined approach for equipment selection: F-PROMETHEE method and zero-one goal programming. *Expert Systems with Applications*, 38, 11641–11650.
- Yong, D. (2006). Plant location selection based on fuzzy TOPSIS. The International Journal of Advanced Manufacturing Technology, 28, 839–844.
- Yong, D., & Qi, L. (2005). A TOPSIS-based centroid-index ranking method of fuzzy numbers and its application in decision-making. *Cybernetics and Systems: An International Journal*, 36, 581–595.
- Yu, X., Guo, S., Guo, J., & Huang, X. (2011). Rank B2C e-commerce websites in ealliance based on AHP and fuzzy TOPSIS. *Expert Systems with Applications*, 38, 3550–3557.
- Yu, V. F., & Hu, K.-J. (2010). An integrated fuzzy multi-criteria approach for the performance evaluation of multiple manufacturing plants. *Computers & Industrial Engineering*, 58, 269–277.
   Yu, D., Wu, Y., & Zhou, W. (2012). Generalized hesitant fuzzy Bonferroni mean and
- Yu, D., Wu, Y., & Zhou, W. (2012). Generalized hesitant fuzzy Bonferroni mean and its application in multi-criteria group decision making. *Journal of Information and Computational Science*, *9*, 267–274.
   Yücenur, G. N., & Demirel, N. Ç. (2012). Group decision making process for insurance
- Yücenur, G. N., & Demirel, N. Ç. (2012). Group decision making process for insurance company selection problem with extended VIKOR method under fuzzy environment. *Expert Systems with Applications*, 39, 3702–3707.

- Yurdakul, M., & İç, Y. T. (2009). Analysis of the benefit generated by using fuzzy numbers in a TOPSIS model developed for machine tool selection problems. *Journal of Materials Processing Technology*, 209, 310–317.
- Zadeh, L. A. (1965). Fuzzy sets. Information and Control, 8, 338-353.
- Zadeh, L. A. (1975). The concept of a linguistic variable and its application to approximate reasoning–I. *Information Sciences*, *8*, 199–249.
- Zaerpour, N., Rabbani, M., Gharehgozli, A. H., & Tavakkoli-Moghaddam, R. (2009). A comprehensive decision making structure for partitioning of make-to-order, make-to-stock and hybrid products. *Soft Computing*, 13, 1035–1054.
- Zagorskas, J., Zavadskas, E. K., Turskis, Z., Burinskiene, M., Blumberga, A., & Blumberga, D. (2014). Thermal insulation alternatives of historic brick buildings in Baltic Sea Region. *Energy and Buildings*, 78, 35–42.
- Zamani, M., Rabbani, A., Yazdani-Chamzini, A., & Turskis, Z. (2014). An integrated model for extending brand based on fuzzy ARAS and ANP methods. *Journal of Business Economics and Management*, 15, 403–423.
- Zangoueinezhad, A., Azar, A., & Kazazi, A. (2011). Using SCOR model with fuzzy MCDM approach to assess competitiveness positioning of supply chains: Focus on shipbuilding supply chains. *Maritime Policy & Management*, 38, 93–109.
- Zangoueinezhad, A., & Moshabaki, A. (2011). Measuring university performance using a knowledge-based balanced scorecard. International Journal of Productivity and Performance Management, 60, 824–843.
- Zare Mehrjerdi, Y. (2014). Strategic system selection with linguistic preferences and grey information using MCDM. Applied Soft Computing, 18, 323–337.
- Zavadskas, E. K., Antucheviciene, J., Hajiagha, S. H. R., & Hashemi, S. S. (2014). Extension of weighted aggregated sum product assessment with intervalvalued intuitionistic fuzzy numbers (WASPAS-IVIF). Applied Soft Computing, 24, 1013–1021.
- Zavadskas, E., Kaklauskas, A., & Sarka, V. (1994). The new method of multicriteria complex proportional assessment of projects. *Technological and Economic Development of Economy*, 1, 131–139.
- Zavadskas, E. K., Skibniewski, M. J., & Antucheviciene, J. (2014). Performance analysis of Civil Engineering Journals based on the Web of Science<sup>®</sup> database. *Archives of Civil and Mechanical Engineering*, 14, 519–527.
- Zavadskas, E. K., & Turskis, Z. (2008). A new logarithmic normalization method in games theory. *Informatica*, 19, 303–314.
- Zavadskas, E. K., & Turskis, Z. (2011). Multiple criteria decision making (MCDM) methods in economics: An overview. *Technological and Economic Development of Economy*, 397–427.
- Zavadskas, E., Turskis, Z., Antucheviciene, J., & Zakarevicius, A. (2012). Optimization of weighted aggregated sum product assessment. *Elektronika ir elektrotechnika*, 122, 3–6.
- Zavadskas, E. K., Turskis, Z., & Bagočius, V. (2014). Multi-criteria selection of a construction site for a deep-water port in the Eastern Baltic Sea. Applied Soft Computing.
- Zeydan, M., & Çolpan, C. (2009). A new decision support system for performance measurement using combined fuzzy TOPSIS/DEA approach. *International Journal* of Production Research, 47, 4327–4349.
- Zhang, H., Gu, C.-L., Gu, L.-W., & Zhang, Y. (2011). The evaluation of tourism destination competitiveness by TOPSIS & information entropy-a case in the Yangtze River Delta of China. *Tourism Management*, 32, 443–451.
- Zhang, X., & Xu, Z. (2014). Extension of TOPSIS to multiple criteria decision making with pythagorean fuzzy sets. International Journal of Intelligent Systems, 29, 1061–1078.
- Zheng, G., Zhu, N., Tian, Z., Chen, Y., & Sun, B. (2012). Application of a trapezoidal fuzzy AHP method for work safety evaluation and early warning rating of hot and humid environments. *Safety Science*, 50, 228–239.
- Zhou, Q., Huang, W., & Zhang, Y. (2011). Identifying critical success factors in emergency management using a fuzzy DEMATEL method. Safety Science, 49, 243–252.
- Zhu, B., Xu, Z., & Xia, M. (2012). Hesitant fuzzy geometric Bonferroni means. Information Sciences, 205, 72–85.
- Zimmermann, H.-J. (1978). Fuzzy programming and linear programming with several objective functions. *Fuzzy Sets and Systems*, 1, 45–55.
   Zouggari, A., & Benyoucef, L. (2012). Simulation based fuzzy TOPSIS approach for
- Zouggari, A., & Benyoucef, L. (2012). Simulation based fuzzy TOPSIS approach for group multi-criteria supplier selection problem. *Engineering Applications of Artificial Intelligence*, 25, 507–519.